

https://theses.gla.ac.uk/

Theses Digitisation:

https://www.gla.ac.uk/myglasgow/research/enlighten/theses/digitisation/

This is a digitised version of the original print thesis.

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses
https://theses.gla.ac.uk/
research-enlighten@glasgow.ac.uk

SUMMARY.

THE SAMARITAN TEXT OF THE PENTATEUCH.

A Comparison of the Samaritan Text with the Masoretic Text.

The purpose of this Thesis is to endeavour to evaluate the textual variants of the Samaritan Hebrew Pentateuch and, if possible, indicate lines upon which the text might be restored as it existed at the time of the Samaritan secession.

Texts collated are:

- 1. The Masoretic Text, as published by Kahle in Kittel's Biblia Hebraica, 7th. ed., Stuttgart, 1951.
- The Samaritan Text, as reconstructed by A. von Gall,
 Dor Hebräische Pentateuch der Samaritaner, Giessen,
 1914-18.
- 3. The Pentatouch texts of the Dead Sea Scrolls in Discoveries in the Judaean Desert, I,II, & III, in the case of all the Samaritan variants.
- 4. The Septuagint, ed. A. Rahlfs, Stuttgart, 1935, in the case of the Samaritan variants that show difference in sense, or where proper names occur.

The introductory Chapter I consists of a short history of/

ProQuest Number: 10647842

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10647842

Published by ProQuest LLC (2017). Copyright of the Dissertation is held by the Author.

All rights reserved.

This work is protected against unauthorized copying under Title 17, United States Code

Microform Edition © ProQuest LLC.

ProQuest LLC.
789 East Eisenhower Parkway
P.O. Box 1346
Ann Arbor, MI 48106 – 1346

of studies on the Samaritan text of the Pentateuch.

Brief mention is made of scholars prior to Gesenius but with him, and thereafter, increasingly detailed treatment is given.

Chapter II states the Methods adopted, namely, the collection, critical examination and classification of all the Samaritan Pentateuch variants. There is a discussion on the objection sometimes levelled against the von Gall text, that it is eclectic and that it has not made use of all the MS material available.

Chapter III prepares the way for the actual classification of the variants by dealing with certain preliminary considerations that affect the comparative study of the texts: it consists of three monographs; oral transmission. Pre-Masoretic Hebrew texts, and Scroll and Codex (the last boing a criticism of Gaster's view that the Scroll and not the Codex should be taken as the basis of comparison).

Chapter IV discusses the Principles to be used in the Classification of the varients and then gives an outline of the Classification which divides the material into three main groups, viz. Orthography, Grammar and Vocabulary and of the sub-division of the last two groups into variants that show (1) agreement in sense or (2) divergence in sense.

Chapter V gives a selective survey of the whole materia into a series of classified paragraphs, giving representative examples/

examples and referring the reader to the Appendix for complete lists.

Chapter VI contains the evidence of the Dead Sea Scrolls, showing in the case of the Samaritan variants where the Scrolls agree with the Samaritan, or the Masoretic Text, or diverge from both.

The final Chapter VII gives conclusions, obtained from the consideration of the Samaritan Pentateuch, regarded as a descendent of a type of text earlier than the Masoretic and belonging to the period of variant texts, prior to the Council of Jamnia, to which the Masoretic, the Samaritan, the Dead Sea Scrolls and the Greek texts commonly referred to collectively under the term LXX all go back.

Appendix A consists of a detailed list of the variants under the categories of the Classification outlined earlier in the Thesis. This exhaustive list constitutes a large part of the Thesis which it is hoped will be of permanent usefulness for reference.

Appendix B is an additional paper on two Samaritan MSS specially examined.

A Bibliography of Books and Periodicals concludes the Thesis.

THE SAMARITAN TEXT

ο£

THE PENTATEUCH.

A Comparison of the Samaritan Text with the Masoretic Text.

Submitted to the University of Glasgow, for the Degree of Ph.D. in the Faculty of Divinity

by

John Douglas McClymont.

Glasgow, 1967.

PREFACE.

Since early in the 17th century Europeans discovered a Samaritan form of the Hebrew Pentateuch, written in an ancient script and containing variants from the text of the MT, many different views have been taken of its value for the Textual Criticism of the MT. A summary of these views will be given in the Short History of Samaritan Studies, mainly from the textual standpoint, which will be found in the General Introduction, Chapter I. It is proposed to undertake a new treatment of the subject and the purpose of this Thesis is to endeavour to determine the value of the Samaritan variants as aids towards reconstructing the text of the Hebrew Pentateuch as it existed at the time of the Samaritan secession.

This new investigation appears to be necessary for the following reasons:

- 1. Earlier students had not considered all the Samaritan Pentateuch variants.
- 2. Von Gall's work, "Der Hebräische Pentateuch der Samaritaner, shows that there is a definite Samaritan recension of the Pentateuch but he offers no classification or discussion on the variants from the MT.

3. The Dead Sea Scrolls have provided new material which may have a bearing on the subject.

Texts collated:

- 1. Masoretic Text Biblia Hebraica, R. Kittel, 7th ed., Stuttgart, 1951.
- 2. Samaritan Text, as reconstructed by A. von Gall,

 Der Hebräische Pentateuch Der Samaritaner,

 Giessen, 1914-18
- 3. The Pentateuch texts of the Dead Sea Scrolls in Discoveries in the Judaean Desert, I, Qumran Cave I., D. Barthelemy and J. T. Milik, Oxford, 1955.
 II.Les Grottes de Murabba^c at, P.Benoit, J.T.Milik, R. de Vaux, Oxford, 1961. III. Les 'Petites Grottes' de Qumran, M.Baillet, J.T.Milik, R. de Vaux, Oxford, 1962. in the case of all the Samaritan variants from the M.T.
- 4. The Septuagint, ed. A. Rahlfs, Stuttgart, 1935, in the case of the Samaritan variants that show differences in sense, or where proper names occur.

The Glassification of the Samaritan variants (see below, Chapters IV and V) is an original one and relevant to the problems that arise in the investigation of the subject: it should be useful to other investigators in their assessments of the value of the Samaritan variants and for easy reference and further investigation. The preceding/

preceding Chapters, it is submitted, also show an independent treatment e.g., Chapter III, 3., discussion on the relative merits of Scroll and Codex. The Conclusions in Chapter VIIare based on the evidence of all the Samaritan variants, and not just a proportion of them, and take into account the relevant evidence of the Dead Sea Scrolls. In the Appendix the citation of the Samaritan variants is considerably fuller than that which has hitherto appeared in any published work on the subject and should also be useful for easy reference and further investigation.

The writer expresses his appreciation of past and present scholars whose works have been consulted and, in some cases, discussed in the Introductory Chapter. He has a large debt of gratitude to the late Prefessor W.B. Stevenson who first introduced him to Samaritan studies, to Professor C. J. Mullo Weir for invaluable guidance and the stimulus of his scholarship and to Professor J. Mauchline for encouragement and interest. He is also grateful to the Librarians and their staff at Glasgow University, Trinity College, Glasgow, and Trinity and Westminster Colleges, Cambridge.

C O N T E N T S.

Preface.

Chapter I.	Page
General Introduction: Short History of Samaritan Studies.	1.
Chapter II.	
Object and Methods of the Present Study	30.
Chapter III.	
Preliminary Considerations affecting Comparative Study of the Texts.	36.
Chapter IV.	
Principles of the Classification.	ls ls .
Chapter V.	
Classification of the Samaritan Variants from the Masoretic Text.	54.
Chapter VI.	
Evidence of Dead Sea Scrolls.	119.
Chapter VII.	
Conclusions.	131.
Appendix A: List of Samaritan Variants under the Categories of the Classification outlined in Chapters IV and V.	155.
Appendix B: Additional paper on two Samaritan Manuscripts.	264.
1. Facsimile of MS belonging to Professor J. Garstang.	
2. MS. R.16.41 Trinity College Library, Cambridge.	
Abbrevlations.	285.
Bibliography.	287.

CHAPTER I.

General Introduction.

Short history of Samaritan Studies.

The Samaritans in the time of Christ were inhabitants of a part of the territory of old North Israel who still kept themselves independent of the priesthood and temple at Jerusalem. They had a temple and priesthood of their own on Mount Gerizim where they practised forms of worship descended to them from old Hebrew times. They derived the authority of these forms from Moses and treated his book, the Pentateuch, as their sacred book.

After the Jewish state and temple were destroyed in the first Christian century they carried on without change till 130 A.D., when their temple also was destroyed and much of their treasures and of their ancient literature was lost. The Samaritans, like the Jews, now lived largely outside of Palestine, for example in Damascus, but unlike the Jewish priesthood their priesthood carried on in Samaria as best they could. The observances of the Mosaic Law were maintained, including down to modern times without a break the Passover ceremony. Both in Samaria and elsewhere they preserved independently of the Jews, and in a more ancient form/

form of writing, copies of the Torah of Moses.

Early in the 17th century came the discovery by Europeans of the Samaritan form of the Pentateuch. discovery was truly a landmark in Old Testament studies for here was a Hebrew Pentateuch in a strange script and with a text different from that of the Jews. Of course the Talmud and early Jewish writers recognised that the Samaritans had the Law of Moses, and indeed both Jews and Samaritans agreed that the Samaritan script was older than the square script subsequently adopted by the Jews. There is a wellknown passage in the Babylonian Talmud which says that originally the Law was given to Israel in the Hebrew script and in the sacred tongue and that it was again given in the days of Ezra in the Assyrian script and the Aramaic tongue: Israel chose the Assyrian script and the sacred tongue and the Samaritans were left with the Hebrew script and the Aramaic tongue2. The Christian Fathers also made reference to the Samaritan Pentateuch. Origen used it in his Hexapla3. Then the silence of the centuries descended and it seemed that even the memory of it had perished. Early in the 17th century came the astounding news that copies of the Samaritan Pentateuch had been obtained.

In 1616 Pietro della Valle obtained two copies at Damascus:/

Damascus: one was presented to the Vatican and the other was published in the Paris Polyglott (1645) and the London Polyglott (1657). Morinus, who edited the Paris Polyglott. claimed that the Samaritan text was considerably superior to that of the MT. His opinion was adopted in great measure by some scholars and rejected just as strongly by others⁵. There followed a period of acute controversy, much of the discussion and dispute being invalidated by the fact that the participants allowed pre-conceived theological ideas to influence them in their textual judgments. In the 18th century, however, a more objective approach was shown by Houbigant and Kennicott who drew attention to valuable readings of the Samaritan. Kennicott carefully noted its variants in his famous textual work on the Hebrew Bible, Vetus Testamentum Hebraicum 6. Scholars that followed such as Michaelis and Jahn took an eclectic view, holding fast to the general superiority of the MT but being prepared to adopt a Samaritan reading where the MT was clearly defective. A noteworthy attempt to bring scientific method to bear on the problem was made by Gesenius. De Pentateuchi Samaritani Origine Indole et Auctoritate Commentatio Philologica - Critica7, who was the first to make a comprehensive classification of a large number of the Samaritan variants, dividing them into eight main groups. His/

His view was that the Samaritan Pentateuch was a modernised form of the Masoretic Text, sometimes misinterpreted and sometimes intentionally altered. There the matter was to rest for some years. Yet Gesenius had proceeded on an unsound assumption and an unsound assertion. He assumed that the MT was primary and the Samaritan was secondary, arising by alteration from it; such an approach was obviously unsound, as both Pentateuchs should be given an equal hearing. He asserted (p.16 of his Thesis) that the Samaritan was derived from a MS in the square script, brought to Samaria in the time of Alexander the Great: this was impossible as the square script was not in use at that period.

The way was prepared for a more comprehensive study of the relationship between the two recensions by A. Geiger who in 1857 correctly emphasised that the care exercised by the Masoretic scholars for the purity of the Biblical text was not true of an earlier period when the MSS were somewhat arbitrarily handled and there were many corruptions.

He went on to show that only a small number of these corruptions survived the later critical methods and maintained themselves in the MT. Variants of that earlier period due to local, individual or sectarian difference were testified by the Samaritan Pentateuch and the Septuagint: they consisted of the removal of difficulties in morphology or/

or syntax, supplementations from parallel passages and the That Geiger's attitude to the Samaritan community like. was not unsympathetic was shown in a number of monographs. For example, in Zur Theologie u. Schrifterklärung der Samaritaner he referred to the similarity of their theological views with those of Judaism, especially the Sadducees, except in so far as their political antipathies were not involved9. Other monographs indicating this conservative attitude of the Samaritans were Neuere Mittheilungen über die Samaritaner 10, and Die Gesetzlichen Differenzen Zwischen Samaritanor u.Juden 11. Of a more textual nature was Geiger's Nachgelassene Schriften 12 wherein were discussed the Samaritan variants of a doctrinal nature, such as the reading of Gerizim for Ebal in Deut.274 or the Aramaisms and late Hebrew forms found in their text. His conclusion regarding the worth of the Samaritan Pentateuch is best expressed in the following translation of his own words which shows how different his attitude was from that of "The Samaritan text is a highly important Gesenlus: historical document which, apart from particular Samaritanisms in the case of Schachem - where it is not always in the wrong - and the Aramaisms, represents an old recension as it at that time had general distribution, and consequently appears/

appears also in the translations, namely the LXX, at times even in old Talmudic writings".

Another scholar at that period who prepared the way for a revival of interest in Samaritan studies was H. Petermann who actually journeyed to Samaria and met with Amram, son of the still living high priest, Schalama. He describes, Versuch einer Hebraischen Formenlehre nach der Aussprache der heutigen Samaritaner 13. p.4. how he read aloud a portion of the Samaritan Pentateuch (in the edition of Benjamin Blayney, Oxford, 1790) in the presence of Amram who informed him that he read according to the way of the Jews, but not according to that of the Israelites. He was invited by Petermann to read from the Pentateuch and readily Whereupon he read the passage aloud in agreed to do so. such a way that Petermann would not have understood it, if it had not been already known to him. Petermann lost no He at once vocalised the text to the reading of His first impression was that the reading was quite Amram. arbitrary but subsequently he discovered "consequences and laws which afforded cause for thought". For example, in all the multiplicity of forms there was still a great simplicity, "the most natural vowel a prevailed". the Qal. Pract., Ptc., Infin. Absolute and Imperative often had the same form, and the vowels maintained themselves firmer with not/

not so many changes as was the case with what Petermann called the artificial. Masoretic vocalisation, indicating that the Samaritan vocalisation was the older. of fact Petermann gave the preference to the Masoretic vocalisation. as he detected Aramaic and Arabic influence in the Samaritan vowels. At times, too, obvious errors occurred in the Samaritan reading and often to his ears there appeared to be divergence in pronunciation of one and the same word. On the whole, however, it was to be concluded that he who from youth up had read the Pentateuch and expounded it in the synagogue knew the pronunciation better than anyone else. Petermann then provides (pp.6 - 160) a formal scheme expounding the accidence of the Hebrew according to the pronunciation of Amram. transliteration of Genesis follows (pp.161 - 218) and finally a list of Samaritan variants for the whole Pentateuch (pp.219 - 326).

3.00

Early in the present century E. Meyer defended the Samaritan reading in Deut.XXVII.4 - Gerizim for Ebal 14.

With the notable exception of Kennicott almost all previous scholars had believed that the Samaritans in this passage had altered Ebal into Gerizim in the interests of their cult.

Meyer distinguished an early account of the origin of the cult/

cult at Shechem (Deut. 11^{29} , 27^{11-25}) from the later Deuteronomic interpretation (Deut.11 $^{26-28}$, 31 f, 27 $^{7-10}$). He stated that the MT then removes Gerizim and Ebal from the vicinity of Shechem to the Jordan valley at Gilgal. He maintained that the hill of blessing (Gerizim attested by both Samaritan and MT) should have the altar. Meyer's views are worthy of respect for it is evident that Chapter 27 reveals many critical difficulties. Most commentators are agreed that the various parts of it are imperfectly connected and the Chapter itself interrupts the Discourse of Moses. As against Meyer's view that the Samaritan reading of Deut.274 is original, it may be stated that all the versions agree with the MT and that the latter sees no reason to alter Gerizim as the hill of blessing. arguments on neither side are conclusive and the matter will be dealt with again later in the Thesis.

Another tribute to the Samaritans is paid by

J.Skinner who in his volume on the use of the Divine Names

in Genesis' has a section dealing with the Samaritan Pentateuch
and brings forward evidence that, in the transmission of
the Divine Names, there are at most 9 divergences as compared
with 310 cases of agreement in MT and Samaritan 15.

In 1907 appeared a comprehensive study of the Samaritans by J. A. Montgomery giving a scholarly introduction to/

to their history, literature and theology. His careful, documented account is in marked contrast to many earlier controversial opinions. Regarding the Samaritan Pentatech he says: "As to relation of Samaritan to MT and its origin, most different views have been held varying all the way from the extreme Catholic position that Samaritan was to be preferred to that which would make it dependent on the LXX. All mysteries and prepossessions aside, the simplest hypothesis is that the Samaritan represents an actual early form of the Pentateuch text. When we recall that the orthodox Jews of Alexandria produced a translation that varies from our present MT., we cannot be surprised that the heterodox Samaritans, who long before the origin of the LXX possessed their own textual tradition, have preserved a variant text. Indeed, it is not the disagreement that is remarkable so much as the great similarity of the two texts"16. Probably there are not many that would quarrel with this view.

An important event in Samaritan studies was marked by the publication of A. von Gall's <u>Der Hebräische Pentateuch der Samaritaner</u> which is used as one of the sources listed in the Preface. The author gives an impressive catalogue of the Samaritan MSS used in the work extending to 40 more or less complete MSS, which are designated by letters of the/

the alphabet. He exhausts the whole Roman alphabet and has to draw on German type to the extent of 15 letters. In addition, there are 30 groups of fragments of MS scrolls and 25 groups of fragments of MS codices. Many of these MSS he describes in great detail. Von Gall's work has been criticised on the ground that it has as its basis an eclectic method as compared, for example, with Biblia Hebraica III which reproduces the text just as it stands in the Leningrad MS and further that it has not used all the MS material available. For a full treatment of these objections see Chapter II pp. 31-34. At this stage it is only fair to point out that in his Prolegomena (p.68 ff) "Grundsätze fur die Bearbeitung der Vorliegenden Text - Ausgabe! he states his view that the printing could not rest on one MS alone. He found he could not produce a family tree of the MSS. Relationships were discernible but no families could be reconstructed. He admits he made continual comparison with the LXX and Jewish text and that his results have a measure of subjectivity. As he says, he makes no claim to infallibility and, if the reader is not satisfied with the proposed text, he is at liberty to choose from the variants in the extensive critical apparatus available on each page. It is generally agreed that the publication of von/

von Gall's work marked a new stage in the Samaritan field.

A modern scholar who prepared the way for a better understanding of the Samaritan Pentateuch is P.Kahle who. as far back as 1915, in Untersuchungen zur Geschichte des Pentateuch-Textes, p. 399 ff., develops the view that the received text, while considerably older than the Samaritan Pentateuch, had not been handed down unchanged from the pre-Christian era 18. Kahle shows that the LXX and citations in Pseudepigrapha show obvious unfamiliarity with the M.T. and numerous correspondences with readings in the Samaritan Pentateuch. He concludes that the received text did not represent an unbroken textual tradition but emanated from the textual criticism of the Masoretes, being a result of a critical restoration of the current vulgar texts. The aim of this movement was to get back to an ancient reliable text in which ancient readings and forms long since discarded in current texts were now restored. Glosses, additions, Aramaisms, etc., all were cut out. That the Jewish scholars possessed ancient textual material and that MSS were collated by them is attested by a well-known Rabbinical tradition. e.g. Jerusalem Talmud, Tacanith 4.2., to the effect that, when the three copies of the Torah preserved in the temple did not agree, the reading given by two of them was selected. The LXX original, the Pentateuch citations in the Apocrypha and/

and Pseudepigrapha and the Samaritan Pentateuch stood outside this movement and provide examples today of the erratic vulgar texts rejected by the Masoretes. The Apocrypha and Pseudepigrapha were left untouched as they were viewed by the orthodox Jews with indifference, while the LXX, as the Bible of the Christians, and the Samaritan Pentateuch remained uninfluenced.

Kahle argued that the Samaritan must be viewed as a text of high antiquity adjusted to practical use. Apart from rare emendations due to the Samaritan cult, there was no reason to regard it as specifically Samaritan. It was in fact simply a text in common use among Jews in pre-Christian times and this is borne out by its relationship to the LXX and Pseudepigraphic literature. Most scholars accept Kahle's view that prior to the formation of the MT there were quite a number of variant texts and his view that the Masoretes collated a large number of texts when they formed the MT.

Among the many publications from Kahle during his long life was The Cairo Geniza¹⁹ which is an important volume for Samaritan studies and to which references are made throughout this Thesis. Again, in a paper contributed to the Pedersen Festschrift in 1953 Kahle has a long, detailed account of the famous Abisha scroll in Nablus²⁰, which is first/

first mentioned in a marginal note in a Tolida or Chronicle of the Priesthood in 1149, recording how the Scroll being carried in procession was being swept away by a sudden storm whereupon the congregation grasped it and a portion remained with them - Numbers 35² - Deuteronomy 34.10 The next reference is in 1352 when the Samaritan historian Abu'l-Fath reported that the Scroll was shown to the community and that all present were able to see the cryptogram beginning at Deuteronomy 610 stating that the Scroll was written by Abisha, son of Pinhas, son of Eleazar, son of Aaron in the thirteenth year after the Israelites had entered the Holv Land. Descriptions of the scroll have been supplied by Europeans from time to time, as mentioned in the monograph. himself studied it in the synagogue at Nablus. According to him the ancient part of the Scroll consists of Deuteronomy and the last chapter of Numbers supplemented by a text written about the 13th century and the really old part may take a significant place in the textual criticism of the Hebrew Bible. He also stated that Professor Perez Castro had been successful in obtaining a photograph of the complete Scroll21, and the publication of his results would enable a final decision to See below, p. 34 for E. Robertson's review of Castro's work which has now been published.

It has been noted how Kahle has emphasised that the LXX/

LXX and citations in the NT and the Pseudepigrapha show obvious unfamiliarity with the MT and numerous correpondences with readings in the Samaritan Pentateuch and that he has developed the view, generally accepted, that prior to the formation of the MT there was quite a number of variant texts. An interesting and convincing supplement to Kahle's position is to be found in a monograph by J. Hempel Innermasoretische Bestätigungen des Samaritanus²² who shows that a group of mediaeval Masoretic MSS in the material collected by Kennicott and de Rossi show correspondences with the Samaritan Pentateuch in such matters as tendency to plane writing, a preference for syndesis, and omissions and additions, as distinct from the unified impression of the later tradition. The conclusion that suggests itself is that these MSS witness to a variant text form within the Masoretic tradition itself, and which contained readings of the old popular texts which survived for centuries, prior to the unified text-form that is found in 10th century MSS and later.

Another modern scholar is M. Noth²³ who states that the great majority of the Samaritan variants are for the most part of an orthographical nature or are concerned with stylistic minutiae so that the amount of "sachlich bedeutsam" variation is very small. He questions the assumption generally made that after the cultic separation of the Jewish and/

and Samaritan communities the two texts of the Pentateuch ran a completely independent course. In support of his view he refers to Hempel's monograph, that has just been noted, and demonstrates that in the case of Deuteronomy a group of mediaeval Masoretic MSS agree in many details with the Samaritan text. He argues for long lasting relationship between the text traditions on both sides. Noth's argument is valuable at least for giving a salutary warning as to the gaps in our historical knowledge regarding the date and early centuries of the Samaritan schism. Too often the relationship known to exist in later times, when the Jews had no dealings with the Samaritans, has been read into the earlier Yet amidst much that is obscure one fact emerges quite clearly that, with the promulgation of the 2nd canon about 200 B.C. a definite break must have separated Jews and Samaritans regarding the extent of Scripture. The fact that many centuries afterwards a group of MT MSS has variants found in the Samaritan Pentateuch may simply show that in some MSS there were still variants surviving from the old popular texts which had not at that date been wholly eliminated from the MT tradition.

A well known writer on the Samaritans is M.Gaster who published a number of works such as the volume in the Schweich Lectures 24 and an earlier volume on the Samaritan Oral/

Oral Law²⁵, both of which provide a background of Samaritan life and custom. Of a more textual nature and consequently of more value for the purpose of this Thesis is his A Samaritan Scroll of the Law of the XIth century 26. wherein Gaster referred to the extreme care with which the Jews wrote the synagogue scrolls in contrast with the codex. in which there was much greater liberty of production. He claimed that the text of the Jewish Pentateuch Scroll was superior to that of the Codex and he made a good point when he asked if the same circumstance might not hold good for Samaritan Scroll and Codex. The MSS of the Samaritan Pentateuch on which the Paris and London Polyglotts were based were all without exception in codex form. first Appendix to Nutt's Fragments of a Samaritan Targum 27 reference is made to the view of Harkavy that the Samaritans held the Scrolls as especially sacred, from their being intended for use in the synagogue, and so would not part with them to those of another faith. Since then, according to Gaster, a few fragments of scrolls had come into the library of the British Museum and another scroll fragment was in Gaster's possession. He claimed that the comparison of the Samaritan Pentateuch and the MT must be taken up anew and carried out not with the aid of the Codex but with the scroll./

scroll. For a refutation of this view of Gaster's see Chapter III, pp. 39443.

Another writer that should be mentioned is J. E. H. Thomson²⁸ for a comprehensive volume covering such topics aa Samaritan history, topography, theology and the ancient alphabets with special reference to the evolution of the Samaritan script. On the textual side there is a good chapter on the relation of the Samaritan Pentateuch to the LXX wherein Thomson argues convincingly that they are independent witnesses and one has not been corrected from the In another chapter he deals with the Samaritan Pentateuch variants from the MT: he makes an effective criticism of earlier attempts at classification by Gesenius, Kirchhem and Kohn and then proceeds to develop his own scheme, based on "a primary division into two leading classes; first, Variants due to Accident, and next, Variants due to Intention". His scheme of classification is itself defective for surely a logical scheme must avoid an analysis based on a subjective distinction between intention and un-intentional and take the facts as they are in an objective way.

The next publication to be noted is one by Ch.Heller

The Samaritan Pentateuch, an Adaptation of the Masoretic Text

Therein it is categorically stated that the Samaritan text

of the Pentateuch presents only a systematical edition,

adapted/

adapted from the MT. and that it is without justification. that it has hitherto been regarded as an independent textual Heller attempts to justify his uncompromising attitude by arguments which do not prove so much as he contends. For example, he affirms that the deviations of the Samaritan from the MT are in agreement with and dependent on views of late Rabbinical scholars. He quotes Siphre to Deuteronomy 1130: "R. Elieser, the son of R. Jose, said to the Samaritan scholars: You have falsified the Thora text. but you have not gained anything by it. Thus you have written next to the words אוני מורה אר אר לוני מורה אצר אני מורה we also take it in the same sense by drawing a parallel with Genesis 126 where the verse says און מורה עד הלון hence as 3N there is 22ω , so here also it means D>ω". He concludes from this passage that the Samaritans took the liberty of inserting their interpretations into the Bible itself. For, as he says, there could not have been any talk of falsification in the addition of the word αρω, such in fact being the true meaning of the sentence. This method, to alter the text according to the interpretation, was so offensive to the Jewish scholars, who scrupulously insisted on the inviolability of the Biblical text. Of course, even if it were granted that the Samaritans had introduced/

introduced interpretations into their text to a much greater degree than is to be found in the MT, it is quite another matter to argue that these are done in conscious dependence on it. They may well have arisen quite independently. Heller refers to:

Genesis 7³ מעף השמים שבעה שבעה MT

'ש' ש' ה' הטהור ש' ש' ST

where the Samaritan reading is in line with Jewish exegesis. It may be remarked that verse 3a in both Samaritan and MT is obviously an impure text and should probably be deleted as a gloss but, taking it as it stands, it is clear that the correspondence of the Samaritan reading with the Jewish interpretation may have arisen quite fortuitously. Another example cited by Heller may be taken from Deuteronomy 221,4 where the Samaritan has the extra words 7572 35 after the "sheep" or the "ox". These extra words are in accordance with Halakic and Haggadic traditions found in Jewish sources wherein these animals are named only as examples, the precept applying to all animals in like manner. Once again. this need not involve Samaritan dependence. What it does mean here is that the Samaritans have made their Oral Law explicit in their text. It would in fact be difficult for the two traditions to diverge here. These and other arguments brought forward by Heller are not convincing.

Another/

Another modern scholar who has contributed to Samaritan studies is G. Gerleman who makes frequent reference to the Samaritan Pentateuch in Synoptic Studies in the OT30. In Chapter II, p.8 ff. in dealing with the Book of Chronicles and its sources, he states that the reason why Samuel-Kings and Chronicles deviate linguistically is. in his opinion. to be sought in the fact that these writings were treated in different ways by Jewish textual criticism. In the books of Samuel-Kings, classed among the Prophets, we are concerned with a carefully revised and archaised text whereas the Hagiagraphic Books of Chronicles do not in the same degree show traces of critical revision. He also points out that Chronicles has a great linguistic resemblance to the Samaritan For example, in forms of names MT of Genesis 104 Pentateuch. reads D'1771 while Samaritan with MT of I Chronicles 1.7 reads O'J771 . In lists of names Samaritan and Chronicles agree in using syndeche for the archaic asyndetic construction

Linguistically, Chronicles deviates from Samuel-Kings in a way very similar to the deviations of Samaritans from MT. Orthographically, there is the more frequent use of matres lectionis, morphologically, the imperfect consecutive has lst.sing. and lst. pl. ending in 7—, syntactically, there is/

17101

0'7561

610

Genesis 10⁶

is the preference for 3y to 3x, plurals for Collectives and Collectives used oftener with pl. verbs. There is also the preference for Hiphils and the greatly lessened use of Infinitive Absolute as an Imperative or Adverbial Qualification. In the matter of composition also there is affinity for we find a certain supplementation and duplication of narrative in Chronicles similar to the recapitulation technique of the Samaritan Pentateuch. Gerleman affirms that the linguistic affinity of the Samaritan Pentateuch with the Hagiagrapha shows that its variants are due not to a provincial dialect but to a younger and more vernacular stage of Hebrew than in MT.

According to Millar Burrows 31 Gerleman oversimplifies the problem of drawing a line between the Hagiagrapha as representing the vulgar text and the Pentateuch and Prophets as showing results of archaising revision; further and fuller evidence is needed, before it can be affirmed that the Masoretic revision of the current text was based on an older, more accurate tradition which had perished along with the freely developing popular text; for example, there may have been a deliberate revision on the theoretical ground that scriptio plane was a novelty or, yet again, a simplification after the introduction of vowel signs made matres lectionis unnecessary. Further, Gerleman does not give enough weight to the possibility of dialectal difference. For a further discussion/

discussion on this point, cf. the next writer to be considered,
A. Sperber, or C. F. Burney³².

Two important monographs on dialectal difference come from A. Sperber. In the first, Hebrew in Greek and Latin Transliteration 33 he draws his material from the transliterations of proper names in the LXX, the material preserved from the second column of Origen's Hexapha and the transliterations of Jerome, and proves conclusively that the Hebrew revealed had a very different pronunciation prior to the rise of the Tiberian system of vocalisation. Chapter xxxiii, "Two Dialects of Biblical Hebrew" he affirms that the differences are such that they reflect a two-fold way of pronouncing Hebrew. Though going back to a common ancestor, they are two separate branches each with a phonetic development of its own. He regards the Kingdom of Israel and Judah as the homelands of these dialectal differences. Reference is made by him to the well-known Ephraemitic peculiarity of pronunciation, cf. Judges 126, the distinction between & and & belonging to the Judaean dialect. Reference is also made to possible Samaritan dialectal forms e.g. according to Samaritan usage the waw consecutive does not at all affect the accentuation and vocalisation of the respective verbal form. This is also true of the translitera-He concludes that the Samaritan Pentateuch represents the Israelitish recension, while in MT the Judaean recension is/

that neither can be regarded as representing its prototype in original dialectal purity as the textual changes they underwent in various stages of redaction may have eliminated dialectal idioms. Sperber himself admits the limited nature of his material but he argues his case well and his views should receive the highest respect. With reference to this monograph Kahle recognised Sperber had further material to investigate and drew attention to a volume on Samaritan pronunciation by Fritz Diening³⁴.

In the second monograph, Hebrew in parallel transmission Sperber furnishes a careful, detailed classification of variants between Hebrew and Samaritan Pentateuchs according to A. Phonetic Value of Letters: B. Graphic Forms of Letters: C. Vowels and their Phonography: D. The Verb: E. The Noun: F. Nominal and Verbal Suffixes: G. Syntax: H. Vocabulary. It is difficult to determine in many cases whether the Hebrew or the Samaritan is the original and Sperber does not attempt to do so.

Another modern scholar to be mentioned here is

P. W. Skehan for his valuable work on the Qumran Pentateuchal fragments in which he states that books of the Pentateuch still circulated in Palestine down to the First Revolt in copies with varying degrees of recensional backgrounds 36; as evidence of/

of this he instances the palaeo-Hebrew scroll of Exodus which cannot be considered a Samaritan sectarian document - "in view of the known attitude of the community towards 'the men of Ephraim and Manesses' the probabilities are all against it" - though it does offer a type of text the Samaritans have possessed as their own, with its repetitious manner of recounting the plague episode, its borrowings from Deuteronomy and its transpositions. Reference is also made to a MS of Numbers that contains expanded readings hitherto known only from the Samaritan recension. These references in Skehan are indeed valuable in that they clearly indicate that the expansions of the Samaritan Pentateuch are at least as old as Qumran and are not specifically sectarian in character.

A scholar who has performed a valuable service to

Samaritan studies by warning against the unevidenced statements that have often been made regarding Samaritan history is

H. H. Rowley who in a Bulletin of the John Rylands Library,

Manchester, Samballet and the Samaritan Temple 37 demolishes certain assumptions and pre-conceived ideas. He points out the limitations of the data that have been used by some past writers in the endeavour to determine the date of the Samaritan schism. There is the enigmatic reference in Nehemiah 1328 which tells of the banishment of the grandson, unnamed, of the high priest Eliashib for marrying the daughter of/

of Sanballet (432 B.C.) and the circumstantial narrative in the eleventh book of the Antiquities of Josephus, § 302-324, which also tells of a banishment for a like reason, but in this case a great-grandson of Eliashib, called Manasseh, shortly before the coming of Alexander the Great Josephus expressly connects this incident with the erection of the Samaritan temple on Mount Gerizim. Rowley warns against the assumption that the banished Jerusalem priest became the Samaritan high priest. There is no hint of this in the Bible - certainly not in Nehemiah's statement - and Josephus' account is obviously unreliable. Further, as he emphasises, the Samaritan schism and the erection of their temple, are two quite distinct facts and may quite well belong to different periods. It is unnecessary to trace his arguments in detail. His conclusion that the schism occurred within the period 430-350 B.C. would probably be accepted by the majority of scholars today.

In a recent volume of Scripta Hierosolymitana 38

S. Talmon gives an interesting and valuable paper on

Synonymous Readings in the Old Testament. He emphasises that

not all textual variants arose from scribal error or deliberate

emendation but that there are some that had their source in

a different textual tradition from that of the MT. The fact

that the Samaritan community and the sect of the Judaean

desert /

desert have on occasion textual variants in common, and differing from the MT, indicates the antiquity of the sources Further, the Samaritan Pentateuch underlying them. occasionally has readings differing from the MT of the Pentateuch but identical with the parallel reading preserved in the MT of Chronicles. Such readings would belong to different textual traditions going back to the common heritage Talmon goes on to refer to a class of variants of Judaism. which he characterises as synonymous readings: there is no sign of systematic or tendentious emendations and the words or phrases are used interchangeably. Where such variation occurs it is impossible to decide that any one variant is intrinsically preferable to the others. Such readings go back to an early stage in the history of the OT text to that period of divergent traditions that preceded the MT. Talmon then presents a long series of word-pairs, e.g. 727 - 70~ whose separate components may be used interchangeably within the same passage or parallel passages of a text or as alternatives in MT and an extra-Masoretic text, such as the Samaritan or the Dead Sea Scrolls. Talmon provides much interesting material that deserves careful consideration and references to it will be made later in Chapter V, Classification of the Samaritan variants.

A well-known OT scholar who has written on the Samaritans from the textual standpoint is E. Robertson. In his/

his Catalogue of the Samaritan MSS in the John Rylands Library 39. he details MS material not included in the von Gall edition. A statement regarding these MSS is given in Chapter II of this Thesis. Robertson has also written an important paper in the Gaster Centenary Publication on the famous Abisha Scroll 40. He tells something of its long history and gives the accounts of Samaritan and European eye-witnesses, including Gaster. Their descriptions are given in a certain detail. The writing appears to be homogeneous; though there are indications of different hands and different periods, a long interval need not be postulated between them. He hesitated to accept Kahle's theory of a very early portion of the Scroll combined with later material and suspended judgment on this matter until the publication of Professor Pérez Castro's photographs of the complete scroll. results have now been published and a very full review is given by Professor Robertson in Vetus Testamentum 41. He pays tribute to Castro's meticulous and scholarly work which provides a detailed description of the famous scroll and of which he gives quite a full account. It is now possible to reach a decision regarding the textual work of the MS. The Scroll turns out to be not a unit, but a composite of a number of elements. No fewer than nine different hands have been identified. Portions of scrolls of different ages are/

are held together by a backing and built up to form a complete Torah. In the words of Robertson, "The value of the Scroll as a witness to the early text of the Pentateuch must inevitably be greatly impaired, although perhaps not entirely destroyed."

Accordingly there is no question of the von Gall text having been displaced as the standard authority on the Samaritan Hebrew Pentateuch.

Another well-known OT scholar who has worked in the Samaritan field is John Bowman, formerly of Leeds University and now of Melbourne. He has contributed to Samaritan studies by articles in various periodicals on such subjects as Samaritan literature, religion and history, e.g. The Leeds Samaritan Decalogue Inscription 42. The Exegesis of the Pentateuch among the Samaritans and Rabbis 43. The Samaritans and the Book of Deuteronomy 44. The Importance of Samaritan Researches 45. In 1950 he founded a School of Samaritan Studies at Leeds and gave generously of his time Under his guidance comprehensive and detailed research was undertaken by students on such varied topics as Samaritan Liturgy, Texts, Theology and Exegesis. Since his departure to Melbourne the activity of the School is being maintained under his successor. John Macdonald. A wide field for research is being opened up by the extensive library of Samaritan writings consisting of manuscripts and microfilm of the Department of Semitic Languages and Literatures at Leeds. John/

John Macdonald has recently published two studies on Samaritan Theology that will now be considered. The volumes referred to are Memar Margah 46 and The Theology of the The former promises to become the standard work on Memar Margah and its text and the latter should introduce Samaritan Theology to the English speaking world. In nomenclature, Macdonald makes a precise distinction between Judaean and Judaist: he understands Judaean as describing the Hebrew tribes of southern Palestine and Judaist as referring to the religion of post-exilic Judaeans, restricted to the period beginning with Ezra. The term Judaists has been used instead of the term Jews which carries with it too wide a connotation. This precision in definition makes for clarity of thought and expression in his subsequent In his Introduction he correctly emphasises the element of polemic in the historical writings of the OT: when even the two Judaean marratives of Kings and Chronicles can differ widely, it is only reasonable to grant that the Samaritan view of history, in its differences from that of the MT, should not be rejected out of hand but at least Macdonald proceeds to discuss the various examined. Samaritan beliefs such as the Oneness of God, the Law and the Day of Vengeance and Recompense. His exposition of the Samaritan Theology lends support to some of the conclusions reached in the classification of the doctrinal variants later in this Thesis.

Footnotes, Chapter I.

- 1. J.A.Montgomery, The Samaritans; the Barliest Jewish Sect. Their History, Theology and Literature. Bohle Lectures, 1906. Philadelphia, 1907. p.165.
- 2. Sanhedrin, 21 b.
- 3. Montgomery, p.286.
- 4. J.Morinus, Exercitationes Ecclesiasticae in utrumque Samaritanorum Pentateuchus. Paris, 1631.
- 5. De Muis, Assertatio veritatis Hebraicae adverus J.Morini Exercitationes, 1631.
 - J.H.Hottinger, Exercitationes anti-Morinianae de Pentateuche Samaritano. Zurich. 1644.
- 6. Vetus Testamentum Hebraicum cum variis lectionibus, 2 vols.

 Oxford, 1776-80.
- 7. Halle, 1815.
- 8. Urschrift u. Ubersetzungen der Bibel in ihrer Abhängigkeit von der innern Entwickelung der Judenthums.

 Breslau. 1857. pp.97 ff.
- 9. ZDMG xii (1858) pp. 132 42.
- 10. ZDMG xvi (1862) pp.714 28.
- 11. ZDMG xx (1866) pp. 527 73.
- 12. 5 vols., Berlin, 1875 78.
- 13. Leipzig, 1868.
- 14. Die Israeliten u. ihre Nachbarstämme. Halle, 1906, pp. 543 ff.
- 15. The Divine Names in Genesis. London, 1914, p. 289.
- 16. Montgomery, p.289.
- 17. Giessen, 1914 18.
- 18. TSK, 1xxxviii, 1915, pp.399 439.

- 19. London, 1947.
- 20. Studia Orientalia Ioanni Pedersen dicata, Hanniae, 1953, pp. 188-192.
- 21. Perez Castro, 'El sefer Abisa', <u>Sefarad</u>, xiii, 1953, pp. 119-129.
- 22. ZAW xi N.F. 1934, pp. 254-274.
- 23. Die Welt des Alten Testaments. Berlin, 1953.
- 24. The Samaritans. Schweich Lectures, 1923.
 London, 1925.
- 25. Samaritan Oral Law and Ancient Traditions (London), 1932.
- 26. London, 1901.
- 27. London, 1874.
- 28. The Samaritans: Their Testimony to the Religion of Israel. Edinburgh, 1919.
- 29. Berlin, 1923.
- 30. Lund, 1948.
- 31. JBL 1xviii, 1949, pp. 382-84.
- 32. Notes on the Hebrew Text of the Book of Kings.
 Oxford, 1903, p.208 f.
- 33. HUCA xii-xiii, 1937-38, pp. 151 274.
- 34. Das Hebräische bei den Samaritaneru. Stuttgart, 1938.
- 35. HUCA xiv, 1939, pp. 161-249.
- 36. "Exodus in the Samaritan Recension from Qumran"

 JBL, 1xxiv, 1955, pp. 182-187

 "The Qumran MSS and Textual Criticism"

 VTS iv, 1957, pp. 148-60.
- 37. Vol. 38, No. 1, September 1955.
- 38. Vol. viii, Jerusalem, 1961, pp. 335-83.
- 39. Manchester, 1938.
- 40. The Ancient Scroll of the Samaritans'in Gaster Centenary Publication. London, 1958.

- 41. xii, 1962, pp. 228 435.
- 42. Proceedings of the Leeds Philosophical Society. Vol.VI, Part viii, pp. 567-575
- 43. Outtestamentische Studien, viii.
- 44. Glasgow University Oriental Society, Vol. XVIII (1958-59) pp. 9-18.
- 45. Annual of Leeds University Oriental Society, Vol. I., pp.43-54.
- 46. Beih. ZAW.

Berlán, 1963.

47. London, 1964.

160

CHAPTER II.

Object and Methods of the Present Study.

After the short review of Samaritan studies in the previous Chapter it now follows to indicate the scope and method of the investigation now to be undertaken.

With regard to the textual variants of the Samaritan Hebrew Pentateuch from the MT Pfeiffer says that further research is required to appreciate fully their significance and to utilise them for a reconstruction of the earliest text of the Pentateuch¹. The purpose of this Thesis is to endeavour to evaluate these variants and, if possible, restore the text as it existed at the time when the Samaritans seceded. In furtherance of this purpose the aim is to classify existing material and also to take into account new material that has come to light.

examination and classification of the textual variants between the Samaritan Pentateuch and the MT, collating the whole Pentateuch, with a view to determining the number, nature and extent of the variants. Previous writers have discussed a proportion of the Samaritan variants but none appear to have covered the whole field and considered all the evidence. Such/

Such conclusions, drawn from only part of the evidence, may be confirmed, modified or disproved when the whole conspectus is undertaken in this way.

The sources, listed in the Preface, pages i, ii, to which this method is to be applied are the well-known Biblia Hebraica of Kittel and the von Gall text of the Samaritan. Further, in the case of all the Samaritan variants, comparison is made with the Pentateuch Texts in the recently published <u>Discoveries in the Judaean Desert</u>. Apart from these three sources, the LXX is consulted where variants show differences in sense or where proper names occur.

In the criticism of these sources it is hardly necessary to state why the text published by Kahle in Kittel's Biblia Hebraica has been adopted in this Thesis as the standard MT. It is recognised by most scholars today as the closest text to the original MT now available.

The von Gall text requires further discussion.

When it appeared it was hailed as an outstanding achievement in Samaritan Pentateuch studies. There are few today who would seriously question this view. This critical adition of the Samaritan Hebrew Pentateuch was something completely new: the detailed and elaborate critical apparatus on every page of the text showed with what meticulous care the editor had proceeded with his work. Yet in recent years its reliability has been questioned on two grounds, that it is an/

an eclectic text and that it has not made use of all the MS material available.

The first objection comes from Dugmore who questions the von Gall text for its eclectic procedure. He states that there is no standard text of the Samaritan Pentateuch and. in the absence of such, he has collated two Samaritan MSS with that printed by Blaney (1790) based on the Paris Polyglott (1645)². Of course a case might be argued against an eclectic text alleging that it might become an artificial production depending on the subjective judgment of the editor, but it is only fair to point out that von Gall in his introduction frankly acknowledges his eclectic method stating that the prospect of arriving at an archetype text on which the Samaritan was based is as unlikely as in the case of the Jewish text3. Further, while in Samaritan there may be no standard text. such as developed in Masoretic circles, it cannot be denied that there is a uniform text. Apart from orthographical variation such as plane and defective writing of and waw and mistakes with the guttural letters to which the Samaritans were prone, there is an astonishing unanimity between the many MSS used by von Gall and detailed in his critical apparatus. In the words of Kahle: centuries the Samaritan Pentateuch also has been a textus receptus without various readings. We have seen that a text similar to it must have been used in Jewish circles also, in the/

the time before all the earlier texts were replaced by the authoritative Hebrew text" 1. If von Gall has not given an accurate rendering of a particular Samaritan MS he has given what has every appearance of being an accurate representation of the Samaritan text tradition.

The second objection, that exhaustive use has not been made of all the MSS available, appears to have greater weight. E. Robertson gives details of Samaritan MSS. 2 complete codices, 3 large portions and many fragments in the John Rylands' Library, which were not included in the von Gall edition. though the two complete MSS ("Fast Vollständig") were described by him in the Nachträge u. Verbesserungen on information supplied by Kahle . Moreover, von Gall did not consult the Samaritan scrolls at Nablus, among them the famous Abisha Scroll 7. The latter has been rarely shown to visitors. Another MS is substituted. Such seems to have happened in the case of a photograph of a Nablus scroll, affirmed to be the oldest in the Samaritan community, and in the possession of Professor John Garstang. A portion, Leviticus 152- 1618 was made available to me and collated with the von Gall text. See Appendix B . In this passage, at least, as the collation shows, the text of this MS contains a larger proportion of obvious scribal errors than the von Gall text. Presumably this is the photograph referred to by Kahle in his article on the Abisha Scroll in the Pedersen Festschrift⁸.

The/

The specimen contained the text of Leviticus 15 and 16 and was judged by A. E. Cowley to come from a MS written about the 13th century. Another important monograph on the Abisha Scroll is that by E. Robertson in 1958⁹. Both Kahle and Robertson refer to the fact that Professor Perez Castro has been successful in obtaining a photograph of the complete Scroll and that the publication of the text should make a decision possible on the worth of this famous MS. Castro has now published his results ¹⁰ and a very full account of his work is given by Robertson in a review in Vetus Testamentum See above, p. 27-8. It appears to be quite clear that the Scroll has little value as a witness to the early text of the Pentateuch and accordingly there is no reason to believe that the von Gall text has been displaced as the standard authority on the Samaritan Hebrew Pentateuch.

With these reservations the von Gall text has been accepted as the basis of comparison with the MT, having in mind that it is the best known and indeed the most generally acceptable authority and, moreover, in view of the uniformity of the Samaritan textual tradition evidenced by its MSS.

The third source listed in the Preface, the Pentateuch Texts of the Dead Sea Scrolls in <u>Discoveries in the Judaean</u>

<u>Desert</u> 12, is considered by the great majority of scholars to belong to the first century A.D. or earlier. These Pentateuch fragments/

fragments are of great importance in themselves and for the light they may shed on the comparison between the MT and the Samaritan Pentateuch. They bring in a completely new witness whose testimony may be of decisive importance in assessing the worth that is to be assigned to the Samaritan Hebrew Pentateuch.

Footnotes, Chapter II.

- 1. R.H.Pfeiffer, Introduction to the Old Testament. New York, 1948. p. 101.
- 2. JTS, 'Two Sam MSS in the Library of Queen's College, Cambridge', xxxvi, 1935, p. 142.
- 3. Von Gall, op. cit., p. LXVIII
- 4. Kahle, The Cairo Geniza, p. 149.
- 5. Catalogue of the Samaritan MSS in the John Rylands Library, Manchester, 1938.
- 6. Von Gall, op.cit., p. xv.
- 7. Kahle, Cairo Geniza, p.50, footnote.
 - Thomson, The Samaritans. Appendix II, p. 396.
 - Robertson, Bulletin of the John Rylands Library, 1935.
 Notes and Extracts from the Semitic MSS in the
 John Rylands Library. I Concerning the Abisha
 Scroll.
 - Gaster, Schweich Lectures, 1923. The Samaritans, London, 1925, pp.109 f.
- 8. 'The Abisha Scroll of the Samaritans' in Studia Orientalia
 Ioani Pedersen dicata, Hanniae, 1953, pp.188-192
- 9. 'The Ancient Scroll of the Samaritan' in Gaster Centenary Publication, Ed. B. Shindler, London, 1958
- 10. F. Perez Castro, Séfer Abisa. Edicion del fragmento antiquo del rollo sagredo del Pentatenco Hebreo Samaritano de Nablus. Madrid. 1959.
- 11. XII 1962, p. 228.
- 12. I. Qumran Cave I. Oxford, 1955.
 - II. Les Grottes de Murabba'at. Oxford, 1961.
 - III. Les 'Petites Grottes' de Qumran. Oxford, 1962.

CHAPTER III.

Preliminary Considerations affecting

Comparative Study of the Texts.

Before proceeding with the comparison of the MT and the Samaritan Pentateuch and the classification of the data there are certain preliminary considerations that arise.

There are three historical facts that have a bearing on the textual studies that are to follow:

- 1. Prior to the Council of Jamnia there was a period of variant texts.
- 2. There was oral as well as written textual transmission.
- 3. MSS occurred in both Scroll and Codex form.

Only when the significance and relevance of these facts are appreciated is it possible to proceed with the comparative study of the two texts.

1. Pre-Masoretic Hebrew Texts.

The complete MT in its present form is an eighth century production and its earliest MS of the whole OT belongs to 1008 A.D. which is the date of the MS forming the basis of Kittel's Biblia Hebraica III. Yet while this is so, it represents a text very much earlier, a pre-Masoretic and probably Jerusalem tradition. It is significant that all the Biblical/

Biblical texts from Murrabba'at (135 A.D.) show close affinities with the MT. Yet at Qumran, 60 years earlier, a different position is disclosed. The Biblical MSS there reveal variant text traditions. One is a tradition virtually identical with that of our present MT and is represented e.g. by the Hebrew University MS of Isaiah. In addition. there are traces of genuine alternative traditions. There are MS fragments revealing a text closely resembling that of the Samaritan Pentateuch, with the additions and transpositions so typical of it. There are also Hebrew fragments showing strong affinities with the Hebrew recension presupposed by the LXX2. Accordingly, one must conclude that the community at Qumran seemed to have considerable liberty of choice and was not limited to any one textual tradition. These three traditions can' be observed and probably some of a mixed type. Thus prior to the Council of Jamnia convened between 90 and 100 A.D. there was a period of variant texts. Thereafter a standard text was more or less fixed and preserved with relatively few variations.

The method of fixing this text is referred to in Jewish tradition. The Jerusalem Talmud Ta'anith iv. 2 relates that, in order to determine the right text, three temple MSS were consulted and the reading represented by two of them was always preferred to that found in only one. This/

This tradition is not to be lightly dismissed for it may contain real historical fact. The Rabbis would not evolve a completely new text but selected a tradition already in existence. They preserved errors in this tradition that preceded the MT. They did not collate other MSS to correct them but to bring them into line with that existing tradition. Even what was obviously wrong was preserved and a Q^ere was added.

2. Oral Transmissions.

The fact that the transmission of the text was not only written but oral may well explain many of the variants that have arisen between the Samaritan and Masoretic Pentateuch.

It is clear that, in the case of a written text that was the standard norm, schools of scribes would endeavour to make exact copies. Ideally, the copy and its exemplar would read exactly the same but even here there was considerable possibility of error for in the early stages of the language the vocalisation was ora!. A consonantal text must always contain passages more or less doubtful in meaning so that the possibility of mistake arises. In a document copied by hand all sorts of errors of eye and ear and mental association can become operative. In An Aid to the Textual Amendment of the Old Testament, James Kennedy classifies no less than sixteen possibilities of departure from the original text such as Interchange, Transposition, Omission and Addition of letters and/

and words³. His full treatment of these clearly shows the pitfalls that beset the scribe!

In the case of oral transmission the possibility of error is even greater, when a man recites a text that he remembers by heart without reference to any written archetype. Provided he recites the meaning it is not necessary for every word to be exactly the same. It is evident that many examples of spelling and accidence which make no difference to the meaning would not be differentiated in an oral transmission. Further, a man might recite a tradition slightly incorrectly. Should such a tradition be subsequently committed to writing, variation would arise between it and the written standard Further, it may well have been that copies of the Law circulated privately and were not treated with such care as the official standard law books. Oral expansions in the teaching of the schools might well have been incorporated in these private copies. There is no direct evidence of this at an early date but a hypothesis such as this would explain the probable expansions, the smoothing out of difficulties or the correction of eccentricities such as are to be found in the Samaritan text. In this way the Samaritans would be incorporating their teaching into their Pentateuch.

3. Scroll and Codex.

A topic that is now to be considered is a theory put forward/

XIth: Century, that the Scroll and not the Codex should be used as the basis of comparison between the MT and the Samaritan Pentateuch. A scroll fragment, reaching from Deuteronomy 27¹ to the end of the Pentateuch, had come into his possession.

Now by a fortunate circumstance it is possible to test the validity of Gaster's view. There is material that was not available to him in von Gall's text, based upon forty more or less complete MSS, together with thirty groups of fragments of MS scrolls and five groups of fragments of MS codices. Among the many MSS that underlie the von Gall text there are, six scroll fragments containing most of the text of Deuteronomy 27^{1} - 34^{12} . One of them is Gaster's MS, designated Y by von Gall.

Using the critical apparatus it is possible to demonstrate that these six MSS show variation from each other in a way that disproves the hypothesis of Gaster that the Scroll was written with extreme care and according to a rigid tradition.

There are, for example, deletions and corrections. In Deuteronomy 28⁵ five of the Scroll MSS (including that of Gaster) read $7^{\prime\prime}_{-1}^{\prime\prime}_$

deleted, read יאראיך

Mistakes occur with gutturals Exx: Addition: 28^{35} , Teads 0.767 (others 0.767). Transposition: 28^{39} μ reads 0.767 (meaningless, as there is no root 0.767) while others read 0.767 with MT (0.767) occurs here and twice in Proverbs).

Additional | ("and") occurs 27^{12} mby read and $\delta'\theta\Lambda$ 7723

examples that the Scroll MSS are not unanimous in their witness to the text. The same kind of variants occur in the Scroll as in the Codex, and they appear to be just as numerous.

There is no relation between the Scroll MSS such as would lead us/

us to infer a common tradition superior to that of the codex.

Neither is there any justification for the theory that the scroll represents a different tradition of text from that of the codex. Gaster endeavours to show this by a comparison of the Samaritan scroll, Samaritan codex and the MT, and he claims to show that the readings of the scroll differ from those of the codex and lie nearer to the MT. conclusions are unsound by reason of the fact that the readings he ascribes to the codex cannot be maintained in the light of the von Gall texts: the text of the Samaritan codex is generally different from that given by Gaster (only a few MSS have his readings). In most pages of the von Gall volume the text rests upon about two dozen Samaritan MSS. the majority of which are codices. The testimony of these MSS as a whole to the codex is quite different from that given by Gaster. Usually the readings of Gaster can be found in a few MSS. say three or four, very rarely the same three or four. evidence is overwhelming and only a few examples need be given.

	$\underline{\mathbf{MT}}$.	Sam.Scroll	Sam.Codex
28 ²⁵	ובשבלה דרכים	'T '1	/T Y26 21
28 ²⁷	02021	′,	(only a few MSS)
28	ης ης. " τ "	l	ובהים (only 3 out of 27 MSS)
28 ³³	コックママ シラ	c' 'n'	(only 4 out of 27 MSS)

To sum up the evidence; in this portion of the Pentateuch, where the evidence of MSS in Scroll and Codex is available, there is no reason to conclude that the text of the scroll has been more accurately transmitted or that it represents a different tradition from that of the codex.

Footnotes, Chapter III.

- 1. J.T.Milik, Ten Years of Discovery in the Judaean Wilderness and their Importance. London, 1959, p.140.
- 2. F.M.Cross, Jr., The Ancient Library of Qumran.
 London, 1958, pp.131-38.
- 3. Edinburgh, 1928, pp. 4-187.
- 4. London, 1901, p.25 (242)

CHAPTER IV.

Principles of the Classification.

Before proceeding with the actual classification of the Samaritan variants from the MT it is desirable to state certain general principles and to give an outline of the proposed classification itself.

The Classification proposed is aimed at being objective, comprehensive and convenient; objective, by taking the facts as they are and thus endeavouring to avoid a subjective analysis such as that based on a theoretical distinction between intentional and unintentional alteration; comprehensive, by giving both a broadly classified as well as a detailed examination of the material; convenient, by providing a clear-cut division of the material for the purpose of easy reference.

In the Classification which follows the Samaritan variants are divided into three main divisions: Orthography, Grammar and Vocabulary.

Orthography is here used in the sense of two different spellings of the same word. There are also included under this division examples such as $\gamma^{3\gamma}$ and $\tau^{3\gamma}$, showing occurrence of two different but related consonants. These are treated here for purpose of easy reference, though a case might be argued/

argued for their inclusion under difference of Vocabulary.

The number of orthographic differences is very large and there are not only differences of orthographic usage between different Samaritan MSS but sometimes within the one Samaritan MS a variant orthography is found once, or in a few examples, or continued throughout the Pentateuch. Accordingly, exact enumeration is impossible. What will be done is to distinguish certain main groups and elucidate tendencies and endeavour to determine their relevance and significance.

The second main division, Grammar, consists of variants showing inflected forms and derived forms from the same root in both Samaritan and MT. A few words showing transposition of two consonants are included here rather than under Vocabulary.

In this division a new factor emerges as some of the variants show difference of meaning as well as form. Thus variation under this heading is now divided into those with agreement in sense and those with divergence in sense. Forms agreeing in sense and found elsewhere in the Samaritan Pentateuch are distinguished from those not found elsewhere in the Samaritan Pentateuch. There is still further sub-division such as early and late forms, common and rare forms, forms found elsewhere in context or in parallel passages.

Grammatical variants showing difference of meaning have to be classified quite differently and here a distinction is/

is made between doctrinal and non-doctrinal variation.

Doctrinal variants are considered in relation to the known doctrinal background of Judaism or Samaritanism. In the case of some of the non-doctrinal variants a comparison is made with interpretations elsewhere, inside or outside the Samaritan Pentateuch.

The third main division consists of variants comprising differences of Vocabulary, in the sense that a word from a different root occurs in Samaritan.

Here, as under Grammar, there are variants that involve difference of meaning as well as form. Accordingly, there are two sub-divisions into synonyms of various kinds and variants showing difference of sense, some of which show complete divergence in meaning. The same sub-divisions of these are used as in the case of the grammatical groups.

Outline of Classification.

- I. Differences of Orthography.
 - 1. Variants attributed to scribal error.
 - i. Letters similar in sound.
 - (1) Gutturals.
 - (2) Palatals
 - (3) Sibilants.
 - (4) Labials.
 - ii. Letters similar in shape.
 - iii. Other typical scribal errors.
 - iv. Formae Mixtae.
 - v. Proper Names.
 - vi. Passages too corrupt in both ST and MT for any useful comparison.
 - 2. Variants making the pronunciation explicit.
 - i. Use of Aloph
 - ii. Use of He
 - iii. Use of Yodh and Waw.
 - (1) Greater use of vowel letters as compared with MT.
 - (2) Possible orthographic retention of consonantal Yodh.
 - 3. Variants normalizing the spelling of words.
 - i. Archaisms.
 - ii. Errors.
 - ili. Inconsistencies in identical or similar context.
 - 4. Variants in writing of Samaritan MSS.
- III. Differences of Grammar.
 - 1. With agreement in sense.
 - 1. Accidence.

- (1) Variants consisting of forms found elsewhere in Samaritan Pentateuch.
 - (i) Normal forms.
 - a. Common for rare forms.
 - b. Forms suggesting misreading, miswriting or mishearing in ST or MT.
 - c. Forms in identical or similar context.
 - d. Forms which do not belong to any of the above types.
 - (ii) Abnormal forms.
 - a. Early.
 - b. Late
 - c. Forms suggesting misreading, miswriting or mishearing in ST or MT.
 - d. Forms in identical or similar context.
 - e. Forms which do not belong to any of the above types.
- (2) Variants consisting of forms not found elsewhere in Samaritan Pentateuch.
 - (i) In MT of Hebrew Bible:
 - a. In Pentateuch.
 - b. Outside Pentateuch.
 - (ii) In Dead Sea Scrolls
- (iii) In both (i) and (ii)
 - (iv) In no other known source.

ii. Syntax.

- (1) Variants consisting of constructions found elsewhere in Samaritan Pentateuch.
 - (i) Normal constructions
 - a. Common for rare constructions
 - b. Constructions suggesting misreading, miswriting, or mishearing in ST or MT.
 - c. Constructions in identical or similar context.
 - d. Constructions which do not belong to any of the above types.

- (ii) Abnormal constructions.
 - a. Early
 - b. Late
 - c. Constructions suggesting misreading, miswriting or mishearing inST or MT.
 - d. Constructions in identical or similar context.
 - e. Constructions which do not belong to any of the above types.
- (2) Variants consisting of constructions not found elsewhere in Samaritan Pontateuch.
 - (i) In MT of Hebrew Bible:
 - a. In Pentateuch.
 - b. Outside Pentateuch.
 - (ii) In Dead Sea Scrolls.
 - (iii) In both (i) and (ii).
 - (iv) In no other known source.

2. With divergence in sense.

- i. Doctrinal variants.
 - (1) Controversy regarding Gerizim.
 - (2) Other Samaritan beliefs and practices.
- ii. Non-doctrinal variants.
 - (1) Variants where there appear to be cogent reasons in favour of ST.
 - (2) Variants where there appear to be cogent reasons in favour of MT.
 - (3) Variants where there appear to be no cogent reasons in favour of ST or MT.

TII. Differences of Vocabulary.

- 1. With agreement in sense.
 - 1. Substitutions.
 - (1) Pure Synonyms.
 - (1) Common words.
 - a. Common for rare words.
 - b. Words suggesting misreading, miswriting or mishearing in ST or MT.

- c. Words in identical or similar context.
- .d. Words which do not belong to any of the above types.
- (ii) Uncommon words.
 - a. Early
 - b. Late
 - c. Words suggesting misreading, miswriting or mishearing in ST or MT.
 - d. Words in identical or similar context.
 - e. Words which do not belong to any of the above types.
- (2) Synonyms giving greater precision or emphasis.
 - (i) Common words.
 - a. Common for rare words.
 - b. Words suggesting misreading, miswriting or mishearing in ST or MT.
 - c. Words in identical or similar context.
 - d. Words which do not belong to any of the above types.
 - (ii) Uncommon words.
 - a. Early
 - b. Late.
 - c. Words suggesting misreading, miswriting or mishearing in ST. or MT.
 - d. Words in identical or similar context.
 - e. Words which do not belong to any of the above types.

ii. Additions.

- (1) Pure Synonyms.
 - (1) Common words.
 - a. Stylistic supplements in ST.
 - b. Words suggesting misreading, miswriting or mishearing in ST or MT.
 - c. Words in identical or similar context.
 - d. Vords which do not belong to any of the above types.

- (ii) Uncommon words.
 - a. Early
 - b. Late
 - o. Words suggesting misreading, miswriting or mishearing in ST or MT.
 - d. Words in identical or similar context.
 - e. Words which do not belong to any of the above types.

(2) Synonyms giving greater precision or emphasis.

- (i) Common words.
 - a. Record of events implied but not explicitly recorded in MT.
 - b. Words suggesting misreading, miswriting or mishearing in ST or MT.
 - c. Words in identical or similar context.
 - d. Words which do not belong to any of the above types.
- (11) Uncommon words.
 - a. Early
 - b. Late
 - c. Words suggesting misreading, miswriting or mishearing in ST or MT.
 - d. Words in identical or similar context.
 - e. Words which do not belong to any of the above types.

iii. Omissions.

- (1) Pure Synonyms.
 - (i) Common words.
 - a. Stylistic supplements in MT.
 - b. Words suggesting misreading, miswriting or mishearing in ST or MT.
 - c. Words omitted in identical or similar context.
 - d. Words which do not belong to any of the above types.

- (11) Uncommon words.
 - a. Early
 - b. Late
 - c. Words suggesting misreading, miswriting or mishearing in ST or MT.
 - d. Words omitted in identical or similar context.
 - e. Words which do not belong to any of the above types.
- (2) Synonyms giving greater precision or emphasis.
 - (i) Common words.
 - a. Record of events implied but not explicitly recorded in ST.
 - b. Words suggesting misreading, miswriting or mishearing in ST or MT.
 - c. Words omitted in identical or similar context.
 - d. Words which do not belong to any of the above types.
 - (ii) Uncommon words.
 - a. Early
 - b. Late
 - c. Words suggesting misreading, miswriting or mishearing in ST or MT.
 - d. Words omitted in identical or similar context.
 - e. Words which do not belong to any of the above types.
- 2. With divergence in sense.
 - i. Doctrinal variants.
 - (1) Controversy regarding Gerizim.
 - (2) Other Samaritan beliefs and practices.
- ii. Non-doctrinal variants.
 - (1) Variants where there appear to be cogent reasons in favour of ST.
 - (2) Variants where there appear to be cogent reasons in favour of MT.
 - (3) Variants where there appear to be no cogent reasons in favour of ST or NT.

The above scheme of classification has been drawn up in the light of the examination of the Samaritan variants and appears to be adapted to the material. However, in the presentation of this material, in the following Chapter and in Appendix A, a sub-division is omitted when there are no variants under its heading.

CHAPTER V.

Classification of the Samaritan Variants from the Masoretic Text.

As stated in the Preface, the Samaritan variants are collated with the Pentateuchal texts of the Dead Sea Scrolls in Discoveries in the Judaean Desert, I, II and III (designated DSS I, II and III) when the MSS fragments from Qumran and Murabba'at contain the passage in question.

The Samaritan variants that consist of difference in proper names or show difference of sense are also collated with the Septuagint (designated LXX).

T. Differences of Orthography.

transmission/

Although in the examples given the ST and MT orthographies are grouped together, this is not to be taken as implying that these variants were pronounced alike. Petermann¹, Diening² and others have dealt with the Samaritan pronunciation of Hebrew and Kahle³, Sperber¹ and others have shown how early transcriptions imply different pronunciation from that of pointed MT ("Tiberian").

1. Variants attributable to Scribal Error.

Some of the variants between ST and MT are due to the typical errors which arise in the course of manuscript

transmission: there are the unconscious mechanical mistakes on the part of copyists, such as errors in reading and writing. The resultant text may be quite good Hebrew and will be considered elsewhere in the Thesis but there are other cases where the text as it stands, in ST or MT or both, is doubtful or corrupt and these will be grouped here.

i. Letters similar in sound.

(1) Gutturals.

These variants show omission, addition, interchange, or transposition of gutturals and this is not unexpected in view of the well known Samaritan practice of not distinguishing the gutturals (or pronouncing them all as aleph) in the reading of Hebrew. Cf. Petermann⁵, Kahle⁶. A number of examples will now be given.

(1) Omission

(li)Addition

The reading adopted by von Gall presumably implies 3 s.f. pf. Aphel of $\sqrt{73}$ y which in form and meaning/

meaning is difficult to accept. A3AY read by 9 MSS and involving transposition of the gutturals, which could easily occur, may be a late orthographical variant to the MT pt. form from $\sqrt{3}AY$ showing a vocalic letter other than Yodh or Waw representing a vowel within a word. See discussion by Petermann, Formenlehre, pp.124-5.

(111) Interchange.

In Gn. $30^{38,41}$ some Samaritan MSS agree with the MT in reading $p' \nu \pi \gamma \pi$

(iv) Transposition.

In addition to the foregoing examples, some of which are adopted by von Gall and all of which are strongly attested in von Gall's critical apparatus, there are many others where the evidence is/

is not so strong, e.g.

(i) Interchange of guttural

(ii) Additional guttural sometimes with metathesis

$$Gn.42^{21}$$
 ST און הצרעה הזאת (11 MSS)
ST הצערה ה' (3 MSS)
 R

(iii) More than two gutturals.

These examples, representative of a much larger corpus of material, show that the gutturals were a fruitful cause of scribal error among the Samaritans.

(2) Palatals.

Homorganic consonants, such as the palatals 2 and > sometimes interchange. There are only a few examples of this but they are nevertheless noted as representative of a type of variation, e.g.

(3) Sibilants.

In a somewhat similar way there are a few cases where the sibilants 5 and 1 interchange, e.g.

(4) Labials.

Likewise the labials 2 and 5 may also interchange and there are some examples of this type, e.g.

ii. Letters similar in shape.

The interchange of letters which look alike is a common cause of error. Such errors could, and did, occur between resemblant letters in the Samaritan script and in the square script of the MT.

(Joseph's brethzen remind him that they had brought back the money found in their sacks). The Samaritan reading shows an additional waw in the preformative syllable of the Hiphil inconsistent with the forms of ("y verbs. Thus Samaritan would imply a $\sqrt{2}$ " which is quite meaningless in the context. Further, as/

as 7 and 1 are very similar in Samaritan script, the 1 could easily arise by dittography.

Square script: and con. 10²⁷ ST Griv (Many MSS, also LXX)

MT Griv (Agreeing with Ichron. 1²¹)

7 and 7

Gn. 36³⁹ ST 77 (Many MSS, also 1chron. 1⁵⁰)

MT 277

iii. Other typical scribal errors, such as incorrect joining or dividing of words.

In the Samaritan MSS it was customary to find a dot between each word and in the square script a small gap. Sometimes the dot could disappear or the gap become so small that words might be wrongly joined together.

iv. Formae Mixtae.

This special variety of textual error, not an accidental miscopying but the use of a form conveying two different readings, is evidenced in ST as well as in

v. Proper Names, attributable to scribal error and classifiable under the variant types of this first section of orthographical variants, are grouped together under this heading for convenience of reference.

The complete list is given in Appendix A.

vi. Passages too corrupt in both ST and MT for any useful comparison.

There is a small group of passages that I am not prepared to classify. These are corrupted passages, noted in Appendix A, where the ST and MT differ from each other and both appear to have such a difficult, or even meaningless, reading that it is reasonable to assume that in both ST and MT there is ancient textual corruption that antedates their divergence from each other.

2. Variants making the pronunciation explicit.

Originally Hebrew orthography had no vowel letters and Hebrew and its cognate languages had consonantal scripts. Yet at quite an early period, for example in Hebrew and Aramaic inscriptions, consonants were used to represent vowel sounds and so to indicate the pronunciation of the text./

text. An explanation of the rise of these matres lectionis is given in Gesenius-Kantzsch, Hebrew Grammar, 2nd Eng. ed.,

7b. Both ST and MT Pentateuchs use vowel letters to make the pronunciation of the text explicit: these vowel letters are

77 / which will now be considered in order

i. Use of Aleph.

(1) Elision (Syncope)

In ST, as in MT, there are examples of an <u>aleph</u> being omitted which had lost its consonantal value.

In these examples this omission of <u>aleph</u> is similar to that of Aramaic orthography, where <u>aleph</u> between two vowels was pronounced <u>yodh</u>⁸.

Proper Names, showing elision of aleph are kept as a separate group and are listed in Appendix A.

(2) <u>Vowel Letter</u>

(i) Medial

In the MT <u>aleph</u> is occasionally used to indicate a long "a", as in Hosea $10^{14}\,\text{p/p}$ for p/p. This usage occurs more frequently in the Samaritan

Cf. Petermann, Formenlehre, p.205, "Lafret".

(ii) Final

In ST, as in MT, aleph can represent a final vowel, sometimes interchanging with he of the MT. This is frequently the case with verbs

Lamedh He which in ST sometimes show orthographies similar to those of verbs Lamedh Aleph.

(3) Final (prosthetic) after long vowel

Aleph (prosthetic) may or may not occur as an orthographic closing to a final long vowel

ii. Use of He

To indicate presence of final vowel.

Both Samaritan and Masoretic Text agree in using 7 frequently as the orthographic representation of a final vowel. There are also certain readings where the final vowel is not orthographically represented in this way. As both orthographies were known to the scribes there are readings where both versions agree/

agree in reading the final , or in not reading it, or where one reads it and the other does not. There are many examples of this type of orthographic variation which arose from the interchange of two orthographies.

In ST

(2) <u>Suffix</u>, 2 s.m.

Gn.19²² ST 75/2

MT 7/2

(3) <u>Verb.</u> Pf. 2 s.m. Ex.25¹⁶ ST 7AAJI

MT AAII

Pf. 3 s.f.

Lv.25²¹ ST 7/4 WY I

Impf. 2 pl.f.

Ex.1¹⁸ ST מיז את היצדים (von Gall)

ST מ'ז א ה'ידן א' ה'

ST מ'ז א היינה (1 MS)

MT מ'ז א היינה (1 MS)

This is a rare form and only one variant has been found.

found.

Impf. 3 pl.f

$$Gn.19^{33}$$
 ST יין א א אביקן יין (von Gall) ST 'י א א' א' יין (9 MSS) MT '' א' א' יין MT '' א' א' יין MT

In this example, and in all others where **i**s found, von Gall reads the usual form in 11-9 Samaritan MSS, however, read |πρωλι and for ן there is a certain amount of this ending in MS evidence elsewhere e.g. Gn. 1935 אשקיהן (11 MSS), 30³⁸ (4 MSS). For full details see Appendix A. In so far as this is not a textual error showing transposition of letters, one of which is a guttural, it might be conjectured that this orthography shows 17 an afformative as an alternative to the usual אנה, both connected with 3 pl.f.suffix, אוה. , both connected with 3 pl.f.suffix, אוה. (after prefixes, [7,]7). In MT

(1) <u>Pronoun</u>, 2 pl.f

Gn.31⁶ ST | ANI

MT JJANI

(2) <u>Suffix</u>, 2 s.m.

| Gn.27⁷ | ST | קסיפר | MT | אברככה | MT | Gn.27¹ | ST | Care | Ca

(3) <u>Verb.</u> Pf. 2.sm.

Gn. 31³⁰ ST ADDD1

MT AASO DI

iii. Use of Yodh and Waw.

(1) Greater use of vowel letters as compared with MT. In marked contrast to the MT the Samaritan Pentateuch is abundantly supplied with the vocalic consonants. Obviously the usage is intentional Yodh and Waw. and the purpose is to make the pronunciation explicit. Little purpose would be served by attempting an exhaustive classification of all the examples of plane writing in the Samaritan Pentateuch. Millar Burrows stated regarding the St. Mark's Isaiah Scroll: "One gets the impression that if the scribe could think of making a word longer by extra letters. he did it. "9. This may also be true to some extent of the Samaritan orthography and renders complete classification of this immense group well nigh impossible. At the same time certain facts and tendencies can be discovered and these will be duly noted and an endeavour made to determine their relevance and significance.

(i) General observations.

Before enumerating certain selected classes of orthographical variation in this group, there are/

are three general observations to be made.

- a. As already stated, see above, p. 54, though

 ST and MT are grouped together, this is not

 to be considered as implying that these variants

 were pronounced alike. This appears to be borne

 out by the ST often having a scriptic plane of

 Yodh or Waw not homogeneous with the pronunciation

 of the MT.
- b. In ST there are instances of a vocalic consonant where MT has a short vowel.

The Samaritan may show a later orthography than MT or, conceivably, a plane orthography of a different pronunciation.

c. In ST there are instances of a vocalic consonant where MT has vocal shewa.

Once/

Once again, the Samaritans may show a later orthography than MT or, conceivably, a plene orthography of a different pronunciation. The Samaritans may have maintained the fuller pronunciation of the vowel. Cf.Petermann, Formenlehre, p.10.

A statement of the evidence on these three topics is given in Appendix A.

(ii) Selected Classes.

a. Nouns.

(a) Segholate forms.

Ex. 29²⁰ ST | 172

Such Samaritan segholate variants, with the characteristic vowel in the second syllable and not the first, do not stand alone. Similar forms are to be found in the Qumran MSS as well as in the second column of the Hexapla, in LXX, and in transcriptions from Hebrew by Jerome 10.

They reveal a pronunciation distinctly different from that adopted by the Masoretes.

(b) Plural termination.

No examples given here. A full statement of the many Scripture references for plene and/

and defective spelling of masculine and feminine forms, absolute, construct and with suffixes, is given in Appendix A. It shows an overwhelming preponderance of scriptio plend.

b. Pronouns.

Suffixes, 2 s.f.

c. Verbs.

.: ,[‡], .

Appendix A gives references for scription plena/defectiva of the following:

- (a) Verbal ending before suffix
- (b) Participle, first stem syllable
- (c) Infinite Absolute, second stem syllable
- (d) Niphal, second stem syllable
- (e) Hiphil, second stem syllable

Thereafter, scriptio plena/defectiva for the following weak verbs:

- (f)(g)(h) Initial Medial and Final Yodh
- (i)(j) Initial and Medial Naw

d. Particles.

- (a) AN (Acc) ST always has scriptio defectiva
- (b) | ST always has suffixes indicating the singular.
- e. Proper Names, involving scriptio plena or defectiva/

defective of Yodh or Waw, are grouped together.

(111) Remaining Variants, a large corpus of material, are arranged under grammatical classes of nouns, adjectives, verbs, and particles, for convenience of reference. The foregoing selected classes just considered have been extracted from the immense mass of Samaritan variants whose orthography shows a much greater use of the letters, yodh and waw, than that of the MT. No attempt has been made at a complete classifigation of the whole group, apart from the grammatical arrangement of the residuum into nouns, adjectives, verbs and other parts of speech. Even if a meaningful classification were possible, it hardly appears to be necessary for the significance of this profuse use of matres lectionis is not really in the details but in the group as a whole. The question arises as to whether this abundance of vocalic consonants is an early or late usage. One consideration that immediately suggests itself is that the whole purpose of plene writing is to make the pronunciation and meaning explicit. This would not be of

great/

great value unless Hebrew were still understood by the mass of the people - scholars would know it all already. Accordingly, a relatively early date might be argued for, before Hebrew was displaced by the popular Aramaic. As distinct from this possibility there is another line of evidence that might indicate dating. According to Cross and Friedman "the fullest development of plene writing was achieved during the Maccabean era"i.e. circa 100 B.C. 11 It has been suggested by Kahle that this period of nationalist revival stimulated a renewed interest in Hebrew and that to facilitate reading among the people there was a greater use of the matres lectionis, which of course was possible as the consonantal text was not yet fixed. A different view is that there was a gradual increase in the use of scriptio plena for some centuries prior to the establishment of the standard text which eliminated many of the vowel letters on the evidence of MS available. In any case it might be argued that the full orthography of the Samaritan Pentateuch represents a type of text prior to this standardisation.

The evidence of the Pentateuchal fragments in Discoveries in the Judaean Desert I. II and III gives strong support to this view. Some of them have an orthography practically identical with that of the MT, while others witness to different traditions including one very similar to that of the Samaritan with a frequent use of the vocalic consonants, yodh and waw. See further. Chapter VI, Evidence of Dead Sea Scrolls. It may be added that the St. Mark's Isaiah Scroll. which has been frequently consulted but not collated, shows the orthography at a still more developed stage than that of the Samaritan. this, for example, there is a veritable abundance of vocalic consonants and it would appear as if the Scribe was inserting them much more than was customary, in fact in a lavish and unsystematic way. A comparatively rare use in ST or MT, such as 2 s.m. pronominal suffix in 75 becomes quite general. A further example is the use of metathetic aleph in such forms as N.D. N.D. N'IN which occur often but not invariably. Regarding the date of the St. Mark's Isaiah Scroll it would appear hardly possible that such a MS could have been written at all in an orthodox community/

community in Jerusalem after the standardisation of the consonantal text.

(2) Possible orthographic retention of consonantal yodh

(1) Lamedh He verbs and nouns.

It is to be noted that these orthographies are taken from the critical apparatus of von Gall and not from his actual text which, in certain aspects, as he explains in his preface,

has been harmonised with the Masoretic tradition.

In some cases only a few Samaritan MSS witness
to these readings but they are not to be rejected
without examination.

The fact that this usage is only occasional may mean that it is the survival of an earlier usage or the arrival of a new one. That one and the same MS can vary in the usage is clear from the phrase $(700 \, a\cdot n\omega)$ ip $(Gn.14^{19},22)$ where the von Gall critical apparatus clearly shows that a MS can pass from one orthography to the other.

Hypothetically, this use of final <u>yodh</u> might be explained/

explained in two quite different ways:

Either an archaic spelling preserving the original third radical. or

a late phonetic spelling with <u>vodh</u> as the bearer of the vowel <u>cere</u>.

Further evidence is needed before a decision can

be made between the two possibilities. Evidence is furnished, to some extent at least, by the regular use of 17 in the Samaritan Pentateuch corresponding to MT $\pi_{i}^{*}\pi_{j}^{*}$, a clear Aramaism on the part of the Samaritan. On the other hand in Gn.28⁸ almost all the von Gall MSS read f.s. pt. ויה clearly showing the retention of the third radical as sometimes happene in f.pt.forms in MT12. Further, the St. Mark's Isaiah Scroll has the reading '15 (Is.381). This form '15 read in the Samaritan Pentateuch in Lv.242, showing that the usage was to be found in the pre-Masoretic popular texts. It would appear that the evidence is not decisive either way and it may well be that the two usages, early and late. existed alongside each other.

(ii) Prepositions 3N and 3Y with suffixes.

These Samaritan forms may show 3% and 3% resuming before suffixes the Yodh which originally formed part of the root. To this extent the ST preserves the earlier form in Yodh.

The usage is that the ST always reads the plene orthography while the MT usually reads the plene form but not invariably.

As in the preceding section (Lamedh He verbs and nouns), this yodh before the suffix might be the archaic spelling or possibly a late phonetic spelling with yodh as the bearer of cere.

3. Variants normalizing the spelling of words.

i. Archaisms.

3 s.m.suffix: ST.has normal form where MT reads \vec{n} —

Gn.9²¹ ST $i\vec{\beta}\vec{n}$ $7i\lambda$ 2

MT $\pi \hat{\beta}\vec{n}$ n

There are 14 examples in the Pontateuch of the MT reading the unusual (and generally supposed archaic) orthography in 7 in the 3 s.m.suffix. 13. In no case does the ST read this form. "Das Suffix 3 p.s.m. hat der sam. Text nicht, sondern dafür stets 1 " 14. In 9 examples, noted in Appendix A, the ST reads the customary form in 7 . In the remaining 5 examples the Samaritans either interpret the word differently e.g. Ex. 32 7 7772 or read another form e.g. Nu. 10 36 70 11021 for MT 7772

ii. Errors.

Personal Pronoun Nin as common gender. There are only eleven examples of the MT using the f. ~ 1.7 in the Pentateuch 15: elsewhere the pronoun Qere perpetuum NIN occurs. In every case the ST היא It has been argued that this reads orthography is an archaism preserved in almost every case by the MT. It has also been argued against this that the epicene use is not found in any other Semitic language. The distinction too does occur in the Pentateuch, and outside it in the older books. the view of the present writer that for some reason, now unknown, Nid was written almost invariably for Nid in the Pentateuch and this orthographical peculiarity, or error, was faithfully repeated.

iii. Inconsistencies in identical or similar context.

Both ST and MT agree in reading Skilb Gn. 3232

Elsewhere in both ST and MT, 174'

ST everywhere ywin.

There/

There is clearly discrepancy in the MT. The ST readings, admittedly correct, are so much easier that one hesitates simply to accept them, without asking why the MT faithfully preserved readings so evidently unusual.

4. Variants in writing of Samaritan MSS.

An outline of the main differences between ST and MT is now given.

- i. Form of the Old Hebrew script with words divided by separator dot.
- ii. Samaritan Pentateuch MSS are written in a carefully formed script generally designated majuscule; other non-Biblical MSS have a cursive or minuscule script.
- iii. Vocalisation signs used sparingly and apparently in later MSS.
 - iv. Sentences or sense-units of text marked by semi-colon, sometimes followed by a horizontal stroke.
 - v. Division of text into probably sense paragraphs or lectionary divisions.
 - vi. Use of taskil or cryptogram in body of text itself, and using some of its letters, to give such information as date and circumstances of the MS e.g., in Abisha Scroll, Dt. 6¹⁰-13¹⁹, "in the 13th year of the possession of the Children of Israel of the land of Canaan": or to give the position in the Pentateuch e.g. Lv.7¹⁵

II. Differences of Grammar.

- 1. Differences of Grammar with agreement in sonse.
 - i. Accidence.
 - (1) <u>Variants consisting of forms found elsewhere in</u> Samaritan Pentateuch.
 - (1) Normal forms
 - a. Common for rare.

There are quite a large number of Samaritan variants of a morphological nature that show the ST.reading a common form where the MT has a rarcr one. The ST reading is given first in the examples that follow:

- (a) Common form of pronoun. 73N7 for 3N7

 Gn. 19⁸ 73N7 D'WJN 3 / 3N7 '3

 Other examples in Appendix A.
- (b) Common form of suffix.

 Gn.2²¹ ה'אחא / האחא / האחא / מצודי צ'ה'ה' מצודי צ'ה' מצ'ה' מצ
- (c) Residuum unclassified further. $Gn.46^3$ A77B A77B A77B A77B A77B A77B A77B
- b. Forms suggesting misreading, miswriting or mishearing in ST or MT.

ST or MT

Another group of variants may be considered as showing evidence of misreading, miswriting or mishearing/

The resultant text may be quite good Hebrew but shows signs that at some stage typical scribal error, due for example to similarity in sound or shape of letters, has been at work. No general rule can be applied in the criticism of these variants and each case must be judged separately. Such decisions can naturally be somewhat subjective in nature but, apart from confusion with gutturals to which the Samaritans were specially liable, a broad picture emerges in which sometimes ST appears to have preserved a superior reading, in other cases MT. In the majority of cases no firm decision is possible and it is wise to leave the question open.

Dittography or haplography of A

c. Forms in identical or similar context.

There are cases where the Samaritan variant consists of a form used elsewhere by both ST and MT, in the same context (sometimes the same verse), or in a similar or parallel passage in another part of the Pentateuch.

d. Forms which do not belong to any of the above types.

This is a large group of variants consisting of ST and MT forms which are not only found elsewhere in the Pentateuch but appear to be used indiscriminately. Thus a scribe could pass from one to the other: accordingly, no cogent reason can be adduced in favour of either the ST or MT reading,

(a) Imperfects with ending in Nun

The older and fuller termination of the Imperfect in Nun is quite a common form in MT (over 300 occurrences, cf. GK, \$47.m) and appears to be used indiscriminately by both ST and MT.

$$Gn.41^{55}$$
 $| \omega y \wedge | \omega y \wedge$ $| \omega y \wedge | \omega y \wedge$ $| \omega y \wedge | \omega y \wedge$

(b) אחרי and אחרי אחרי Gn. 9²⁸ אחרי / אחרי / אחרי

(c) Forms with He paragogic.

Gn.27¹¹ 717 / /7

(d) Two forms of suffix.

(e) Residuum - unclassified further.

(11) Abnormal forms.

a. Early

These are, in the nature of the case, small in number but relatively important. The examples given show ST preserving an undoubtedly early form, with original Yodh re-appearing.

b. Late

There are, however, other forms that appear to be late, some of them similar to, if not actually influenced by, Aramaic forms.

As in Aramaic and Mishna

Aramaic Construct Plural in ST

c. Forms suggesting misreading, miswriting or mishearing in ST or MT.

There appears to be a single example of a type

This is a rare word occurring only here in Pentateuch, although both forms occur in MT.

d. Forms in identical or similar context.

The parallel is shown in brackets Gn. $3^{16} (3^{17})$ $| 12^{5}72 / 2^{5}72 |$ $| 20^{12} (18^{13})$

e. Forms which do not belong to any of the above types.

There are 15 variants consisting of forms from the Hithpa'lel of $\sqrt{10}\omega$. These show clearly the usage of ST. According to Petermann, p.85, the <u>Naw</u> was pronounced hard, as b, in all forms of this verb. They may conceivably be indicative of a dialectal usage in the Samaritan.

(2) Variants consisting of forms not found elsehwere in Samaritan Pentateuch.

Such forms, if paralleled elsewhere in MT of the Hebrew Bible or in Dead Sea Scrolls, may be regarded as receiving strong confirmation.

(1) IN MT of Hebrew Bible.

Outside Pentateuch. 2 examples.

Gn.3¹⁶ | Treft | ST form in Ru.4¹³ | Ho.9¹¹

Ex.19⁹ | $2\sqrt{2}$ | ST form in Jb.15²⁶ | 1 K 7²⁶ | 2 Ch.4⁵ | Je.52²¹ =

(ii) In no other known source. 1 example $Gn.49^{10}$ IARP' ARP'

The form in ST is not found elsewhere but may represent an old construct state ending preserved here by ST. Cf. $Gn.1^{24}$ $\zeta \gamma \wedge I \wedge \sigma I$ (MT)

2. ii. Syntax.

Α,

- (1) Variants consisting of constructions found elsewhere in Samaritan Pentateuch.
 - (i) Normal constructions.
 - a. Common for rare.

As with accidence so with syntax: there are Samaritan variants that consist of the ST reading a common for a rare construction.

ועצ השנות ה'/ ועצה שנית החצום 32 An.4132

Niphal of MT only here

Common for rare gender of subject.

c. Constructions in identical or similar context.

There are also many cases where the Samaritan variant shows a construction used elsewhere by both ST and MT in the same context or in a similar or parallel passage in another part of the Pentateuch.

The type of variant, as well as that immediately preceding, "common for rare construction", suggests the possibility of harmonisation on the part of ST in these cases and these two types, taken together, make the possibility all the stronger. Some of the variants of this class. C, admit of still further arrangement so that the material can be set out as follows:

(a) Differences of word order

The parallel is shown in brackets.

Gn. 2^4 (earlier in verse) $\xi^{\gamma N}$ $\Delta^{\gamma N}$ $\Delta^{\gamma N}$ $\Delta^{\gamma N}$ $42^{32}(42^{13})$ 13 $\beta^{\gamma N}$ $\Delta^{\gamma N}$ $\Delta^{\gamma N}$ 13 $\beta^{\gamma N}$

(b) Differences of verbal form

(c) Absence of suffix in ST or MT.

Ex. 23²⁰ (23²³) '>ν³ν η³ω '> γγ³ν 'ω 'γ וראה ה' / וראה הכהן (30,6,8,10,20) / יחוד / יוראה ל

(d) Residuum - unclassified further.

Cf. Petermann, p. 56 "Wahrscheinlich durch Versehen v.18, wo כהריתי steht".

Gn.4318 (4312) וויאר באמאחאינו (4312 המושב באמאחאינו

d. Variants which do not belong to any of the above types.

No cogent reason can be adduced in favour of ST or MY.

Under this general heading is grouped a very large number of Samaritan variants. Many of them are due to the interchange of grammatical constructions; for example, an indefinite subject may be expressed by a 3 s.m. or 3 pl. verbal form in both ST and MT. Sometimes both agree in reading the singular verbs, or the plural verb, or diverge when ST reads a singular and MY a plural. Both constructions were familiar or vice versa. to ST and MT and either might be used in a particular passage. Accordingly, when divergence occurs in the case of grammatical constructions. both of which occur without appreciable distinction in usage, either reading might conceivably be the original.

This/

This is the significant result of the hundreds of variants of this class that now follow. In all these cases the readings of ST and MT are equally acceptable and no preference should be given to either.

The material may be arranged thus:-

(a) Differences of word order

(c) Infinite absol: qualifying finite verb.

Gn. 199 ST 694 01 6941

(9 Samaritan MSS, والاستان Of.Petermann, p.18. Ptc. Act.)

as substitute for finite verb.

(d) Impf. ending in 7 -

In ST

Ex. 3⁸ וארד / וארדה

In MT

In ST Gn. 14²¹ 711 / /

In MT

השבעה / והשבע ל Gn.21²³

(f) Difference of conjugation.

Two actives

Gn. 19¹³ השחיתה / המחש3

Nu. 22²⁶ 90'1 / 901'1

Two passives
Gn.40¹⁵ 'A2121 / 'A212

Lv. 10 16 קורש / קיש

Active for passive

Gn. 426 3nn /3n17

Gn. 4519 1118 /7118

Passive for active

Gn. 5026 2011 /2011

(g) Agreement of subject and predicate.

Gn. 136 ST GNA DAN ANG 1 N31

MT 7 W NW1 1

ורחץ ממנו אהרן ובנין ST אחנו אהרן ורחצו מ' א' ו

(h) Transitions in personal reference.

from singular to plural and vice versa

10x. 117 YTA / 11774

Dt. 125 DAMAI / AMAI

(i) Feminine termination.

Gn. 6¹⁹ החיה 3 החיה Abstract Nouns.

(1) (contd.)

Gn. 95 'n 35 / 7'n 's

(j) Plural (or dial) termination.

Gn.15¹⁰ arrest / Star Collective in MT In MT

Gn. 411 7'00 D7 / 7' Intensive plural in MT

(k) Suffixes:

Difference in person or number

Gn. 6^{20} Dn. 1.63 / Inj.63 Collective subject Ex. 22²⁵ 7/2'\u0300 / 1/2'\u0300 Fem. subject preceding

Absence

"ל כראות א' ה'/ ויהי כראותו את הנום 2430 Gm. 2430

With pronoun object expressed by AN with suffix:

In ST

LV. 105 DAN 1401 / DINW11

In MT

Gn. 187 | 12 | 3 | 12 | 3

(1) Form with 7 locale

In ST

Gn. 256 אוץ קדם / סדף א In MT

Gn. 201 2217 (7 /7 757 N

(m) Prepositions

Difference
Gn. 22¹² TYLE SY TE NOON SN / A SN ' A'N 3416 אוד אוד אוד אוד אוד אוד אוד איינו כעם אחד 'ניינו כעם אחד Absence./

Absence

In MT Gn.485 בראובן וכשמעון ל פראובן וכשמעון ל

(n) Definite article

In ST

Gn. 228 א' א' ב השה צ' / אצהים יראה צו שה ציצה א

In ST Gn. 417 או שם העיך / 'ה שם 'ו

Gn. 335 (1') את צ' / וישא ציניו

(p) Residuum - unclassified further.

Gn.4115, AN 7603 DISA YAWA (70 N 3) / W '3 'A YAWA ('3)

שמרים מ (המקדש) שמרי משמרג (הקדש) אנים מים

(11) Abnormal Constructions.

d. Constructions where no cogent reason can be adduced in favour of ST or MT.

Absence of apocopation in Impf. or Impv. in ST. While there are cases where ST and MT agree in reading apocopated forms, these are many other cases in the ST where apocopation of lamedh he forms does not occur.

ו ירב ב' והעוף ירבה בארץ בארץ לבו. ו ירא י' צ' / יראה יהוה עציכם 21 Ex. 521 With waw ("and")

INDA / INDAF

Lv. 15²⁴ ואה'ה נדתה עציו / אה'ה ודתה עציו

There are so many of these examples, fully noted in Appendix A, that they cannot be dismissed as insignificant or arising from scribal error. Evidently the fuller form of the verb was frequently employed by the Samaritans. A case might be argued for the Samaritan here. Apocopation, so distinctive of MT, is unknown in Aramaic and Arabic. Logically, at least, it would be a later development which occurred regularly in MT, to some extent in ST, and not at all in Aramaic or Arabic. It is also conceivable that this preservation of the fuller form might have a dialectal significance indicative of the language of Northern Israel.

- (2) Variants consisting of constructions not found elsewhere in Samaritan Pentateuch.
 - (1) In MT of Hebrew Bible.

Outside Pentateuch 2 examples.

2. Differences of Grammar with divergence in sense.

i. Doctrinal.

(1) Controversy regarding Gerizim.

The controversy regarding the sacred mountain of the Samaritans will be treated in detail under the third main group of the classification.

Vocabulary. Here, however, consideration is to be given to some syntactical variants that have a bearing on the problem. They will all be dealt with now as well as detailed in Appendix A. Too much importance should not be attached to the support given by the Versions to MT in this respect for presumably doctrinal passages antedated them all.

ואראש הרי קדם ואמגד גבעג עולם ST באוג עולם ארי קדם ואמגד גבעג עולם ארי ארי לי ו' גבעוג ע'

and taking orp in the adverbial sense as "anciently, of old". The parallel phrase later in the line is read of y Ay2; (and not 'y Ay2; as in MT). Clearly ST refers to Gerizim. On the other hand MT is quite general and does not refer to Jerusalem or any central sanctuary.

$$Dt.33^{19}$$
 ST אמים הרי יקראו V'

ST = "They call peoples to my mountain"

MT = "They call peoples to a mountain"

ST obviously refers to Gerizim. MT is indeterminate, conceivably a mountain sanctuary in the territory of Zebulun or Issachar.

This phrase occurs 20 times in the Pentateuch, in every case ST reading a Perfect and MT an Imperfect tense of the verb. The former tense would imply that Shechem had been chosen as the place of sanctuary as far back as the time of Abraham while the latter tense, of MT, would refer to Jerusalem chosen in time of David. These 21 variants obviously show intentional alteration, but taking them alone it would be unwise to say on which side the alteration lies: on general grounds a decision is not possible for in the polemic relationship between Jew and Samaritan either side might conceivably be responsible for this alteration.

(2) Other Samaritan beliefs and practices.

(i) Unity of the delty.

It is generally considered that these passages have been altered to the singular verb by the Samaritans to conform to their doctrine of the essential unity of God. Certainly the variant shows intentional alteration and it is difficult to imagine what motive MT might have had if it had changed the verb to a plural.

(11) Avoidance of anthropomorphism. Gn. 16¹³ 7~7 3~ / '~7 3~

ST = "a seeing God", reading pt.

MT = "God of vision", reading noun (genitive both subjective and objective)

Ex.34²⁴ (16 out of 20 v.Gall MSS) it actually writes a Niphal, carrying the process even further in its desire to avoid any irreverence to the deity.

(111) Resurrection.

The Samaritans find in their reading with the suffix the justification for belief in resurrection. This has all the appearance of a later interpretation incorporated into ST.

(iv)Legal enactment.

Lv. 613, Dt. 164 בערב / בין הערבים

The phrase "between the two evenings" is generally understood to mean "between sunset and dark" and is read by both ST and MT eleven times elsewhere in the Pentateuch. It might appear that these 2 variants show intentional alteration by ST to make their text conform to these other passages. If so, it is difficult to understand why ST, on this reasoning, changed Dt 16^h and left Dt 16⁶ untouched, for in the latter passage both ST and MT read

בארב. The explanation may be that the two phrases were virtually synonymous with בין הארב'ם expressing greater precision. This view seems to be strengthened by the fact that LXX implies fuller form בין הארבים only in one passage, Lv.23⁵, and in all the other passages.

ii. Non-doctrinal.

Selected examples are given. Detailed lists in Appendix A. Naturally it has not been practicable to discuss in detail all the variants listed in the Appendix, but the principle followed has been to apply the criteria of intelligible grammar and/or sense in every case. In certain examples further criteria apply when, for example, a variant consists of a common form or word for an unusual one. or of a form or word from a parallel passage. Further, the line structure of a poem may act as a test of the soundness of a reading. There is, of course, the detection of the typical causes of scribal error such as dittography, haplography and the like. Where ST or MT is difficult in sense, in some cases to the point of unintelligibility, a decision in favour of the more meaningful version is considered as further strengthened if it has the support of the LXX. The choice here is frequently not between an easier and more difficult reading but between a meaningful and meaningless one. The support of the LXX is indicated by an asterisk in the Appendix.

(1) Variants where there appear to be cogent reasons in favour of ST.

Gn.25²⁸ 1'92 17'5 '7 / '2 7'5 '5

Sense of MT is unsatisfactory. ST with suffix gives such/

such good sense, with support of LXX, that it should be adopted.

בני של שים / בנים של ישים / סיש 3 Gn. 50

ST = "great grandchildren" MT = "great, great grandchildren".

No reason why descent should be carried further in line of Ephraim than Manasseh. LXX with ST.

בי תקראנה מ' / כי תקראנו מצחמה Ex. 10

The form in MT is difficult (cf GK § 47.k) and the grammar and sense of ST is excellent. LXX with ST.

(2) Variants where there appear to be cogent reasons in favour of MT.

בשקחות ה' / בהשקות החים מותם ב

 $\lambda \gamma \omega$ a rare word, occurring in MT only here and in 24^{20} .

ST reads Inf.Const. in both places - common for rare form.

אלהי אבאיך אלהי אברהם ואלהי יצחק ואלהי יעקב ST אלהי אבאיך אלהי אברהם ואלהי יצחק ואלהי יעקב ב Ex.36 MT '' ''

Probably an unnecessary attempt at correction on the part of ST. The reading is so much easier that one hesitates to adopt it.

LV.1518,24 AAN TON 2001 / 'N ON "

Appears to be cuphemistic reading by ST.

(3) Variants where there appear to be no cogent reason in favour of ST or MT.

In many of these examples there is evidently difference/

ence of interpretation implying different grammatical subjects, at least equally possible.

ו' האדם 3' ח' / ויהי אדם 3נכש חיה 'Gn.27'

Use of Q7% as proper name, or as generic termwith definite article. Usage is not consistent in either version, e.g. earlier in verse 7 both agree in reading

ST refers to Noah, MT to God. Both make excellent sense and either reading possible.

קרא ש' 3 / קראה שמו צוי ש'3 / מרא

Naming of child by mother in ST. In almost every example of this type ST reads feminine subject and MT indefinite masculine, though in $Gn.4^{25}$ the position is reversed and ST has the indefinite usage while MT reads $N^{2}N^{2}$. Either usage might be defended here and either reading might be original.

III. Differences of Vocabulary.

1. With agreement in sense.

1. Substitutions.

This term is used simply to mean that ST and MT read words from different roots. There is no implication that one is necessarily original and the other has arisen by intentional alteration.

(1) Pure synonyms.

(i) Common words.

a. Common for rare word.

There are many Samaritan variants consisting of pure synonyms that are words common in both ST and MT and that show the ST with a well known word instead of a less familiar one in MT.

(The ST reading is given first)

b. Words suggesting misreading, miswriting or mishearing in ST or MT.

While the ST and MT are quite good Hebrew, it is clear that one of the typical causes of scribal error (such as confusion with letters of similar shape) could account for the change. In many cases it is not possible to determine on which side the error lies.

$$Gn.27^{13}$$
 $\eta \wedge / \gamma \wedge$ (alike in both ST and MT scripts)

Nu.229 and 'are certainly alike in ST script but in context the two words could be so readily interchanged that unwise to decide for one or other).

c. Words in identical or similar context.

There are many cases where the ST has a word or phrase used elsewhere by both ST or MT in the/

the same context or in a similar passage in the Pentateuch.

Clearly this type of variant, like that noted above under a. "common for rare" suggests the possibility of harmonisation on the part of the ST.

d. Words which do not belong to any of the above types.

The variants under this heading consist of words that could so readily interchange, and may actually do so in the context, that it is impossible to say which reading is original.

Included under this heading is a group, consisting of variations in the Divine Names:

There are in all 35 variants of this interchange of Divine Names, of which the references
are given in Appendix A. According to Skinner,
see above p.8, in the transmission of the
Divine Names in Genesis there are at most 9
divergences as compared with 310 cases of
agreement/

agreement in ST and MT. For the whole

Pentateuch there are only the 35 variants

mentioned and, in view of the frequency of

occurrence of the Divine Names, this small

number of variants is indicative of the care

exercised by both ST and MT in its transmission

(ii) Uncommon words.

b. Late

There are 10 passages involving 3 words where the ST reads a later form similar to that of Aramaic. The Impv. form '17 occurs 7 times for the MT 7'7 . Again, in Gn.18²¹ and 34³¹ the ST reads 77 (= MT 7 /), Cf. Petermann, p.149 and finally in Gn.14¹⁴ p7'! "and he scrutinised" for the equally difficult p7'! "and he emptied out" of the MT,

c. <u>Words suggesting misreading</u>, miswriting or mishearing in ST or MT.

There are 5 passages, where unusual words occur in ST and MT, and where typical scribal error could easily arise. It is difficult to say on which side the error lies. The examples are given here:

Ex. 18^{19} $7^{5'9}$ / $7^{59'}$ /
Dt. 30^3 $7^{52'}$ ω / 7^{512} ω

d. Words in identical or similar context.

Only 3 examples, which are all noted here.

- (2) Synonyms giving greater precision or emphasis.
 - (1) Common words.
 - b. Words suggesting misreading, miswriting or mishearing in ST or MT.

Only 4 examples.

Ex.
$$4^2$$
 nr ns / nr
Ex. 12^{22} 3 \sim / 3
 23^8 3 \sim / 3
Lv. 10^6 \sim 3 / 3

c. Words in identical or similar context.

Only 4 examples, of which 2 are given here:

d. Words which do not belong to any of the above types.

The readings of both ST and MT are so suited to the context that it seems unwise to adjudicate between them. In the two passages noted here the LXX supports the MT in the first and the ST in the second.

ii. Additions.

This term is used simply in the sense that the ST has extra word(s) as compared with the MT. There is no assumption in the term itself that this longer reading represents intentional alteration.

(1) Pure synonyms.

(1) Common words

a. Stylistic supplements in ST.

There are many passages whore the ST has additional common words which are of a purely stylistic nature and add nothing to the meanin They consist of

(a) Explicit subjects or objects.

(b) Words in apposition.

(c)Additional Naw ("and")

b. Words suggesting misreading, miswriting or mishearing in ST or MT.

There are 4 passages where typical scribal error could easily arise and where no decision for/

for or against ST seems possible.

c. Words in identical or similar context.

There are 14 Samaritan variants where additional words in ST are read by both ST and MT in the same context or in a similar passage elsewhere in the Pentateuch.

d. Words which do not belong to any of the above types.

These numerous Samaritan variants, detailed in Appendix A, could so easily belong to the original text that it is impossible to accept them or reject them as such. Their significance will lie not in the individual readings but in the broad picture they give of a fuller text than the MT. It remains to be seen under "Omissions", which follow this section of the classification, to what extent the MT has readings of this nature. Along this line, and with other considerations, the cumulative evidence/

evidence should be such as to make certain conclusions possible. Examples of this type of variation are:

$$\mathbb{E}_{\mathbb{X}}. \ 6^{27}$$
 מיזים $\mathcal{E}_{\mathbb{X}}. \ 6^{27}$ מיזים $\mathcal{E}_{\mathbb{X}}. \ 6^{27}$ מארץ מצרים $\mathcal{E}_{\mathbb{X}}. \ 6^{27}$

- (2) Synonyms giving greater precision or emphasis.
 - a. Record of events implied but not explicitly recorded in MT.

In Ex. 716,17,18 God has pronounced a warning regarding an impending plague and has commissioned Moses and Aaron to deliver the warning to Pharaoh. It is not stated in the · MT that these warnings are ever delivered. The Samaritan supplies the lack as in 718b it represents Moses and Aaron as going to Pharaoh and repeating almost literally the words of the divine warning. Other "plague" passages of this type are noted in Appendix A. While it is possible that these passages are expansions. it is at least equally possible that they are original, for they contain statements of events logically implied but not definitely recorded in the MT. these variants has the support of the LXX and all the commentators known to me condemn them.

Yet/

Yet surely they suggest interesting possibilities regarding the original form of the
OT writings, some of which must lie very
near to oral transmission. In such circumstances a narrative might conceivably be
recorded in full or, alternately, in an
abbreviated form to be supplemented by the
reader.

This theory may receive support from the views expressed in a monograph by I.V.Slotki who states that many portions of poems - words, phrases, lines or strophes - which in our present text appear only once may have been occasionally or regularly repeated more than once. In other words the MT may have preserved the shorter forms of poems that were originally much longer 17. Surely it is conceivable that this practice, discernible in poetry, may on occasion have been followed in prose narratives also.

Included under this heading are two other passages. The first is Samaritan Gn.30^{37a} = 31¹¹⁻¹³ in both ST and MT with minor adjustments. The Samaritan first relates the dream as it appeared unto Jacob and thereafter Jacob/

Jacob literally repeats the dream to Rachel and Leah, while the MT gives only the repetition to Rachel and Leah. Cf., however, Gn.41, repetition of Pharach's dream in both ST and MT. The second passage is Samaritan Nu 31^{21a} = 31^{21,22,23,24} with minor adjustments. Moses commands Eleazar who in turn commands the men of war.

c. Words in identical or similar context.

There is a very large number of Samaritan variants consisting of one or two additional words forming a phrase read by both ST and MT elsewhere in the passage, or in a similar passage at some other place in the Pentateuch. There are so many examples of this type of variant that it is clearly a distinctive feature of the Samaritan Pentateuch. Once again (see above, p. 101) it remains to be seen if under "Omissions" there is a comparable number of MT readings of this nature. A few examples will be given and a full statement furnished in Appendix A. Gn. 114 (115) 3 - TERES (115) 3 - TERES (115) 3 - TERES (115) 3 - TERES (116) Nu. 1418 (Ex. 346) AND 700 Dt.168 (Lv.237) אלאכה / מצאכה / מאכה אלאכה

d. Words which do not belong to any of the above types.

This/

This is quite a numerous class also. The additional words give some slight emphasis or precision or nuance of meaning that is generally in keeping with the context. A few examples will be given, with a full statement in Appendix A.

iii. Omissions.

This term is used simply in the sense that ST has fewer words as compared with MT, without any assumption of intentional alteration.

(1) Pure synonyms.

(i) Common words.

a. Stylistic supplements.

Stylistic supplements, such as occurred in ST, also occur here, though they are much fewer in number.

- (c) Additional <u>Naw</u> ("and")

 Gn. 7¹³ Λοιι οπ οω / 'ι οπι ω

 11²⁷ πολ Λτζιλ σζω / 'λ 'λ τζνι
- d. Words which do not belong to the above type.
 (Note: there are no omissions classified as due to misreadings etc. or to possible harmonisations).

Though the variants here are of the same type as in ST (see above, "Additions",p.100) they are much fewer in number.

- (2) Synonyms giving greater precision or emphasis.
 - (i) Common words

It will be noted that there are no examples of variants in MT showing explicit statements the occurrence of which is logically implied: neither are there any that are obviously due to scribal error such as misreading etc.

c. Words omitted in identical or similar context.

There are 9 passages where ST omits a word or phrase as omitted by both ST and MT elsewhere in Pentateuch.

נ' צפנ' יהוה א' ה' צ' / ניחוח אשה הוא ציהוה (2918) מו צפני יהוה א' ה' צ' אוח אשה הוא ציהוה (2918) מו בה ע' צ' / האיפב עצ עצאיך (233) מ' בה ע' צ' / האיפב

d. Words which do not belong to any of the above types.

The variants under this heading are suited to the context and it would be difficult to prefer either ST or MT in any particular case. The overall picture, however, is quite clear which is that ST has more readings of this type than MT so that in this respect ST is the longer document.

Ex. 17^3 13 17^3 75 3 77 3 77 3 10 17^3 10 17^3 10 17^3 10 17^3 11 17^3 11 17^3 11 17^3 12 17^3 12 17^3 13 18^3 13 cases where ST omits 18^3 3 e.g. Dt. 18^3 15 18^3 16 MT

2. Differences of vocabulary with divergence in sense.

i. Doctrinal.

Variants of Syntax that show difference of meaning of a doctrinal nature have been already considered. There are also variants of Vocabulary revealing doctrinal difference.

(1) Controversy regarding Gerizim.

This controversy has separated Jew and Samaritan (and many scholars!) down the long centuries.

There are 4 scripture passages that are relevant here: two of them will now be considered together. Dt. 11³⁰

Dt. 11³⁰

ODE SID HOLD / RODE (MANY MSS)

ODE SID HOLD / RODE (MANY MSS)

The Jews have accused the Samaritans of altering the/

the text here in the interests of their cult and the Samaritans in turn have hurled the accusation back at the Jews. Scholars such as Gesenius. Montgomery and König have taken the view that these variants are examples of deliberate Samaritan alteration for polemic reasons. Kennicott. however. defended the Samaritan reading of Gerizim for Ebal on the ground that it was more natural for the memorial stones to be erected on Gerizim. the hill of blessing, than on Ebal, the hill of cursing. In more recent times he was followed by Meyer who not only adopted the Samaritan reading but maintained that the Jews had adjusted the text of Deuteronomy 27 in the interests of the Jerusalem sanctuary by removing Gerizim and Ebal from near Shechem to the Jordan valley at Gilgal - an alteration surprisingly accepted by the Samaritans 18. Torrey stressed that the bitter controversy that developed between Jew and Samaritan and which is clearly indicated again and again in the partisan attitude of the writer of Chronicles resulted in the Jerusalem text of Dt 274 being altered to Ebal 19. The arguments adduced for and against these two Samaritan readings appear general/

general and inconclusive and the view taken by
the present writer is that it is not possible to
decide one way or another regarding the variants
as such. In the polemic relationship that existed
between Jew and Samaritan either side might, conceivably, have altered the text. Too much weight should
not be attached to the fact that MT is supported
here by the Versions for such doctrinal variants
would have antedated them all.

The two remaining scripture passages are:
Ex.20^{17b} and Dt. 5^{18b}

The Samaritan Tenth Commandment.

At the end of the legislation in Dt.26, in both ST and MT there is added an appendix Dt.27²⁻⁷ enjoining that the laws be engraved on stones to be erected on a mountain, Gerlzim in ST and Ebal in MT. This appendix is also found in ST at end of the Decalogue, Ex.20^{17b} and Dt 5^{18b}, together with a supplementary note regarding the location of the mountain, Dt. 11³⁰. It is regarded by the Samaritans as the Tenth Commandment, the First and Second Jewish Commandments being treated as one. The Commandment reads as follows:

"And/

"And it shall come to pass when Yahweh thy God will bring thee into the land of the Canaanite whither thou art going to possess it. then thou shalt erect for thyself large stones and thou shalt cover them with lime and thou shalt write upon the stones all the words of this Law and it shall come to pass when you cross the Jordan you shall erect these stones which I command you to-day on Mt.Gerizim and thou shalt build there an altar to Yahweh thy God. an altar of stones, thou shalt not lift up upon them iron, of perfect stones thou shalt build the altar of Yahweh thy God. and thou shalt sacrifice peace offerings. and thou shalt eat there and rejoice before Yahweh thy God: that mountain is on the other side of the Jordan at the end of the road towards the going down of the sun in the land of the Canaanite who dwells in the Arabah opposite Gilgal near Elon Moreh opposite Shechem."

While a good case can be put forward for the Samaritan reading of Gerizim instead of the MT Ebal in Dt 27⁴ and incorporated in the Commandment, it would indeed be difficult to defend the authenticity of this long passage. It is rejected for the following reasons:

 Detailed nature of the final verse compared to other passages in the Old Testament referring to a particular place.

- 2. Undue length as compared with the other Commandments.

 3. Almost literal parallelism with $Dt.27^{2-7}$ and 11^{30} .
- 4.A ritualistic prescription of a different nature from the other Commandments. These do not deal with ritual but are concerned with men's relation to God and to one another. This Samaritan Tenth

Commandment appears to lie outside the logical

- 5.Complete lack of support from the Versions apart from the Syro-Hexaplar where it is quoted as belonging to the Samaritan text²⁰. However, too much weight should not be attached to this absence of support from the Versions for doctrinal variants
- (2) Other Samaritan beliefs and practices.

would have antedated them.

Many, though not all, of the following doctrinal variants may reflect distinctive Samaritan views and to that extent may be discredited.

(1) Majesty of God.

grouping.

of Yahweh's presence". Cf.B.D.B., p.817.

(ii) Avoidance of anthropomorphism.

- (iii) Day of Vengeance and Recompense.

 Dt. 3235 a 361 apr a 1.3 / '1 '1 3
- (iv) Legal enactments.
 Ex. 21²⁸ שור או כ3 בהמה / אשור /

8 other exx. of this phrase are noted in Appendix A. Clearly a Samaritan insertion. See above, p. 19.

(v) Veneration for religious leaders.

(vi)/

(vi) Genealogies of patriarchs.

• 0	Year	9	
Gn.5 ¹⁸	62)	
5 ¹⁹	785	{	Jored
530	847	}	
5 ²⁵	67)	
5 ²⁶	653	{	Methuselah
5 ²⁷	720	5	
5 ²⁸	53)	
₅ 30	600	{	Lamech
531	653	5	
1112	135)	Arpaksad
1113	303	}	rea greener con
1114	130)	
1115	303	}	Shelah
1116	134)	Eber
1.17	270	}	
1118	130)	Peleg
11 ¹⁹	109)	•
11 ²⁰	132)	
1121	107	}	Reu
11 ²²	130)	
1123	100	}	Serug
1124	79	,	Nah o r
11 ²⁵	69	3	TA CITTAT
1132	145		Terah

According to Skinner (ICC, Genesis, p.135), the ST, MT and LXX have each "carefully constructed systems of chronology". The evidence does not appear to be sufficient to allow of a decision in/

in favour of the priority of any one of them.

(vii) Sectarian opposition.

A hated opponent.

Gn.35¹⁸

O'b'12 / |'b'12

ST = "son of days".Cf."a late-born son", $Gn.37^3$.

MT title is honorific "son of the right hand" i.e. favoured son.

Dt.33¹² 7' 7' / 7'7'

The almost meaningless 7' 7' to avoid "beloved" of MT being applied to the tribe of Benjamin.

ii. Non-doctrinal.

The method here followed is to give representative examples, with detailed lists in Appendix A. The critical rules applied here have been already outlined in the relevant section of Syntax. See above, p. 93.

(1) Variants where there appear to be cogent reasons in favour of ST.

Substitutions

Grammar and sense of MT are almost unintelligible. ST, with support of LXX, gives excellent sense.

Sense of MT is unsatisfactory: in verse 6 Jothros
is/

is represented as already speaking to Moses and yet in verse 7 Moses is stated as going out to meet his father-in-law and asking after his welfare.

Read \nn' as indefinite "they said" followed by \\
\(\pi_17 \) of ST "behold", with support of LXX, and the sense is excellent.

Additions

MT omits השרה מצכה השרה

Usually direct speech after >>> Mistake has arisen through homoeoteleston in MT. LXX reads with ST.

ST, supported by LXX, may well be correct. MT impossible.

Omissions.

Gn.36¹⁶ ST omits Top 713N before OAY2 713N

ST reading has parallel <u>outside</u> Pentateuch

(1 Chron. 1³⁵)

יף קוא read by both in 36^{18} . Appears misplaced in MT.

Nu.26⁴⁵ ST omits גני בריעה before אונג 3 before בעי בריעה 3 LXX with ST. MT has redundant words at beginning of verse; out of context.

(2) Variants where there appear to be cogent reasons in favour of MT.

(2) Substitutions.

Attempt at clarification by ST. Evidently misunderstood ancient ceremony of adoption. LXX with MT.

Word in MT occurs only here but with support of LXX.

ST may have read the word more familiar or less euphemistically objectionable.

Additions.

Avoidance of anthropomorphism in ST. Possible harmonisation with 22^{22} .

Longer passages

Incorporation of passage from parallel narrative

Ex.20^{21b} = Dt. 5²⁸⁻³¹

Substitution of passage from parallel narrative

Dt.10⁶⁻⁷ ST = Nu.33³⁰⁻³⁵

Omissions

Nu.21 35 (1) Dt.3 3) 1/2 AM1 omitted after 1AM LXX with MT.

(3) Variants where there appear to be no cogent reasons in favour of ST or MT.

Substitutions.

Both words occur in the context and either gives excellent sense here. Unwise to docide in favour of ST or MT. LXX reads with ST.

$$Ex.5^9$$
 ST וישעו בה וצא ישעו בדבר שקר או אישנו בי ו' י' ב' שי איש

ST, with its excellent sense and with support of LXX, at least equally possible. Transposition of Y and \(\psi\) obviously possible with these verbs. Additions.

ST at least equally possible with MT. Omission of

ST with support of LXX gives excellent sense but MT may show suppressed apodosis. Cf. GK, \$1.6.0.dd. Omissions.

LXX supports ST but excellent sense with either ST/

ST or MT.

LXX with MT. In preceding verses: 9^9 "the tables of stone, tables of the covenant", 9^{10} "two tables of stone". In 9^{11} either phrase of ST or MT seems equally possible.

Footnotes, Chapter V.

- 1. Petermann, Formenlehre.
- 2. F.Diening, Das Hebräische bei den Samaritanern, 1938.
- 3. Kahle, Geniza, pp. 231 ff.
- 4. A.Sperber, 'Hebrew based upon Greek and Latin Translations', H.U.C.A. xii-xiii, 1937-8, pp. 103 ff.
- 5. Petermann, p.9
- 6. Kahle, Geniza, pp. 86 ff.
- 7. Petermann, p.57
- 8. W.B. Stevenson, Grammar of Palestinian Jewish Aramaic, 1924, § 23, 9.
- 9. B.J.Roberts, The Old Testament Text and Versions, 1951,p.6
- 10. P. Wernberg-Møller, The Manual of Discipline (in Studies on the Texts of the Desert of Judah) 1951, p.8
- 11. F.M.Cross and D.N.Freedman, <u>Early Hebrew Orthography</u>, 1952, p.69
- 12. Gesenius Kautzsch, Grammar, 1910, 9 75 v.
- 13. Driver, Samuel, p. xxxv
- 14. Petermann, p.93
- 15. Brown, Driver and Briggs, Lexicon, p.214
- 16. For fuller treatment cf. E. Robertson, Catalogue of the Samaritan MSS in the John Rylands Library at Manchester, 1938: von Gall, Pentateuch, Schreibweise und Einrichtung Samaritanischen Handschriften, pp.lxii ff.
- 17. I.W.Slotki, Long and Shorter Versions of Ancient Hebrew Poems, A.J.S.L., vol.50, pp. 15-31
- 18. Die Israeliten u. ihre Nachbarstamme. Halle, 1906, pp. 534-47.
- 19. Ezra Studies. Chicago, 1910, p.329
- 20. Kahle, Geniza, p.145.

CHAPTER VI.

Evidence of Dead Sea Scrolls.

The Texts of the Dead Sea Scrolls in Discoveries in the Judaean Desert I.Qumran Cave 1, II Les Grottes de Murabba'at III Les Potites Grottes de Qumran have provided new material which may have a bearing on the relationship between ST and MT. As stated above, p. 37, the Biblical texts from Murabba'at show close affinity with the MT. "Le texte, conservé sur les fragments, est tout à fait identique an textus Receptus établi verse la fin du ier siècle de notre ère à l'Académie de Jamnia."

At Qumran, 60 years earlier, a different position is disclosed. See above. p. 37. The Biblical MSS there reveal variant text traditions. One is virtually identical with that of our present MT. "Texte et graphie semblent peu différents du texte massorétique" Another closely resembles that of the ST. "Il semble que ce MS soit plus proche du Texte Samaritain que du Texte Reçu"2. Another has strong affinities with the Hebrew recension underlying the LXX. "Le texte se rapproche de la LXX et de la Vulgate. On relève des variantes morpholegiques et des graphies très pleines charactéristiques de Qumran"3 As well as these three text traditions there are others of a mixed type. "On notera que les graphies sont de type plutôt Samaritain (c'est-a-dire de type classique pre massoretique), tandis/

tandis que le texte - pour antant qu'on en puisse juger est de tradition judéenne 4.

Accordingly, it is the earlier evidence from Qumran (DSS I and III), and not that from Marabba'at (DSS II), that will be relevant to the present study. The Pentateuch Texts from Qumran have been collated by me with a view to determining where the Scrolls support the ST or the MT or diverge from both and thus the classification of the evidence proceeds along these lines:

- 1. Agreement of Scrolls with ST.
- 2. Agreement of Scrolls with MY.
- 3. Divergence of Scrolls from both.

The threefold division of variants under Orthography.

Grammar and Vocabulary is followed.

These signs used in the critical apparatus of DSS I are also employed:

N very uncertain reading

N probable reading

ooo exact number of letters missing

... uncertain number of letters missing

[] lacuna

DSS I

Difference of Orthography.

Agreement/

Agreement of Scrolls with ST.

<i>(</i>)	3 ¹¹	ער[ום	MT
			לי רם לי רם
Ex.	20 ²⁶	NIZNZ	13x2
Lv.	1931	הידעונים	R·アヤL·ロ
	2021	N.1	הוא
	2023	בחקות	Vbus
	237	הראישון	וראשון
Dt.	11 ²⁹	גריזים	גרזים
	Maria.		
	2411	" NS1"	N.21,
	29 ¹⁶	ב לוליהם	Q7,332

Agreement of Scrolls with MT.

_	22		<u>ST</u>
Lv.	20 ²²	'∧ ק ∧'	<i>חק</i> ו ת'
Dt.	1130	M3 [1	N137

Divergence of Scrolls from both ST and MT.

Gn.	3 ¹⁴	315	errettiiviteit valetiitiite valet roost tootta talettiinisteri valet valet valet valet on talettiinis on talet	ST/MT 35
			2	

Fragment 15, p.50 praisk

Lv. 2	3 ⁴ DAIN	٨٨٥
Dt.	1 ²³ acar	ロンカ
	והצכ]תה פנ8	١ π 3 ح ۸
	8 ¹⁹ 4'5	د،
	9 ²⁷ 215[y'	יעקב
1.	אעכי ²⁷	n1c,
1	1 ²⁸ NIIC	,51W
1	3 ⁴ 712	הוביא
1	היש כמ [ה 3 ⁴	הישכם

		(contd.)
Dt. 13 ⁴	ההואה	ST/MT
13 ⁵	אצודיכמה	א ל הי כ ם
13 ⁵	ואוֹאו	IVVI
_		
136	והנבי	והנביא
136	מלקרבכה	מלנבר
136	ההואה	ההוא
13 ¹³	CIN	۲)
1313	אָר יכ _{ּק}	עריך
13 ¹⁴]ערימה	צירם
1424	1203	11146
1424	۵،۷	٠,5
14^{24}	אצוהיכה	7.234
1425	7517.54	7,234
16 ^l \$	413	<i>~</i> 3
164	å o 3	73
16 ⁶	אצו]היכה	7.73~
167	713621[ادسائم
2413	ひマソコ	51277
2414	706	γ· > ω
32 ¹⁹	1'1121	1 2 L V.1
33 ¹⁸	113213[1	1812131

Differences of Grammar.

Agreement	01	Scro.	<u>l. l. s</u>	with	ST.
	7 L				

Gn.	אג שם המקום 22	ויקרא אברהם
	20 ²⁵	17 33 n A
Dt.	13 ⁵ 290 13 ⁶	103 A 10 E O 10 [w'c O Nwein
	*	ا دُلِس ح ۵
	3217 A	, 1.2 ω 1.1.

MT 'a' w' 'n' 1	(AMomitted)
73311	
1531	
_D 3 n	
נטיכם	
てるりょうべん	

Agreement o	f Scrolls with MT.	CITI.
Ex. 214	3 471.1	<u>ST</u> 1'37 43
Iv. 20 ²	אתכם הארף ב	ENTS HACO
Dt. 30 ²	וצדבקה	727 n31
30 ²		2 > 3
31 ³	צובר	フ ュタオ
311	אצהיכם 3	אצהיהם
32 ²	ام ما	[haj
32 ²	rs[a3	8 n a 1
32 ²]، الرا	וינק
32 ²	יבינן	ויבינן.
33 ²	⁴ (327	1.327

Diverge	nce of	Scrolls from	both ST	and	MT.	ST/MT
Gn.	1 ²⁰	ر س] در ۱۶				18701
ex.	16 ⁴	Cngca			C	SONS
Lv.	22 ⁶	ျာ ၁၁၊				ω ∋ J
Dt.	123	<i>ن</i> ا ن				0'14
	9 ²⁸	11481,			137	N817
	134	ا به اسکاه ۱			;	γρωΛ
	1424	1203			i	N M W
	1 6 ⁶	1263 12				1263
	2416	٠ ٩ ١ ٨				יו א √ו
	31 ⁵	7.203			ء ۾	3 פנים

Differences of Vocabulary.

Agreemen	t of	Scrolls	3 W:	i th	ST.	,	
Ex.	(C) (H)	הנם					۱۱. د له

'א 'א' ה' () omitted)

Agreement	of	Scrolls	with	MT'.	
Lv.22	-			8745	
nt.	49	9 7:	2777	a '	

	Scrolls from both ST	and MT. ST/N	NT'
Dt. 1 ²⁴	47 N 7	NAG	АДЗАния
13 ⁵	ゖ゙゙゙゙゙゙゙゙゙゙゙゙ヿ゚ヹ゚゚゚゚゚゙゙゙゙゚゚゚゚	١ ٨٠ ١٨١	elsewhere
13 ⁵	٨٥٥٦	117271	in verse)
1514	, 1 ° [N	าเสา	
15 ¹⁵	Allov 3 7 18 n OIN	'b'N	
29 ¹⁴	ע מן ים	צמנו צמד	
311	ויכל משה לדבר	וידבר 'מ' וידבר	
31 ¹	את כל הדברים האלה	'a 'a 'N	
32 ² 7	*** *JTN	ลเสร	

DSS. III

Differences of Orthography.

Agreement of Sci	colls with ST	
		<u>MT</u>
Gn. 18 ²⁵	Λιωγη	ΛωΥΙ
Ex. 9 ²⁹	MISIA	N1377
11 ²⁶	4017 [DD	ለዕግ ១ አ
12 ³⁴	∧(n] ins	አ ንጉ ሄ
	112 [332	1~332
26 ¹¹		
Lv. 8 ¹³	<pre></pre>	$A_{1}A_{2}$
Nu. 33 ⁵²	12611	1241
Agreement of Sci	colls with MT	വന
Ex. 1 ¹²	1-5-7-	<u>ST</u> פרק'
	12[7]	
Nu. 4 ²	ολήσω63	D NIA DE B 3

Divergence of	Scrolls from both ST and MT	
Gn. 6 ¹⁹	אגפק	ST/MT
Ex. 1 ¹²	and (sic)	722
112	1215,1 1210)	1,451
-	מאדוב[ע	2/12
114	C	
7 ²	ואהרון	ואהרן
74	N131	184
9 ²⁹	7441.1	ויאמר
9 ²⁹	alm di	Jen
1.14	7016	7645
116	23172	23c 2
77.67		
12 ³⁴	anindab	anneb
18 ²²	3เึอ้ฉ	322
1 9 ⁹	7WID	746
26 ¹¹	717271	λ 72 n 1
26 ¹²	(sic) האלה (המהצ
30 ²⁵	קודה	970
30 ²⁵	700007	רקח
3410	ן האוניו]	ויאמר כצ
3410	315	
34 10	ץ אכק • • • • • • • • • • • • • • • • • • •	لامح
Lv. 8 ¹²	٥٠١١ [C 7 P
Nu. 18 ⁸	<i>ج</i> ا ٦ [س ا	קדשי
33 ^{5&}	12611 312	1261 35
33 ⁵²	ב d ι λ α α Γ	ΔΛ ħ ૠ
33 ⁵²	[והורשגמ]ה	ДДШ (П I
Dt. 18	EN211,c[a	באב איכם
19	ואומר	ן אמר ע <i>3</i> יקם
716	ע3 ה מ	
717	3 הורשם	3 הורישם

75 144	•	ST/MT
Dt. 815	NSINA	ST/MT Raisin
8 1 6	MACAF	7^. >"~
1010	ההיאה	ההיא '
1011	ויו אמר	ויאמר
10 11	צאבוגיהמה	anzn3
17 ¹⁴	JA11	\^J
17 17 ¹⁴	723	73
1714	7/1/2120]	,V5,20

Differences of Grammar.

MT	Scrolls with ST	nt of	Agreeme
Saite1	צמי]ניהם	6 ²⁰	Gn.
ברשו מחצרים		1239	Ex.
מבנט	אבנטים	813	T.v.
ってい	الباع	18	
当り、4	איך	717	37 0 6
3ب	1 [E14	1010	

Agreement of Scrolls with MT Nu. 43 Dt. 17¹⁵ Dt. 17¹⁵ No. 43 No. 44 No. 44

Divergence of Scrolls from both ST and MT

Lv. 11 ²⁶	14041	ST - 77001	MT	YOUI
Dt. 7 ¹⁷	3 N 7	ST/MT 73 ~ 7		
89	ומ] בריה	ומקרריה		
8 ¹³	בֿקרך	ובקרך		
β17	0 2 2 2 3 2	72232		
108	าปีตั้ง3เ	£ 2015		
1011	מ מבותי המ [ה	e nan 3		

Differences of Vocabulary.

Agreement of S	erolls with ST	MT
Ex. 11 ³	באון הרא מצרים	בארץ מצרים
114	ויאמר מושה אל פרע [ה	'n '\
3410] > 114 [NI,

MT

	0.5						
Lv.	1125	D' 52 9	ורח		′ 2 ′ı	1'722	035,
Dt.	87	7217	1 72	116			را دا
	818	294.31 p	15.3	צאברהם	צמבתיך	7'	N=1/3

Agreement of	Scrolls with MT	ST	
Ex. 12 ⁴⁰	34761 132	בו י' ואבאם	
27 ¹⁹	·3 > 3 = 3	וצשית את כל כ'	
Nu. 3 ³⁹	משה ואהרן	'h	

Divergence of	Scrolls from both ST an	d MT st/MT
Ex. 1 ¹²	18761	
		۶٦٥'
9 ²⁸	וברך ואט	てつるし
30 ² 5	יה'ה צדורוגי כם	4
31 ¹⁶	הש]ביל ליה[ה	ัส
34 10	ויאמר ין הוה	1
34 ¹⁰	ויאמר יהוה אלב אשה את יושבי הארץ	'n 's 35 'N
Nu. 33 ⁻ Dt. 7 ¹⁹	הממת הבדלת היום	ASTIR ADDR
7 ¹⁹	ה לנאת [ביא] זא תוכנ	ST ALANA MT AANAL

Conclusions from the foregoing evidence.

Before attempting to interpret the evidence it is necessary to assess the evidence itself. The following considerations should be kept in mind:-

1. Fragments of Texts. The evidence consists of fragments of texts and not long continuous documents. On occasion, the fragments can be tantalisingly incomplete. For example, in Dt 14²⁵ the reading of DSS I is \n\(\begin{align*} \left(2)\end{align*}\), one of the well-known passages where ST reads (the place which the Lord thy/

- thy God) "has chosen" and the MT "will choose". The lacuna here makes it uncertain on which side the evidence of the DSS lies.
- 2. Areas of Pentateuch unrepresented. The fragments are drawn from comparatively limited areas and there are stretches of the Pentateuch from which there are no samples whatever.
- 3. Representative samples. Samples of variants from a large corpus of material such as the Pentateuch may be relatively few in number but may consist of examples of types of variation and so, by their representative nature, be applicable to a much larger area of text than that from which they are actually drawn.
- together, may have a cumulative significance and importance greater than when they are considered separately. To give a hypothetical example: if a passage shows evidence of possible expansion, together with a literary style different from that of the rest of the document, together with actual wording found elsewhere, these three considerations would cause a strong presumption against the reading.

 Bearing these considerations in mind it is now possible to ask what results have accrued from the investigation.

 While the evidence of DSS I and III has been outlined

separately,/

separately, examination shows that the textual variation is of the same nature in both and so the conclusions have been listed together. The three sub-divious of Orthography, Grammar and Vocabulary have been followed.

Orthography.

The Scrolls agree in a few cases with the MT, more frequently with the ST but, in the great majority of passages available, diverge from both ST and MT, which are in agreement with each other. The picture of orthographical variation is clear and definite. The ST has a fuller orthography than MT and the DSS have a still fuller orthography than the ST.

Grammar.

Unlike variations in orthography, those in grammar do not show a large number of cases in which DSS diverge from both ST and MT which are in agreement with each other. The three sub-divions of grammatical variants are remarkably similar in size. The reason appears to lie in the fact that few of these variants show any divergence, or even difference, of meaning. They are due to chance agreement or disagreement arising from interchange of grammatical usage: two grammatical constructions were known to the scribes and so there are passages where all three agree in reading a construction, or in not reading it, or where one reads it and the other two do not.

Vocabulary.

Vocabulary.

As with orthography, so with vocabulary: the Scrolls support the ST or MT in a limited number of cases but in the great majority of them follow an independent line. An examination of the nature of the variants suggests the cause. Many of them consist of extra words or expressions giving greater precision, clarity or emphasis so that the tradition, or traditions, represented by them would appear to be that of a longer or more explicit document. On general grounds perhaps a later document, though of course each variant reading has in the first instance to be considered on its own merits, before being grouped with others of the same type.

It is significant that the ST is a longer document than the MT by reason of such extra words showing greater precision, clarity or emphasis. Accordingly, if there examples of the DSS are representative of a much larger corpus of material, as they appear to be, then the DSS, as compared with the MT, carry the process of amplification even further than the ST.

Footnotes, Chapter VI.

- 1. DSS III, p.105
- 2. DSS III, p.169
- 3. DSS III, p. 61
- 4. DSS I, p. 52

CHAPTER VII.

Conclusions.

It now remains to summarise the results obtained and these will be set out under <u>Particular Conclusions</u> and <u>General</u> Conclusions.

Particular Conclusions.

I. Orthography.

1. Variants attributable to scribal error.

There are variants in passages where the text, as it stands, in ST or MT or both, is doubtful or corrupt. Some of these variants arise from the typical errors which can take place in the course of MS transmission e.g., confusion between letters similar in shape or sound, and which obviously operated in both ST and MT. In addition, there was another cause of confusion to which the Samaritans were specially prone, namely, their practice of not distinguishing gutturals in the reading of Hebrew. This led to interchange, addition, omission or transposition of gutturals, not only as between ST and MT but within the Samaritan MSS themselves.

- 2. Variants making the pronunciation explicit.
 - The ST like the MT uses letters $\nearrow 7'$ / to make the pronunciation explicit.
 - i. Use of Aleph.

There appears to be more examples of the clision of consonantal/

consonantal aleph in ST although many of the readings are not found in all the von Gall MSS. Likewise aleph, as a vocalic consonant in medial position, occurs more frequently in ST. As in MT aleph frequently represents a final vowel, sometimes interchanging with he of MT.

A prosthetic aleph occurs more frequently in ST as the orthographic closing to a final long vowel.

11. Use of He

A final vowel may, or may not, be represented by he.

Accordingly apart from cases where both ST and MT agree
in reading the final he, or in not reading it, variation
arises where one reads it and the other does not.

iii. Use of Yodh and Waw.

(1) As compared with MT there is a greater use of yodh and way as vowel letters.

Certain significant categories have been extracted from this immense group of variants but no attempt has been made at a complete classification apart from the grammatical arrangement in Appendix A. If a meaningful, complete classification is not possible, it hardly appears to be necessary for the significance of the profuse use of yodh and way is not really in the details but in the group as a whole. This very full orthography represents a type of text prior to the standardisation of the consonantal text. This viewpoint receives strong support from the fact of the still fuller/

fuller orthography of the Dead Sea Scrolls discussed in Chapter V (see above, p.71) and more fully in Chapter VI.

(2) Lamedh He verbs and nouns. Yodh occurs occasionally as an archaic spelling with retention of the third radical or a late phonetic spelling with yodh as the bearer of the vowel Sere

Such an occasional usage may indicate the survival of an earlier usage or the arrival of a new one. The evidence is not decisive and it may be that the two usages, early and late, existed alongside each other.

3. Variants that normalise the spelling of words.

If MT appears to have archaisms, errors or inconsistencies.

ST frequently reads the normal form. The ST reading is so much easier that it is unwise simply to accept it without asking why MT preserved another reading so evidently contrary to normal practice.

4. Orthographical variations with proper nouns.

There is a large number of proper nouns that show orthographical variation arising from such causes as confusion
with letters, especially gutturals, and the presence or
absence of vocalic letters making the pronunciation explicit

5. Collation with Dead Sea Scrolls.

The wvidence of the Dead Sea Scrolls is summarised under General Conclusions but it may be noted here that some of/ of the MSS of the Scrolls, notably all those from Murrabba'at, show an orthographical usage in complete agreement with what is found in MT. Others from Qumran have a more developed orthography, such as the frequent plene writing of yodh and waw, in close agreement with the usage of ST. Others follow yet another tradition and have a still fuller orthography revealing itself in such forms as No 312 N13 7287 (7 ?)

II. Grammar.

1. Variants suggesting harmonisation.

There is a large number of variants that consist of ST employing a common for a rare usage in MT, or a form or construction used elsewhere by both in identical or similar context. Each of these types of variation suggests the possibility of Samaritan harmonisation, and taken together the possibility of harmonisation becomes even stronger.

2. Variants attributable to scribal error.

Another group of variants may be considered as being due to the typical scribal errors which occur in textual transmission. Reference has already been made to this cause of variation under orthographical conclusions.

3. Variants arising from the indiscriminate use of grammatical forms or constructions.

This has given rise to one of the largest groups of variants in the Classification. ST and MT used certain forms or constructions/

construction indiscriminately e.g., 3 s.m. or 3 pl.verbs must or may be understood to have an indefinite subject. Both constructions were familiar to the scribes and one might be substituted for the other at almost any point. Accordingly, no cogent reason can be adduced in favour of ST or MT in these cases, so that either the ST or MT reading may have belonged to the original text as it existed at the time of the Samaritan secession.

4. Abnormal forms.

There is a relatively small but important group of variants consisting of abnormal forms that are yet found elsewhere in the Samaritan Pentateuch and that may be classed as early or late as well as under the foregoing categories.

5. Parallels outside Samapitan Pentateuch.

Morphological and syntactical variants not found elsewhere in Samaritan Pentateuch may be paralleled in MT of the Hebrew Bible or in Dead Sea Scrolls, and if so, may be considered as receiving strong confirmation. In the event of no parallel at all being found it may be wiser to suspend judgment rather than dismiss the Samaritan variant out of hand.

6. Doctrinal variants.

Grammatical variants of a doctrinal nature are considered in relation to the known doctrinal background of Jews and Samaritans. It cannot be too strongly emphasized that no passage/

passage which differentiates the location of the temple can be correct or incorrect merely on the ground of its saying that the location is in Jerusalem or Shechem, the reason being that a satisfactory logical case can be made for either location. Other considerations are to be noted, sometimes small in themselves but cumulatively more conclusive: there appear to be 10 of such passages in ST (involving 7 variants, as one occurs 4 times) where the text has been conformed to a specific Samaritan doctrine or practice.

7. Non-doctrinal variants.

Grammatical variants with divergences in sense. Any that show clarification of meaning, logical correction or euphemistic tendencies may reasonably be considered as secondary in nature. Again, any that show identity of meaning with a parallel elsewhere in the Pentateuch may have been harmonised: a completely different view is taken of variants paralleled outside the Samaritan Pentateuch. that is, in the MT of the Hebrew Bible for these may well be original on account of such strong independent confirmation. Variants in meaning that do not fall within the foregoing categories reveal a Samaritan tradition equally possible with that of the MT. In such cases, where the internal evidence is indecisive, the external evidence of the LXX may be valuable as lending support to one side of the other.

A full statement of <u>Non-doctrinal variants</u> with divergence in sense is given in Appendix A.

8. Collation with Dead Sea Scrolls.

The evidence is summarised under General Conclusions and shows grammatical variants few of which reveal any real difference of meaning but apparently arise from chance disagreement due to interchange of grammatical usage.

III. Vocabulary.

1. Substitutions.

i. Variants suggesting harmonisation.

There is a large group of variation where the ST variants that are substituted consist of pure synonyms. Where these consist of a common word instead of a rare one or a common word found elsewhere in identical or similar context, each type suggests the possibility of harmonisation and, taken together, such possibility becomes all the stronger.

11. Variants attributable to scribal error.

Of a different type are the synonyms originating in scribal error in ST or MT. As indicated under Orthographical and Grammatical Conclusions, each variant has to be judged on its own merits by the methods of textual criticism.

iii. Variants consisting of synonyms readily interchanged. A large group of variants consists of synonyms that could easily interchange in both ST and MT and where 1t/

it is obviously impossible to decide between one Pentateuch and the other. It is at least valuable to notice the high proportion of variation of this type where either the ST or MT reading is equally possible. There is also a small group of variants consisting of virtually synonymous words or phrases which give greater precision or emphasis. The readings of both ST and MT are so suited to the context that it is unwise to adjudicate between them.

2. Additions and/or Omissions.

1. Alternative, fuller forms of text in ST.

There are 10 long passages that are fully discussed under the relevant sub-heading in Chapter V where ST provides an explicit statement of events implied but not actually recorded in MT. Unlike most commentators the present writer is disposed to accept these passages as preserving a mode of writing that may point back towards the distant period of oral transmission, when a text might be written either in an abbreviated or a fuller form. See also below.

General Conclusions, No.8.

ii. Stylistic supplements.

A comparison of Additions and Omissions under Vocabulary shows clearly that ST, as compared with MT, has more stylistic supplements such as explicit subjects or objects./

objects, words in apposition or additional waw ("and"). As well as extra words of a stylistic nature there are those that are virtually synonymous expressions, perhaps giving greater emphasis, precision or clarity. These variants are numerous and include such common words as 30 at Art are 70.03 Many of these variants could reasonably belong to the original text but, cumulatively, they are significant as showing ST to be a fuller document, with extra words of this supplementary nature.

iii. Explanatory expansions.

There are many cases where extra words in ST appear to be more than stylistic supplements. While many of them show little difference in meaning they reproduce the actual wording of both ST and MT either in the immediate context or another part of the Pentateuch. They have every appearance of being expansions of an explanatory type.

iv. Doctrinal variants.

Variants of Vocabulary that are of a doctrinal nature are relatively few in number but their importance is obvious. They are all discussed in Chapter V where the controversy regarding Gerizim, and other Samaritan beliefs and practices, is treated in detail. As was emphasized under grammatical conclusions, no variant differentiating the location of the sanctuary in Zion or Gerizim/

Gerizim can be accepted or rejected merely as it stands for a good logical case can be advocated for either location. There must be a search for other considerations, perhaps inconclusive in themselves but cumulatively more cogent, before a decision is possible. Bearing these facts in mind, a case has been made out against the acceptance of the Samaritan 10th Commandment, Although no decision has been offered regarding the disputed variant $\rho(r, r, r) / (32)r$. Further, many of the doctrinal variants under "Other Samaritan beliefs and practices" may well show adjustment to distinctive Samaritan views. A further statement is made regarding doctrinal variants under General Conclusion, No. 11.

v. Non-doctrinal variants.

Variants of vocabulary showing clear divergence in meaning differ greatly in value. With some of them there appear to be cogent reasons in favour of the Samaritan reading. For example, there are certain passages in the MT where the text appears to be defective by the omission of a word or words and ST fills the "gap". The Samaritan reading gives such excellent sense that it seems reasonable to adopt it. Naturally these variants are not numerous: over the whole Pentateuch they amount to 16. In 15 of these cases the LXX supports ST; in the remaining one it diverges from both ST and MT.

In other examples the evidence is on the side of MT where, for example, ST appears to have made an attempt at clarification of meaning or logical correction.

There are many passages where the evidence is inconclusive and where no cogent reasons can be adduced in favour of ST or MT.

A full statement of <u>Non-doctrinal variants</u> with divergence in sense is given in Appendix A.

vi. Collation with Dead Sea Scrolls.

The evidence is summarised under General Conclusions and shows that many of the variants in the Scrolls are virtual synonyms of the type already noted under Additions in ST.

CHAPTER VII.

General Conclusions.

1. The Samaritan Pentateuch MSS witness to a homogeneous text tradition.

Unlike the Jews the Samaritans had no Council of Jamnia to fix the consonantal text of the Pentateuch and. as far as is known, no subsequent scribal activity comparable to the Masoretic development. Yet it cannot be denied that their Pentateuch has a uniform text. Apart from orthographical variation, such as plene or defective writing of yodh or waw. or mistakes with the gutturals, to which the Samaritans were prone, there is a unanimity among the many MSS used by von Gall and detailed in his critical apparatus. Even those Samaritan variants which consist of modifications or expansions are reproduced with precise exactitude throughout the Pentateuch MSS showing that, though of a secondary nature, their inclusion was long established and accepted by the whole Samaritan It is true to say with Kahle: "For many centuries community. the Samaritan Pentatouch has been a textus receptus without variant readings. We have seen that a text similar to it must have been used in Jewish circles also, in the time before all earlier texts were replaced by the authoritative Hebrew text." (See Cairo Geniza, p.149).

2. The Samaritan Pentateuch is not an amendation of MT but a descendant from a type of text earlier than the Masoretic.

Many of the Samaritan variants from MT go far beyond differences of orthography; and grammar and include modifications and long expansions. That the latter go back to an early time is a valid inference from the uniformity of the Samaritan Pentateuch tradition, for the Samaritan MSS reveal a homogeneous text. Presumably for many centuries the text has been fixed in all essential details. The antiquity of the Samaritan variants has received striking confirmation by the Qumran discoveries. As stated earlier, the Biblical MSS at Qumran reveal three variant text traditions as well as some of a mixed type. One is a text with additions and transpositions typical of the Samaritan Pentateuch. There is no evidence that this was a Samaritan document so that presumably it would belong to a type of text in general use at that time. With the establishment of the fixed consonantal text by the Council of Jamnia, such variant texts would be eliminated more and more among the Jews. The Samaritans, however, adopted a different standard text from the Jews so that their Pentateuch is valuable as witnessing to a different text tradition from that which ultimately became established as the MT. Such a conclusion would lend additional weight to the views of such scholars as Kahle who consider that the Samaritan represents a text

at one time widely expanded and in use among the Jews in pre-Christian times.

3. The text found in Samaritan Pentateuch Scrolls has not been more accurately transmitted than that in Samaritan Pentateuch Codices and neither does it represent a different tradition.

In refutation of the view that the Samaritan Scroll has a superior text to that of the Codex and that accordingly the Scroll, and not the Codex, should be used as the basis of comparison between ST and MT, detailed evidence has been led in Chapter III, 3, "Scroll and Codex", with the following results:

- i. The Scroll MSS are not unanimous in their witness to the text.
- ii. The same kind of variants occur in Scroll as in Codex.
- iii. They are just as numerous.
- iv. There appears to be no relation between the Scroll MSS to justify the view of a common tradition different from, or superior to, that of the Codex.
- 4. The Samaritan Pentateuch to a greater extent than MT may show evidence of oral, as well as written, tradition.

 In Chapter III, 2, "Oral Transmission", the writer put

forward certain arguments in favour of his view that many of the Samaritan variants may have arisen in the course of oral transmission wherein the possibility of error is even greater than in the case of written transmission e.g., many examples of spelling and accidence which make no difference to the meaning would not be differentiated if a text were being/

being recited. It is unnecessary to recapitulate all the arguments. Such a hypothesis would explain the smoothing out of difficulties or the correction of eccentricities such as are to be found in ST. Oral expansions in the teaching of the schools might well be reflected in its text.

5. There are certain types of variation which are due to misreading, miswriting or mishearing in ST or MT.

A large group of variants may be considered as having originated in the typical scribal errors that can occur in the course of textual transmissions. There is no general principle that can be applied in the criticism of these variants and each example has to be judged individually on its own merits. Discussion in the nature of the case can easily become subjective in particular examples but a broad, overall picture emerges showing that, apart from variants due to confusion with gutturals to which the Samaritans were specially prone, there are many variants that have every appearance of having arisen from typical scribal errors. Confusion between letters similar in shape or sound, dittography, haplagraphy, homocoteleuton and the like, either in ST or MT have been the root cause of divergence in certain cases.

6. There are certain types of variation which are more common in ST than in MT and which may be presumed on various grounds to be later than MT.

There are cases where ST employs a common for a rare form in MT,/

MT, in accidence, syntax or vocabulary, and where the Samaritan reading is presumably later.

There are also variants which show not only identity in content but even in actual wording with other passages in OT. Such variants differ greatly in value. Where the parallel passages are outwith the Pentateuch it is reasonable to believe they may furnish strong confirmation of the Samaritan reading. There are a small number of these. On the other hand, where the parallel passages are within the Pentateuch and the Samaritan variant reproduces the actual wording, as often happens, it would appear as if harmonisation has taken place. There are many of these Samaritan variants consisting of single words, short word groups, and longer passages extending through a number of verses. While individual examples of single words and short word groups do not provide conclusive evidence, their collective significance appears to do so and in the case of the long incorporations or amplifications from parallel passages the evidence is surely overwhelming.

Apart from the Samaritan Tenth Commandment (see also Conclusion 11) there is nothing specifically Samaritan about them in the sense of doctrinal difference and they presumably reflect a type of text that happened to be in the possession of the Samaritans but might have a wider circulation. This view seems to be strengthened by the fact that such an expanded and/

and adjusted text was found among the Qumran documents.

7. There are certain types of variation where on internal evidence no decision appears possible between ST and MT and where in certain cases external evidence may be adduced in support of either ST or MT.

There is a large number of variants where internal evidence does not permit a decision regarding the original reading. Such variants include those that involve no change of meaning and consist of interchange of grammatical constructions between ST and MT e.g. 3rd singular or plural verbs must or may be understood to have an indefinite subject. Both constructions were familiar to the scribes and one might be substituted for the other at almost any point and this has evidently There are also variants of Vecabulary consisting been done. of synonyms easily and actually interchanged in both ST and MT and where it is obviously impossible to decide between one Pentateuch and other. It is at least valuable to notice the high proportion of variation of this type where either the ST or MT reading is equally possible. There are also other variants which show difference of meaning equally as suitable to the context as that of MT. In such cases the support of the Versions may be adduced in favour of ST or MT: where the Versions are of no assistance it may not be possible to give a preference to either reading.

8. The Samaritan may preserve in certain passages an alternative. fuller form of text to that of MT.

Under/

Under Vocabulary, Additions, there is a full discussion of certain passages consisting of explicit statements in the Samaritan of events the occurrence of which is logically implied in MT. Good examples of this kind of variant occur in the long additional passages of the narrative of the plagues in Exodus: for example, in Ex. 716-18 God pronounces a warning regarding an impending plague and has commissioned Moses and Aaron to deliver the warning to It is not stated in MT that this warning was In ST, however, 7^{18b} represents Moses ever delivered. and Aaron as going to Pharach and repeating almost literally the words of the divine warning. There are 10 long variants of this nature all listed in Appendix A which contain statements of events logically implied but not definitely recorded in MT. While it is possible that these long passages may be later expansions, it can also be argued that they are additions of a different type from other long expansions in ST. They may be considered as belonging to the original text, being explicit statements of what actually occurred but is not actually mentioned in MT. If so, a most interesting possibility arises regarding the original form of the OT writings, some of which must lie very near to oral transmission: conceivably, a narrative might be recorded in full or in an abbreviated form, to be supplemented by the narrator.

9. The ST is a longer document than the MT.

The additions now to be considered are evidently of a different type from those treated in Conclusion 8. Under Vocabulary, Additions/Omissions it was shown that a distinctive feature of ST is the large number of synonyms, consisting explicit subjects or objects, words in apposition, or \mathbf{of} additional waw ("and"): or of virtual synonyms, consisting 30 of additional words such as זה and conveying ideas like precision, emphasis and clarity. Many of these variants could easily belong to the original text and it is impossible to accept or reject them as such but cumulatively they convey a picture that is significant, namely, that in this respect the Samaritan has a fuller text. There is also a large number of variants consisting of extra words or phrases, some of which convey greater precision and similar ideas and all of which reproduce the actual wording found elsewhere in the Pentateuch, in identical or similar context. Such variants may be viewed with suspicion on account of possible harmonisation by ST.

In addition to those groups there are long passages in Samaritan consisting of incorporations or substitutions from parallel narratives repeating almost literally whole verses found elsewhere in the Pentateuch. These passages are clearly of a secondary nature.

Accordingly, the resultant text of ST is clearly a longer and/

and fuller document with the possibility that in this respect it is an expanded and later production.

10. The Samaritan may contain dialect forms.

In Chapter I reference is made to Sperber's use of the LXX and the Samaritan Pentateuch to detect traces of a northern dialect of Hebrew. Though going back to a common ancestor the languages of ST and MT are separate branches each with a phonetic history of its own. He regards the kingdoms of Israel and Judah as the homes of these dialectic differences. The respective recensions of the Pentateuch handed down in these two areas were preserved almost unchanged from an early period in the dialects of these two districts. See above, p.22. Gerleman cannot accept this theory of a provincial dialect and holds the view that many of the Samaritan variants are due to a younger and more advanced stage of Hebrew of the Chronicles period. See above, p. 20.

With regard to these theories it may be said that neither of them can be accepted to the complete exclusion of the other. The theory of dialectic difference, if pressed to its logical conclusion, would imply that the Judaean history of Chronicles was written in a Hebrew reminiscent of the northern kingdom. On the other hand there are more factors in the situation than Gerleman has considered. Some of the Samaritan variants he adduces, such as absence of apocopation in verbal forms, are to be found in the MT Pentateuch.

According/

According to Millar Burrows he has over-simplified the problem. See above, p. 21.

The natural inference from this evidence is that traces of dialectic difference are to be observed in the two Pentateuchs for the following reasons:

- i. At a later date there is clear evidence of a Samaritan Hebrew showing Aramaic influence and the intrusion of what appears to be dialect forms.
 - (1) Occasional forms and syntax are evidently Aramaic.

(i) 2 s.m. Impv.	'17	passim
(ii) 3 s.f. Pf.Aphel	N 8 N	Gn. 33 ¹³
(iii)Pt.Pass.Peil	בעי3 ת בע	Gn. 20 ³
(iv)F.pl.nominal forms	החמירו∧	Gn. 27 ¹⁵
	ושדיפּλ	Gn. 41 ²³
	ומח גיאין	Ex. 27 ³
	ומחגיא'ה	Ex. 25 ³⁸
	ו מח ג' ג'ה	Ex. 37 ²³

CI. NA'AAA, AA'AAA

- G.H.Dalman, Aramaisch-Neuhebräischer Wörterbuch, Frankfurt, 1901, p.221.
- (2) Uncertainty in writing the gutturals implies a weakened pronunciation wherein they were not distinguished or were all pronounced as aleph.
- ii. Regarding the earlier period.

(1)

- (1) There is the clear and unambiguous statement in Judges 12^6 of the fact of an Ephraemitic pronunciation of 32ω differing from the South.
- (2) Some of the Samaritan variants such as 2 s.f. personal pronoun '^ are commonly agreed by scholars to be older than the customary MT form and are found elsewhere in the Kethib.
- (3) There is cogent evidence for a Hebrew different in pronunciation from that of MT. Petermann, Kahle,

 Diening and Sperber have all been considered as bringing forward evidence in support of this view.

11. Doctrinal variants are to be considered in relation to the known background of Jews and Samaritans.

The doctrinal variants, which are fully treated under the relevant sections in Grammar and Vocabulary, have been considered in relation to the known background of Jews and Samaritans. This was one of bitter controversy that developed regarding the claims of the rival sanctuaries on Gerizim and Ebal. A doctrinal variant differentiating the location of the temple cannot be considered as correct or incorrect merely on the ground of its statement that the location is in Jerusalem or Shechem for a satisfactory logical case can be argued for either site. There must be other considerations if a decision is to be reached: such considerations may in themselves be inconclusive but in conjunction with others the cumulative result may permit

of a reasonable conclusion or at least a strong presumption. It is the view of the present writer that a good case can be put forward for the ST reading of Gerizim for Ebal in Dt.27⁴ but he would find it difficult to defend the authenticity of the long passage, known as the Samaritan Tenth Commandment, in which that reading is incorporated. For discussion, see above, p. 107 "Controversy regarding Gerizim".

12. The Dead Sea Scrolls have many variants of the same type as the Samaritan.

The evidence of the DSS, which has been discussed in Chapter VI, is valuable for the light that it sheds on the pre-Masoretic period, as the Scrolls sometimes agree with MT, sometimes with ST, and sometimes follow an independent line. In Orthography there are Scroll variants which show agreement with MT, more frequently with ST but in great majority of passages available diverge from both MT and ST. The orthographic position is that DSS is fuller than ST which in turn is fuller than MT. In Grammar variants arise from the most part from chance agreement or disagreement due to interchange of grammatical usage between MT, ST, or DSS. Only a few of them show any real divergence in meaning. In Vocabulary the Scrolls support either the MT or ST in a limited number of cases but frequently diverge from both by having extra words giving/

giving greater precision, emphasis, clarity and the like.

However, these extra words are precisely the same kind

of extra words that make ST a longer document than MT.

In so far as the Pentateuch fragments from Qumran are

representative samples of a much longer corpus of material,

many of them reveal a type of text fuller than either ST or

MT.

APPENDIX A.

List of Samaritan variants under the categories of the Classification outlined in Chapters IV and V.

(<u>NOTE</u>: The variants are either adopted by von Gall in his text or are strongly attested in his critical apparatus. Where the Scripture reference is not sufficient to identify the variants, the actual Hebrew is added).

I. DIFFERENCES OF ORTHOGRAPHY.

- 1. Variants attributable to scribal error.
 - 1. Letters similar in sound
 - (1) Gutturals.
 - (1) Omission
 Gn.13¹⁸ 3 21 20¹⁷ 110 11 43¹⁰ 115 75 75 77
 - (ii) Addition

(111) Interchange

Interchange (contd.)

Gn. 32²⁵ panı f and N

n and y Gn. 497 017211

N and 7 Gn. 14²³MYW 19²⁹ TONT 22¹⁷ TENNI 2921 427 3816 427

Nu. 1419 71N 3221 17.41N

Dt. 41 MAYI

 π and \aleph

Gn. 4147'NO71612 17517210'F7 198 N'S 17 2124 Y2 WT בוה 37³⁰ הפן 24²⁵ אש קאה 124⁴⁶ אם חספה 37³⁰ הוה 4227,2004324 ,200499, 23215017 718

Ex. 2832 774 3923 7744

Nu. 1120 77132324 71232249712321

Dt. 2120 72101 3320 71235

(iv) Transposition

Gn. 2530 115377 412 714 714 41 187171

Ex. 156,7471

Examples not so strongly attested.

(i) Interchange of guttural

Gn. 2¹²0161 25⁴ 15'N

(ii) Additional guttural, sometimes with metathesis Gn. 4221 7757 (11 MSS)
Gn. 4221 7757 (3 MSS)

(iii) More than two gutturals

Gn. 4333 17771 (3 MSS, 1 Cour), 17771 (3 MSS), 17771 (1MS)

(2) <u>Palatals</u> Gn. 21²³ ידגי 3 Lv.11¹⁹ אסיגים Dt. 14¹⁸ אסגדה

(3) Sibilants

Gn. 18²⁰ A 244 Ex. 2²³ (245')

(4) Labials

ii. Letters similar in shape

(1) Within ST script

(2) Within MT script

T and T

(3) Within ST and old Hebrew script

111. Other typical scribal errors, such as incorrect joining or dividing of words, laplography or dittography, omission or intrusion of letters or words.

iv. Formae Mixtae

Ex.
$$15^1$$
 17 wh 20^{24} 17 ord Lv. $15^{23}_{\pi^2}$ $_{25}^{22}$ $_{10}$ $_{12}$ $_{12}$ $_{13}$ $_{13}$ $_{14}$ $_{13}$ $_{15}$ $_{13}$ $_{15}$ $_$

v. Proper Names, attributable to scribal error and classifiable under the variant types of this first section of orthographical variants, and entered here as a group for convenience of reference. (See below, pp. 159 and 180 for other examples of Proper Names)

$$Gn.10^{7}_{72}A_{201}11^{31}$$
 | 7712^{4} | 7710 12⁵ | 7710 12⁶ | 7710 14² 801.10^{7}_{14} | 14^{5} | 14^{5} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{17} | 14^{1

$$Nu.10^{12}$$
 אונ. 11^{35} פראן 13^{6} פראן 13^{6} פראן 13^{2} אור פראן 11^{30} פראן 13^{2} אור פראן 13^{2} אור פראן 13^{2} פראן 13^{2} אור פראן 13^{2} פראן 13^{2} אור פראן 13^{2} אור ארגב 13^{2} פראן 13^{2} אור פראן 13^{2} אור פראן 13^{2} אור פראן 13^{2} אור ארגב 13^{2} אור ארגב

vi. Passages too corrupt in both ST and MT for any useful comparison.

Variants making the pronunciation explicit. 2.

i. Use of Aelph.

(1) Elision (Syncope)

In ST Gn. 3¹³, ωπ18⁶ αιο 24¹⁷, ιη πη 24²¹ πλωη 30¹⁴ αιτι τ30¹⁵ ιτιτ (bis)
30¹⁶ ατιτ 41² λιτι 41² λιτι 41⁵ λιτι 2π λιτι 3η πι 3η πι 3η πι 41 18 17241 22 3n49 27 2'T

Elision (contd.)

In MT

Gn. 2524 D'DNA 4151 JNO 1

Ex. 2624 D' NAN 3629 D' NAN

Lv. 11430ANDUI

Nu. 1111 'ANSA 1524 ANDA 3

Dt. 11 12 NOW 21 23 WISA 2857 NAS T 29 22 NAS I 32 32 WAY

Proper Names, showing elision of aleph.

In ST

Gn. 2224 7117 251332741

Ex. 624 90'241 Dt.43 AIN

Nu. 13¹⁵ 3H12 26¹⁷ (317H3)

Gn. 142 DINAS 148 DINAS 284 AINAI 363 AINAI 467 MINII 4613 NIDI

Dt. 29²²0'N25132⁴⁹ N21 34¹N21

(2) Vowel Letter

(1) Media@

In ST

Gn. 37אפרח 22 אפרח 22 אפרח 22 אפרח 23 איידאן 22 יחידאן 22 איידאן 40 אחד 37 אפרח 40 אחד און 22 אפרח 238 איידאן 40 איידאן 40 ארביאבון 4238 איידאן 40 איידאן

Ex. 1237. NS27 15803TN1 2411 TTAN.

Nu.1121 ארבלה 11 14 רגנאו

Dt. 2870'0 PT 336 'AND

(ii) Final Corresponding to verbs in MT.

Gn. 14^{14} $\mu_{2\omega_{1}}$ 24^{12} μ_{777} 24^{41} μ_{71} μ_{77} μ_{777} μ_{777}

Lv. 115 NEDIL 59 NED. 1345 NOY.

(11) Final (contd.)

Nu.
$$5^{21}$$
 $h_{2}s$ 11^{23} $7h_{7}$ 13^{18} h_{9} 11^{18} 21^{18} $21^{$

(3) Final (prosthetic) after a long vowel

In ST

In MT

- ii. Use of He To indicate presence of final vowel
 In ST
 - (1) <u>Pronoun</u> 2 s.m.

 Nu. 11¹⁵ 7^\square Dt. 5²⁴ 7^\square
 - (2) <u>Suffix</u> 2 s.m.

(3) Verbs Pf. 2.s.m.



Impf. 2 pl.f.

Ex.
$$1^{18}$$
 $\pi J'' \pi \Lambda I - I MS, |' \pi \Lambda I - I MS) 3 pl.f.$

$$35$$
 אהרנה 36 ותשקינה 30^{35} ואשקינה 36^{35} ואקינה 30^{35} ואקינה 30^{35} ואקינה 30^{35} ואקטנה 30^{35} ואקטנה 30^{35} ואקטנה 30^{35} ואקטנה 30^{35} ואקטנה 30^{35} ואשאחוינה 30^{35}

Ex.
$$1^{17}$$
 71 117 71 117 71 119 119 117 119

Note: Examples ending in 7— attested by a number of Samaritan MSS in von Gall's critical apparatus, but not adopted by him -

In MT

1

Gn.
$$3^9$$
 7'N 27^7 7572NI 27^{37} 731
Ex. 7^{29a} 721 13^{16} 7'7' 15^{11} 7'7066)29³⁵ 7NN
Nu. 22^{33} 7NN
Dt. 28^{22} 7NN
Gn. 41^{19} 3 pl.f.

(3) <u>Verb</u> pf. 2 s.m. Gn. 31³⁰ A2021

iii. Use of Yodh or Waw.

- (1) Greater use of vowel letters as compared with MT.
 - (i) General observations.
 - a. Evidence of Samaritan pronunciation different from that of MT.

Yodh or Waw in ST not homogeneous with

(a) vocalic consonant in same syllable in MT $6n.40^{10}$ $6n.40^{10}$ $6n.40^{12}$ $6n.40^{12}$ $6n.40^{10}$ $6n.40^{10}$

(b) pointing in MT

 $6n.3^{21}$ 1.7.5 1.2^{23} 1.7.5 1.3^{2} 1.5 1.4 1.3^{2} 1.6 1.3 1.6 1.3 1.6 1.3 1.8 1.4 1.5

 $\mathrm{Ex.1}^{14}$ אשכינאה 3^{22} בחימר 2^3 בחימר 3^{22} או געים, 3^{22} בכבודוא 3^{14} הכנים, 3^{25} או הכנים, 3^{25} או הכנים, 3^{25}

 28^{39} $\lambda_1\lambda_1 > 28^{40}$ $\lambda_1\lambda_1 > 29^{5}$ $\lambda_1\lambda_1 > 30^{28}$ $\lambda_1\lambda_2 > 30^{23}$ $\lambda_1\lambda_2 > 30^{24}$ $\lambda_1\lambda_2 > 30^{21}$ $\lambda_1\lambda_2 > 30^{27}$ $\lambda_$

Nu. 21 8 717 ω 26 10 013 3 Dt. 4 ω 28 8 7000 23 9 70700

b. Vocalic consonant where MT has short vowel.

c. Vocalic consonant where MT has vocal shewa.

(11) Selected Classes.

a. Nouns

(a) Segholate forms.

(b) Plural Termination.

For convenience of reference, adjectives and participles, with like terminations as nouns, are also included here.

Scriptio plena

Masc. abs.

$$6n.1^{21}$$
 3^{7} 17^{20} 18^{24} (bis) 18^{26} 18^{28} 21^{15} 25^{16} 25^{24} 32^{16} 36^{24} 39^{22} 40^{10}

Ex.
$$7^{12}$$
 8^{10} 8^{12} 8^{15} 9^{11} 14^{7} 15^{11} 25^{5} 26^{14} 26^{27} 26^{29} (bis) 29^{1} 29^{3} 35^{7} 35^{23} 35^{27} 36^{19} 36^{29} 36^{32} (bis) 36^{34} (bis) 39^{3} 39^{34}

(b) Plural Termination (contd.)

 $Lv.10^{16}$ 16^7 16^8 (bis) 17^7 19^4 21^6 23^{18} 26^1 $Mu.2^{13}$ 39 65 7^{10} 7^{17} 7^{87} 7^{88} 17^{20} 21^{29} 23^{29} 27^2 29^{13} 29^{14} 29^{17} 29^{18} 29^{20} 29^{21} 29^{23} 29^{24} 29^{26} 29^{27} 29^{29} 29^{30} 29^{32} 29^{33} 33^{53}

Dt.48 1619 322 3233

Masc. with sf. (Unusual onthography in M T)

מקניהם 1777 ומחצביהן 24 מיניהם 177

ואסריה 30 ונסכיהם 29 1 שיאיהם 17 חדשי בס 33

Fem. abs.

 $Gn.1^{14}$ (bis) 1^{15} 1^{16} 3^7 5^4 6^2 8^{20} 12^{16} (bis) 19^{15} 20^{14} 21^{29} 21^{30} 23^{15} 24^{11} 24^{13} 24^{35} 24⁵³ 25⁶ 26³ 26⁴ 26^{15} 26^{18} 27^{15} 27^{23} 32^{16} 33^{17} 41^2 41^4 41^5 41 18 41 20 41 22 41 22 41 24 41 26 41 26 41 27 41^{27} 41^{35} 42^{29} 45^2 45^2 45^{23} 45^{23} 47^{24} 49^{13} Ex.115 117 118 119 119 119 119 120 121 27 3²² 4¹⁷ 4²⁶ 4²⁸ 4³⁰ 8^{1a} 8⁹ 8⁹ 9⁹ 9¹⁰ 9²³ 9^{32} 9^{34} 10^{35} 12^{3} 12^{4} 12^{18} 12^{34} 12^{35} 13^{6} 15^5 15^5 15^8 15^{20} 16^{12} 18^8 18^{21} 19^{16} 20^5 20¹⁸ 20²⁶ 22¹⁶ 23² 24⁵ 24⁶ 25¹² 25¹⁴ 25²⁶ $25^{26} \ 25^{27} \ 26^{1} \ 26^{2} \ 26^{3} \ 26^{3} \ 26^{3} \ 26^{5} \ 26^{5}$ 26^5 26^5 26^6 26^7 26^8 26^9 26^9 26^{10} 26^{10} 26^{11} 26^{17} 26^{24} 30^4 30^7 30^8 27^7 27^8 28^7 28^7 28^{14} 2814 2820

(b) Plural Termination (contd.)

Fem. abs. (contd)

 $Ex.28^{24}$ 28^{25} 28^{40} 29^2 29^2 29^2 29^8 29^9 29^{13} 29^{22} 31^4 32^6 32^{15} (2 mo) 32^{16} 32^{16} 32^{19} 32^{19} 34^{1} 34^{10} 34^{22} 34^{28} 35^{32} 35^{35} 36^{8} 36^9 36^{10} 36^{10} 36^{12} 36^{12} 36^{12} 36^{12} 36^{13} 36¹⁴ 36¹⁵ 36¹⁶ 36¹⁶ 36¹⁷ 36¹⁷ 36²¹ 36²² $\frac{36^{22}}{36^{29}}$ $\frac{37^3}{37^5}$ $\frac{37^{13}}{37^{13}}$ $\frac{37^{13}}{36^{14}}$ $\frac{36^{15}}{36^{16}}$ 36^{16} 36^{17} 36^{17} 36^{21} 36^{22} 36^{22} 36^{29} 37^3 37^5 37^{13} 37^{13} 37^{14} 38^3 38^3 38^3 38^5 38^7 38^7 38^8 38^{20} 39^4 39^4 39^6 39^6 39^{13} 39^{16} 39^{17} 39^{18} 39^{18} 39^{28} 40^{14} 40^{25} Lv.24 24 34 310 310 315 315 49 74 74 712 7^{12} 8^{13} 8^{16} 8^{25} 9^{10} 9^{19} 11^{3} 13^{38} 13^{38} 13^{38} 13³⁹ 13³⁹ 14³⁷ 14³⁷ 14³⁷ 18²⁶ 18²⁷ 18²⁹ 18²⁹ 18^{30} 19^{31} 20^6 23^{15} 23^{42} 23^{42} 26^{16} 26^{46} 27^2 Nu.1²⁶ 1²⁸ 1³⁰ 1³² 1³⁶ 1³⁸ 1⁴⁰ 1⁴² 2² 2¹⁷ 2^{32} 3^{31} 3^{43} 4^{7} 4^{7} 4^{7} 4^{14} 4^{14} 4^{14} 4^{32} 5^{23} 6^{15} 6^{15} 7^2 7^6 7^8 7^{86} 8^2 10^9 10^{10a} 10^{25} 12^{8} 13^{28} 16^{17} 16^{22} 17^{2} 17^{22} 17^{24} 23^{1} 23^{14} 23^{29} 24^{6} 27^{7} 27^{16} 32^{38} Dt.115 128 128 35 321 434 434 443 68 610 610 6^{20} 7^{19} 7^{19} 8^{7} 9^{1} 9^{1} 9^{17} 10^{2} 10^{2} 10^{3} 10^{4}

(b) Plural Terminations(contd:)

Fem. abs. (contd.)

 p_{t} , 10^{5} 10^{21} 10^{21} 11^{7} 11^{11} 11^{18} 16^{9} 16^{13} 17^{8} 20^{15} 27^{6} 28^{52} 29^{2} 29^{2} 29^{28} 32^{16} 32^{24} 32^{35} 34^{11}

Fem. cstr.

 $cm.6^{5}$ 7^{11} 8^{2} 8^{2} 9^{12} 10^{32} 12^{3} 14^{10} 14^{10} 21^{11} 21^{28} 26^{18} 27^{16} 28^{14} 41^{2} 41^{4} 41^{4} 41^{6} 41^{18} 45^{22} 46^{2} 46^{7}

Ex.1¹⁵ 6^6 6^{14} 6^{15} 6^{19} 6^{24} 9^{28} 12^{39} 15^{27} 16^{12} 18^8 25^{12} 25^{14} 25^{15} 25^{26} 26^4 26^7 26^{12} 26^{13} 26^{14} 26^{23} 27^4 27^7 27^{19} 28^{11} 28^{12} 28^{13} 28^{14} 28^{14} 28^{21} 28^{24} 28^{38} 29^{12} 31^{18} 35^7 (2^{28}) 35^{18} 35^{18} 35^{25} (2^{28}) 36^{11} 36^{14} 36^{19} 36^{28} 37^3 37^5 37^{13} 37^{27} 38^7 38^8 38^{31} 39^6 39^7 39^{13} 39^{14} 39^{15} 39^{16} 39^{16} 39^{16} 39^{17} 39^{18} 39^{19} 39^{20} 39^{20} 39^{34} 39^{37}

Lv.4¹⁸ 4²⁵ 4³⁰ 4³⁴ 14⁹ 14³⁷ 14³⁹ 20²³ 21¹¹
23³⁸ 23⁴⁰ 25⁸ 26¹³ 26³⁶ 26³⁹ 26³⁹ 26³⁹
Nu.3²⁰ 3²¹ 3²⁷ 3³⁰ 3³³ 3³⁵ 4²⁵ 7³ 7⁸⁴ 10² 21¹⁴
26⁷ 26¹² 26¹⁴ 26¹⁸ 26²² 26²⁵ 26²⁷ 26³⁴

(b) Plural Terminations (contd)

Fem. cstr. (contd)

Nu.
$$26^{37}$$
 26^{42} 26^{43} 26^{47} 26^{50} 26^{58} 27^1 32^{16} 32^{36} 32^{38} 33^9 33^{16} 33^{17} Dt. 7^{13} 26^{51} 32^8 32^8 32^{32} 33^2

Fem. with sf.

Gn.
$$8^{19}$$
 10^5 10^5 10^{20} 10^{20} 10^{20} 10^{31} 10^{31} 10^{31} 10^{31} 15^5 17^7 17^9 17^{12} 19^{15} 25^{13} 25^{16} 25^{16} 26^5 31^{26} 31^{28} 31^{43} 31^{43} 31^{50} 34^{21} 36^{40} 36^{40} 36^{43} 44^{13} 47^9 47^{30} 48^{15} 48^{16} 48^{21} 49^{29} 50^{25}

Ex.1¹¹
$$2^{11}$$
 3^{20} 4^{5} 5^{5} 6^{17} 6^{20} 6^{25} 7^{3} 7^{12} 7^{19}
 10^{2} 12^{21} 12^{13} 12^{21} 12^{34} 25^{36} 28^{10} 28^{20}
 28^{21} 31^{16} 37^{22} 39^{13} 39^{14}

Lv.
$$10^5$$
 19^{19} 19^{37} 20^8 20^{22} 25^{18} 26^3 26^{15} 26^{40} 26^{43}

(b) Plural Termination (contd.)

Fem. with sf. (contd.)

Nu.10³¹ 11¹⁰ 14²² 14²³ 17³ 17¹⁷ 17²¹ 17²⁵ 26²
$$26^{15}$$
 26^{20} 26^{23} 26^{26} 26^{28} 26^{35} 26^{37} 26^{38} 26^{41} 26^{42} 26^{44} 26^{48} 26^{50} 26^{55} 26^{57} 33^{52} 33^{54} 34^{14}

Dt.1011

Scriptio defectiva

Gn.
$$31^3$$
 32^1 35^5 46^{15} 47^3
Ex. 3^{13} 7^{28} 12^{17} 25^{19} 25^{29} 27^4 28^7 37^8 39^4
Lv. 22^{18} 23^{38}

Nu-33⁵²

Dt.614

Note: unusual orthography in S T Gn. 31 31 7A(12 3141 7A(12

unusual orthography in M T

מנאינן ²⁰וצרסאיכם 15²⁰ אנאינן

b. Pronouns

Suffixes 2 s.f. Scriptio plena always in S T $6n.12^{12}$ (2 exx) 12^{13} (2 exx) 13^{16} (4 exx) 16^6 (2 exx) 16^9 16^{10} 16^{11} 20^{13} 21^{18} 24^{14} 24^{17} 24^{43} 24^{45} 24^{60} 25^{23} 30^{14} 30^{15} (3 exx) 35^7 38^{18}

b. Suffixes (contd.)

Ex.27 29

 $Nu.5^2$ 5^{19} (2exx) 5^{20} (2 exx) 5^{21} (3 exx)

c. Verbs

(a) Verbal ending before suffix

Scriptio plona

Gn. 1 28 1022124 54 1178 4267 112771

Ex. 174,137013918 111113920 11111

Lv.175 מו איבותו

ויצ כדות 32 בהקריבום 17 תיראום 39 איני אום 39

י עזרו מ 22 ווים שוב 22 ויירשום 22 ויירשום 23 ויירשום 28 ויירשום 29 ויירשום 2

(b) Participle

Scriptio plena Way in first stem syllable

Gn.211 918 32³² 34³⁰

Ex.2¹⁴ 5⁶ 5¹⁰ 5¹⁴ 5¹⁵ 5¹⁹ 13⁴ 14⁸ 25³² 25³³

2535 3718 3719 3721

Lv.2510

(b) Participle (contd)

Scriptio Plena $\frac{\sqrt{3}}{2}$ in first stem syllable Nu.11¹⁶ 16¹¹ 21¹³ 21^{22b} (bis) 26⁴ 30³ 31²⁷ 31²⁸ 31³⁶ 32¹⁷ 32²⁴ 33⁵² 36⁴ Dt.1⁴ (bis) 2⁴ 2⁸ 2²² 2²³ (bis) 8⁷ 9²¹ 21²⁰ (bis) 33¹² 33²¹

, Scriptio defectiva

Gn.1³⁰
$$8^{19}$$
 15^{2} 24^{3} 24^{62} 25^{32} 28^{20} 37^{25} 39^{6}
 40^{2} 41^{8}

Ex.3⁵ 8^{16} 8^{25} 9^{3} 11^{4} 18^{14} 19^{19} 30^{35} 36^{29}

Lv.5¹⁶ 11^{27} 11^{42} 15^{10} 15^{27}

Nu.2¹² 13^{19} 13^{29} (ter) 14^{25} 21^{34} 24^{14}

Dt.3² 4^{42} 4^{46} 13^{4} 18^{10} 18^{14}

(c) Infinitive absolute

. Scriptio plena $\underline{\text{Waw}}$ in second stem syllable Gn.37^8

Scriptio defectiva

Gn.
$$8^7$$
 12 9 26 13 31 15 37 8 37 10 43 7 Ex. 15 26 19 5 19 13 21 22 21 28 23 22 Lv. 10 18 13 2 24 16 Nu. 6^{23} 13 20 ut. 6^{17} 15 2 15 5 15 10 21 13 28 1

(d) Niphal

Scriptio plena Yodh in second stem syllable

Gn.4¹⁴ 18⁴

Ex.34¹⁹

Lv.613

Nu.784 788

Dt.29²²

(e) <u>Hiphil</u>

Scriptio plena Yodh in second stem syllable

Pf.

 $Gn.39^1 42^8 43^3$

Nu.7¹⁹ 17¹³ 26⁵⁸ 30¹³ 30¹⁵

Dt.2³⁰ 2³⁴ 7¹⁹

Pf. with Waw

 $Gn.34^5 37^{20}$

Nu.4⁵ 5¹⁶ 30¹²

Dt.214 2214 2811

Impf.

 $Gn.4^{12}$ 8^{21} 19^7 19^9 44^{23} 49^4

 $Ex.9^{28}$ 13^{11} 14^{13} 14^{14} 15^{17} 16^{19} 20^{12} 22^4 22^{12}

Lv.24 13¹¹ 19²⁷ 27¹⁴

 $Nu.6^{10}$ 6^{25} 11^4 11^{18} 30^{13} 30^{14} 30^{16} 32^5 35^{19}

Dt.1¹⁷ 2²⁸ 4² 6² 9²⁶ 17¹⁶ 19²⁰ 29²²

(e) <u>Hiphil</u> (contd)

Impf. with Waw

$$Gn \cdot 2^{15} \cdot 2^{22} \cdot 3^{21} \cdot 5^3 \cdot 5^4 \cdot 5^6$$

$$5$$
7,9,10,12,13,15,16,18,19,21,22,25,26,28,30,32 6 10

$$19^{26}$$
 21^{15} 24^{18} 24^{20} 24^{28} 24^{37} 24^{46} 24^{67} 27^{27}

$$29^{12}$$
 31^{42} 37^5 37^8 37^{24} 41^{14} 43^7 43^{21} 44^{11}

$$Ex.2^{10}$$
 4^{6} 4^{7} 7^{10} 16^{20} 19^{4} 32^{19} 32^{24}

$$16^5 18^{25} 20^{26} 26^{13}$$

$$\text{Nu.14}^{12} \ 16^{10} \ 20^{15} \ 20^{16} \ 31^{12} \ 31^{50} \ 31^{54}$$

Dt.1²² 1²⁵ (bis) 1⁴³ 2³⁴ 4¹⁰ 4³⁷ 5¹⁵ 8³ 9²¹
$$26^6$$
 26^8 26^9 32^7

Imv.

$$Gn.4^{23}$$
 8^{17} 19^{12} 20^{7} 27^{25} 35^{2} 37^{14} 38^{3} 43^{16} 44^{21}

$$Nu.3^6 17^{25}$$

Pt.

(e) <u>Hiphil</u> (contd)

Inf. estr.

Inf. abs.

Scriptio defectiva

Pf.

$$6n.40^{10}$$
 43^9 44^4 45^{26}

Pf. with Way

$$Ex.4^8$$
 $Lv.26^9$ 26^{33} $Nu.14^{24}$ 14^{31}

Impf.

Dt.
$$3^{28}$$
 9^3 14^{28} 19^3 19^7 24^{11} 25^3 25^3 28^{38} 28^{66}

Impf. with Waw

$$Gn.19^{16}$$
 26^{31} 31^{45} 37^{33} 42^{29}

(e) Hiphil (contd)

Imv.

Pt.

Inf. cstr.

ut.2924

The Weak Verbs

(f) Verbs with inital Yodh

Scriptio Plena

Hiphil

Pf.

Gn.12¹⁶ 20 ... Nu.10²⁹1326.71 10³²1326.71

Impf.

Ex. 29 [77]'AL Dt. 32 13 [77]"

Pt.



Inf. abs.

Dt.921 בטייז 1315 ביים 174 בטייז 1918 ביים 278 בטייז 278

(f) <u>Verbs Initial Yodh</u> (contd)

Scriptio defectiva

Highil

Impf.

Gn.47 2'bA

Inf. csty.

Ex. 507 בהטיבן 20.54 בהטיב 10.54 בהטיבו 30 Bx. 816 בהטיבו

(g) Verbs Medial Yodh

Scriptio plena

Qal

Impf.

Gn. 24⁴⁷ סישימה 31³⁴ סישים 48²⁰ אייסן 48²⁰ און שימה 10⁵ און 10⁵

Qal

Impf.

Gn. 925 1001 4133 17101 4332 1001 458 1001

Ex. 111 1001 151 701 151 1700 1718 1001 3919 001

I.v. 20 1 1001

Mu. 1618 1501

Dt.28⁶³ ~~

Folel

Impf.

Dt.32^{1.0} 17112'1

(h) Verbs Final Yodh

Scriptio plena

Inf. estr.

S T ending in A(____

(h) Verbs Final Yodh (contd)

Scriptio Plena

Inf. cstr. S T ending in Al

$$g_{n.6}^{19}$$
 11^{5} 11^{8} 18^{25} 23^{2} 24^{19} 24^{19} 24^{30} 27^{1} 33^{10} 35^{17} 41^{32} 50^{20}

Ex.
$$1^{11}$$
 3^{8} 7^{24} 9^{28} 10^{3} 10^{5} 10^{28} 12^{48} 13^{17} 14^{13}
 15^{23} 17^{1} 17^{7} 19^{16} 19^{25} 19^{24} 23^{1} 23^{2} 23^{2}
 26^{13} (26) 30^{8} 27^{20} 31^{18} 32^{12} 33^{20} 34^{24} 35^{1} 35^{32}
 35^{34} 36^{1} 36^{2} 36^{3} 36^{5} 36^{18} 39^{21} 40^{15} 40^{37}

IN. 4^{27} 8^{34} 9^{22} 10^{11} 11^{45} 14^{57} 24^{2} 26^{13} 26^{44}

Nu. 1^{51} 4^{15} 5^{22} 8^{2} 9^{4} 9^{6} 9^{22} 16^{31} 22^{23} 22^{25} 30^{14}

Dt. 1^{4} 1^{26} 1^{33} 1^{33} 1^{41} 4^{14} 5^{1} 6^{24} 7^{22} 8^{2} 8^{2} 8^{2}

(i) Verbs Initial Waw

816 1122

Scriptio Plena

Gal.

Impf.

Pf.

Gn. 2016 Anoin

Pt.

Ex. 12 10 74171

Hiphil

Pf.

(i) Verbs Initial Waw (contd)

Hiphil

Pf.

Gn. 24 14 Anoin 2444 noin

Impf.

Gn. 412 701A 821 701A 1916 17124 18 771A 4423 10.01A

Ex. 928 10101028 101 1029 401 1413 10011

Nu. 13 12 1115 36 NISTI 165 YTII 1723 NSII 1724 NSII 2016 14511

Hophal

Inf. cstr.

Gn40²⁰ A7317

Scriptio defectiva

(le.1

Imp.

Gn. 27 7511924 9711 2816 9711 414 9711 417 971 4121 9711

(j) <u>Verbs Medial Way</u>

Scriptio Plena

Qal

Impf. Waw in stem syllable Gn.3 3 3 4 24 42 27 21 27 31 32 7 33 14 34 22 44 13 45 20 49 10

(j) <u>Verbs Medial Waw</u> (contd)

Impf. <u>Way</u> in stem syllable (contd)

 $\mathbb{E}_{\mathbf{x}.\mathbf{3}^{3}}$ $\mathbf{14^{2}}$ $\mathbf{14^{26}}$ $\mathbf{14^{28}}$ $\mathbf{18^{15}}$ $\mathbf{18^{23}}$ $\mathbf{31^{3}}$ $\mathbf{22^{8}}$ $\mathbf{23^{27}}$ $\mathbf{25^{15}}$ $\mathbf{29^{30}}$ $\mathbf{30^{20}}$ $\mathbf{30^{21}}$

Lv.10⁶ 10^9 11^{32} 12^4 14^{36} 14^{36} 14^{48} 15^{31} 16^2 16^3 21^{11} 21^{23} 25^{10} 25^{13}

Nu. 4^{19} 6^{10} 13^2 13^{21} 13^{25} 14^{35} 14^{45} 16^{29} 17^{25} 18^3 19^7 20^{24} 26^{65} 27^{17} 30^8 31^{23} 32^{22}

Dt.1¹⁹ 1²² 1³⁷ 1³⁸ 2¹⁹ 5²⁹ 17¹⁴ 18⁶ 18²² $19^{5} 20^{5} 20^{6} 20^{7} 20^{8} 23^{2} 23^{3} 23^{4} 23^{11} 23^{2}$ $23^{25} 23^{26} 24^{5} 24^{10} 27^{3} 29^{1} 33^{6}$

Imv.

 $Gn.43^2$ 44^{25} $Ex.14^{19}$ $Dt.2^{13}$ 32^{50}

Inf. cstr.

Gn.19³⁵ 48⁷

 $Ex.12^{23}$ 14^{12} 33^9

Lv.11³¹ 11³²

Nu.6⁷ 13^{21} 33^{39} 33^{40} 34^{8} 35^{6} Dt.9¹ 11^{31} 20^{19} 34^{7}

Niphal

Pt.

Ex.19¹¹ 19¹⁵ Dt.1¹³

(j) <u>Verbs Medial Waw</u> (contd)

Polel

Impf.

pt.326

Scriptio defectiva

Qal

Impf. Gn. 24⁵⁴ Dt. 4¹⁷ 28⁴⁰

Imv. Gn. 24³¹ 26³

Inf. cstr. Gn.34¹⁵ 38²³ 39¹⁶ 43²⁵ Lv.14⁴³ 25²² Dt.30⁹

Niphal

Pf. Gn.1726

Polel

.Impf. Gn.1²⁰

d. Particles

(a) $\wedge \wedge$ (acc.) ST always has Scriptio defective $6n \cdot 17^2 \cdot 20^6 \cdot 39^9 \cdot 40^{19} \cdot 41^8 \cdot 41^{39} \cdot 49^{28} \cdot 49^{29} \cdot 50^{21}$

Ex.9⁵ 14^9 25^9 25^{22} 29^3 32^{10} Lv. 10^2 14^6 15^{10} 15^{29} 17^5 22^{16} 23^{43} 24^6 25^{55}

 $Nu.4^{12}$ 4^{19} 4^{23} 4^{49} 5^4 5^{21} 6^{20} 7^3 7^5 7^6

(a) AN (acc)(contd).

S T always has Scriptio defective bt.1 15 3 6 3 28 6 23 9 14 9 28 10 15 12 29 18 14 26 16 27 4 31 7 31 10 32 51

(b) 1'2 S F always has suffixes indicating the singular $6n9^{12}$ 9^{15} 16^5 17^{10} 17^{11} 26^{28} 31^{53} $Ex.31^{13}$ $pt.5^5$

G. Proper Names

(a) Scriptio plena or defectiva in same syllables
Yodh

Scriptio plena

Gn. 10¹³ אוזישון 22²² ביט זרה 25² ביט 10¹ אוזים 22²² ביט ורישון 36²¹ ברפידים 17⁸ אוזים 16¹ אוזים 16¹ ברפידים 17⁸ בייוו 17⁸ ברפידים 17⁸ בייוו 17⁸ ברפידים 17⁸ בייוו 17⁸

של 10 ביריתו 14 ביריתו 2¹ בריתו 2¹ ברית

Nu. 2²⁷3ny 297⁷², 773ny 2934²¹ 773n 34²³3n 17

- C. Proper Names (contd)
- (a) Scriptio plena or defectiva in same syllables (contd.)

 <u>Waw</u>

Scriptio plena

 $Gn.10^{8}.9_{1171}10^{11}_{\Lambda 1217}10^{1}_{1717}14^{5}_{1717}14^{7}_{1718}14^{13}_{1717}14^{24}_{31200}$ $15^{19}_{1918}19^{25}_{1717}125^{4}_{1711}125^{13}_{1712}136^{34}_{1712}136^{34}_{1711}136^{35$

בצין ²¹ אורבה צפורה ²⁵ אורבה צפורה ²⁵ פיאן ³⁷ ביאן ³⁷ ביאן ³⁷ ביאן ³⁷ ביאן ³⁷ ביאן ³⁷ ביאן ³⁷ ביאן

Lv. 24 17, n130

Nu. $1^{11}_{1171} 2^{22}_{1173} 3^{21}_{127} 2^{3}_{171} = 3^{27}_{171} 2^{27}_{171} 3^{21}_{1$

שימון 4 סיחון 4 חחורב 1 בחורב 21 בחורב 21 בחורב 22 בחורב 22 בחורב 22 בחורב 22 בחורב 22 בחורב 22 בחורב 23 בחורב 23 בחורב 25 בחורב 25 בחורב 25 בחורב 25 בחורב 25 בחורב 26 בחורב 25 בחורב 26 בחורב 30 בחורב 26 בחורב 26 בחורב 26 בחורב 26 בחורב 26 בחורב 26 בחורב 30 בחורב 26 בחורב

C. Proper Names (contd)

(a) (contd)

Soriptio defectiva

(b) Scriptio plena of Yodh and Waw in same syllables

Yodh in S T, Waw in M T

Waw in S T, Yodh in M T

Ex.624 7104

(c) Scriptio plene of Yoth and Waw in proximate syllables

Nu.22³⁹ As' 21²⁰ | 102'7 23²⁸ | 102'7 (d) Scriptio plena of Waw in prox. Syllables Gn.30, 3 (iii) Remaining Variants, arranged under grammatical classes

of norms, adjectives, verbs and particles.

a. Scriptio plena or defectiva in same syllables

Yodh

Scriptic plena

· Nouris

$$14_{0}^{7}$$
 ביעים 3_{0}^{15} בביעים 25_{0}^{15} ביבעים 25_{0}^{15} בביעים 25_{0}^{1

(iii) Remaining Variants (contd)

a. Nouns (contd)

<u>Adjectives</u>

$$6n.6_{0.0}^{16}$$
, $3\omega.8_{0.0}^{13}$, $3\omega.8_{0.0}^$

Scriptio defectiva

Nouns

Scriptio defectiva

Nouns (contd)

Nu. 13330, 3217

Dt.14 18 19277 226 0182 226 01827 2833 742. 3231.330

Adjectives



Gn. 37 1017 152 152 4724 100

Ex. 264 7115772610 711577

בעי 20²⁰ אררים 20²¹ אררים 27¹⁵ אשי 10⁸

Nu. 285 A > 1

Dt. 3024 01770

Waw

Scriptio plena

Nouns

 8^{22} 717 9^{12} 210 310^{5} 10^{8} 10^{8} 9 712 10^{15} 10^{21} 10^{15} 10^{16} 10^{16} באצוני 14 רכושו 14 רכוש 14 רכוש 13 באצוני 13 בפונה 13 בראישונל אחואי 14¹⁶ מקומו 18²³ ברבא לוני 15¹⁴ ברבום 15¹⁴ לודרבום 14¹⁶ ברבום 14¹⁶ ברבו אמותי 20¹² בחצום 20 בנקיון 20 אחותי 20² יאותא 20³ פישור 19²⁸ פישור 20¹² משבועתי 24 גבולן 23 בכורן 22 בקולי 18 בקולי 22 קולה 21 בקולה 21 בקולה 21 בקולה 21 בקולה 21 בקולה 21 בקולה 21 בי 24 30 NIAN (bis) 24 35 NAM 124 59 NIAN 24 11 NIAN 25 ATSIN 25 13 NIZZ 2530 (bis) 2531 בכורה 2532 בכורה 2532 בכורה 2531 אחום 25 מום 25 מורה 2531 בכורה 2530 בכ 265, Sipa 267, NIUN 269, NIUN 278, Siba 2710, 1272 2713, 1272 2731

Way

Scriptio plena

Nouns (contd)

Gn. 27^{16} ATIV 27^{19} TIDE 27^{22} SIGN 27^{32} TIDE 27^{36} ANIM 27^{10} SIGN 27^{10} SIGN

Ex. 24 AIRW 27 LAIRW 315 Si 3 315 2173 318 1319 49 3173 (bis) 48 AIR TO THE RESTRICT OF THE RESTRICT OF THE RESTRICT OF THE RESTREE OF THE RE

(lii) (contd)

Scriptio plena

Nouns (contd)

 $\mathbb{E}_{\mathbf{x}} \cdot 25^{31}$ And 25^{32} with 25^{33} and 25^{34} and 25^{35} and 26^{3} and 26^{3} and 26^{3} עמודיהם 27 צמודיהם 27 צעמודיה 27 צעמודין 27 צותת 27 בקומתו 27 העדות 30 (26) 27 הל ידות 27 המידות 27 מחד בחום 27 מחד בחום 27 אות בפור 28 למוד בחום 28 למוד בחום 27 מחד בחום באפוד 28¹² צבות 28¹² האפור 28¹⁴ אבות 28¹⁵ ברון 28¹² ברון 28¹ 2829 אורו 29³³ במוני 28³³ במוני 28³³ במוני 28³³ במוני 29³³ במוני 29³³ במוני 29³³ במוני 29³³ במוני 29³³ במוכצמוני 29³³ במוני 29 29²² אראויב9³¹ הכיור 30²⁶ העדות 29⁴³ בכבודי 29⁴⁰ בועשרון 29 במקום 30²⁸ הכיור 30²⁶ העדות 30²⁶ הכיור 30²⁶ הביור 30²⁶ הידות 118 (bis) אות 31 לעולת 17 במתון 17 בארון 17 העדות 18 הארון 17 העדות 18 הארון 17 העדות 18 3322 - 712 > 34 nn 1 3 34 nn 1 3 7 34 Nn 1 3 7 34 24 7 3121 34 28 nn 1 3 7 34 29 nn 1 3 הכיור 35 מנורת 35 הארון 35 צמודיו 35¹ לנונורת 35⁷ מנורת 35¹ העדות 35¹ לנוורת 35¹ העדות 34² הארון 35 ברובים 36 (bis) אירת 36 ברובים 36 ועורת 35 לנמודין 35 3812 העדות 381 ניתות 381 ניתוד 381 ניתוד 381 ניתוד 38 ניתוד 38 ניתוד 38 ניתוד ב האפוד 39 האפוד 39 גבלות 39¹⁵ עבות 39¹⁵ באות 39¹⁵ האפוד 39¹⁶ האפוד ורמון 95 בנמון 95 הרמונים 95 הפנמונים 95 בנמונים 95 בנמונים 96 האפוד 95 האפוד 95 האפוד 95 האפוד ארון 40 עמודיה 95 הכיור 39 המנורה 39 העדות 39 עורת 39 אוריו 39 ארון 40 אמודיה 95 הכיור 39 המנורה 39 העדות 39 אוריה 404 ארדות 40 הארדון 40 הארדון 40²⁰ הרדון 40²⁰ הרדות 40²⁰ הארדות 4 40²⁴ 30 467 40³⁰ 2027

Nouns (contd)

Nu.1²⁰

Nu.1²⁰

Ru.1²⁰

Ru.1²⁰

Rul1²⁰

Rul1²⁰

Republic of the properties of the properti

Nouns (contd)

- Dt. $1^{4}50^{3}72 = 2^{3}31035^{3}70$ $10^{2}73^{3}12215^{3}7312315^{3}7311231123171112^{3}7312315^{3}731231$
- 6^2 סובים 27^1 3 עקוב 27^1 3 עקוב 27^1 3 נקוך 27^1 3 נקוך 27^1 3 עקוב 27^1 3 עקוב 27^1 3 נקוך 27^1 3 הגדו 27^1 4 עקוב 27^1 3 הטובות 27^1 4 הטובות 27
- בגדונ 12⁴ הראיטון 12⁴ הראיכון 26²⁸ הגדונ 18²² בגדונ 15¹⁶ הראיטון 12⁴ הגדונ 26²⁸ הגדונ 15⁴ הראיטון 34⁴ הראיטון 35⁶³³ הראיטון 34⁴ הראיטון 35⁶³³ הראיטון
- ברוש 21 ברוש 619 קדוש 620 קדו
- אנים 27 לברוב 10 לבר
- הטובה 8 הטובה 6 טובול 6 הגדול 47 הראישונים 4 הגדול 27 בנדול 17 בנדול 17 בנדול 17 בנדול 17 בנדול 17 בנדול 18 מובים 19 הגדול 18 מובים 18 הגדול 17 בנדול 18 מובים 18 בנדול 17 בנדול 18 מובים 18 בנדול 17 בנדול 18 מובים 18 בראישונים 19 בראישונים 19 בראישונים 19 בראישונים 19 בראישונים 18 בדול 18 בנדול 18 הקרובים 18 הקרובים 18 הקרובים 18 הקרובים 18 בדול 18 הקרובים 18 בדול 18 בנדול 18 הקרובים 18 בדול 18 בדול 18 בדול 18 בדולים 18 בדול 18 בדולים 18 בדול 18 בדולים 18 בדול 18 בדול 18 בדול 18 בדולים 18 בדול 18 בדולים 18 בדול 18 בדול 18 בדול 18 בדולים 18 בדול 18 בדולים 18 בד

(iii) (contd)

Verbs

בקודים 30 עקודים 30 עקודי

ברוע 32 כמובים 32 כמובים 32 ברוע

ברום 2118 בשמוטה 14 כרומים 18

Nu.1126 אופה 222 בכתובים 28 היושאה

Farticles

Gn.139 x13 1920 x13 205 x13 2439 ·Sin 2736 x13 2925 x13 315 w 3415 con 3725 w13 4415 con 2

Ex. 319 mi 3 n 410 mi 3 cm 411 mi 3 n 414 mi 3 n 57 mi 2 2 5 0 mi

Nu.122 4137 225. SINH 2237 4137 2312 4137 2326 4187 2412 4137

Nouns

 $6n.4^{21}$ 112 4^{21} 212 9^{2} 21 19^{1} 212 24^{25} 200 12^{32} 20 130^{8} $3 \cdot 3 \cdot 10^{1}$ 30 10^{27} 215 10^{27} 216 10^{27} 216 10^{27} 216 10^{27} 216 10^{27} 216 10^{27} 216 10^{27} 216 10^{27} 217 10^{27} 218 10^{27} 218 10^{27} 219 10

באר 2³קס ב⁵קס 13¹⁶ בארים 28²⁰ באל 25¹⁶ בארים 28²⁰ באל 25¹⁶ בארים 29²⁰ באל 25¹⁶ באל 29²⁰ באל 29²⁵ באל 38¹² ברה 38¹² ניחת 29²⁵ ביחת 39²⁶ ביחת 39

(iii) (contd)

Scriptio defectiva

Nouns (contd)

 14^{19} 14^{14} 14^{17} 14^{17} 14^{17} 14^{17} 14^{17} 14^{17} 14^{17} 14^{18} $14^{$

Nu. $10_{\text{Al7}99}^{2}$ fl 3^{20} Al 22^{23} and 22^{23} ywh 22^{23} km w 123^{22} ya 224^{8} A 24^{8} A

Dt. 2²⁷ אושמאל 11²⁵ אולם 2³ צעלם 2³ צעלם 2³ אושמאל 11²⁵ אומראכם 11 אהמה 12³ אושמאל 2³ אושמאל 2³

Adjectives

הטהרק 7⁸הטהרק 7 ערמים 2⁸ה.2

Ex.1 22 73.7 11 36 372

Nu. 22 18, 372 22 32 636

 $\mathsf{Dt.4}^{36}{}_{372}^{372}{}_{7}^{6}{}_{472}^{14}{}^{21}{}_{472}^{14}{}^{21}{}_{572}^{16}{}^{16}{}_{5}{}^{2}{}_{5}{}^{25}{}^{13}{}_{572}^{25}{}^{14}{}_{772}^{28}{}^{9}{}_{17}^{28}{}_{917}^{28}{}_{917}^{28}{}_{19}^{28}{}_$

Verbs

cn.48¹⁰ 35

Ex. 105 30 3825 .770

Lv.214 3P

Nu.1⁴⁵τρο2³²τρο2³²τρο 3³⁹τρο 4³⁷τρο 4⁴¹τρο 4⁴²τρο 4⁴⁵τρο 26⁵¹τρο 26⁵¹τρο 26⁶⁴τρο 26⁶⁴τρο 26⁶⁴τρο 26⁶⁴τρο 26⁵¹τρο 26

Particle

Lv.51 N3

(iii) (contd)

b. Scriptio plena in proximate syllables

- (b) Yodh in S T, Waw in M T

Dt.2918 אבשרירא

- (c) <u>Waw</u> in S T, <u>Yodh</u> in M T

 Gn.3¹⁰ ארום 3¹¹ ארום 25²³ ארום 49¹⁷ ארום 25¹⁰ ביות 36¹⁰ ארום 149 Lv.6¹⁴

 Dt.5¹⁶ ארום 28⁴⁸ ארום 16¹⁶ ארום 16¹⁶
 - (2) Possible orthographic retention of consonantal Yooh
 - (i) Lamadh He verbs and nouns
 - a. Final Yodh in S T: final He in H T

 Gn4² '10 8¹¹ '3 14⁹ '14 9 14²² '17 21²⁰ 21 29³⁴ 3 · 29³⁵ '11 / 30³ '12 14⁹ '17 47⁴ '17 49²⁴ '17 49²⁴ '17 19²¹ '18 12 14⁹ '17 19²¹ '17 19²¹ '17 19²¹ '17 19²¹ '17 19²¹ '17 19²¹ '17 24¹⁶ '17 27¹⁷ '17 19¹³ '19 14⁷ '3 20 23¹¹ '220
 - b. Final <u>He</u> in S T: final <u>Yodh</u> in M T

 Gn16⁶ πων 16⁹ και 18⁶ πων 49¹⁷ π3ν (bis)

 Nu.21¹⁷ π3ν 24 πτω

 Dt.32²⁴ πτη

C. Yodh before consonantal afformative

In M T

Gn.
$$26^{35}$$
 , π Λ 129^{8} , π 141^{21} π 141^{53} π 150^{5} . Λ 150^{5

(ii) Prepositions 3x and 3y with suffixes

3. Variants normalizing the spelling of words

1. Archaisms

3 s.m. Suffix: ST has normal form where MT reads π —

Gn.9²¹ 12⁸ 13³ 35²¹ (error in \vee Gall text) 49¹¹

Ex.22⁴ 22²⁶ 32²⁵

Lv.23¹³

11. Errors

ST 7.7: MT 7.7: 00.2^{12} 3^{12} 3^{20} 7^2 10^{11} 10^{12} 12^{14} 12^{18} 12^{19} 14^7 14^8 17^{14} 18^{10} 19^{20} (bis) 19^{38} 20^2 20^3 20^5 20^{12} 21^{22} 22^{20} 22^{24} 23^2 23^{19} 24^{44} 25^{21} 26^7 (bis) 26^9 26^{12} (bis) 27^{38} 29^2 29^9 29^{25} 32^{19} 32^{20} 34^{14} 35^{19}

11. Errors (cont).

Gn. 35 27 37 38 1 14 16 21 25 10 32 6 17 18 7

Lv. 2^6 2^{15} 5^{11} 5^{12} 6^2 6^{10} 6^8 6^{22} 7^{20} 7^{21} 7^{27} 10^{12} 10^{13} 10^{17} 11^6 (bis) 11^{26} 13^4 13^6 13^8 13^{11} 13^{15} 13^{20} 13^{23} 13^{25} (bis) 13^{26} 13^{27} 13^{28} 13^{22} 13^{52} 13^{55} 13^{57} 14^{44} 15^5 15^{23} 15^{25} 18^7 18^8 18^{11} 18^{12} 18^{13} 18^{14} 18^{15} 18^{16} 18^{17} 18^{22} 19^8 19^{20} 20^6 20^{14} 20^{18} 20^{21} 22^3 22^{12} 23^3 23^{30} 23^{36} 25^{35} 27^4

Nu. 5^6 5^{13} 5^{14} 5^{18} 5^{28} 5^{31} 8^4 9^{13} 13^{18} 13^{19} 13^{8} 13^{7} 13^{8} 14^{8} 14^{415} 15^{25} 15^{30} 15^{31} 19^9 19^{13} 19^{20} 21^{16} 21^{26} 22^{4} 32^{4} 32^{26}

Dt. 1^9 1^{16} 1^{18} z^{20} z^{34} 3^4 3^8 3^{11} 3^{12} 3^{18} 3^{21} 3^{23} 4^6 4^{14} 5^5 9^{20} 10^1 10^8 11^{10} 13^{16} 14^{28} 17^5 20^{20} 21^3 22^{18} 22^{24} 24^4 29^{21} 29^{26} 30^{11} 30^{12} 30^{13} 21^4 21^6

111. Inconsistencies in identical or similar context
ST has usage elsewhere in both ST and MT

Gn. 32³¹3/112

Ex. 4 171'

Nu. 13⁸ אשוחי 13¹⁶ איהושץ

Dt. 32. איהושע

4. Variants in writing of Samaritan MSS

Nothing to add to treatment already given in Chapter V.

11. Differences of Grammar

- 1: Differences of Grammar with agreement in sense
- i. Accidence
- (1) Variants consisting of forms found elsewhere in Samaritan
 Pentateuch

(i) Normal forms

a... Common for rare

(b) Common form of suffix

(c) Residuum - unclassified further

Gn.
$$1^{24}$$
 AIR 20^6 AIR 10^{11} 13^8 15^8 15^8 10^{11} 10^{1

b. Forms suggesting misreading, miswriting or mishearing in S T or M T.

Ex.37 INT

Mu. 143 7210 2127 11311

c. Forms in identical or similar context

The parallel is shown in brackets.

Gn.620 (later in verse) qıya p

Ex.28²¹(earlier in verse) 25°15¹¹(15⁶) 15¹²(30³⁷ -

Perallel in M T) 1/10/21 228(22²⁶) 73/2 22²⁵(22⁶) 1/3/2 (32¹⁹(12¹⁷) 1/10/2

IV. 2537 (2536) A'27/21

Dt. 37(235) IJTT2 2413(Ex2226) 113002 294(Ex2226) 00-11300

d. Forms which do not belong to any of the above types

(a) Impf. with ending in Nun

In 5 T

Gn. 20 | 10 y. 20 | 10 y. 41 | 15 | 10 y. 42 | 10 y. 42 | 10 y. 20 |

 $Nu.27^{20}$ | 1900' 32^{22} | 12000 52^{24} | 1000 Dt. 4^{1} | 1100 100

a.

(a) Impf. with ending in Nun (contd)

In M.T. (contd)

 11^{27} | yaw | 4^6 | yaw | 6^{14} | 6^{13} | 3 | 197' | 8^{16} | yaw | 12^{12} | yaw | 12^{12} | 120 | 12^{12} | 121' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 17' | 13^{12} | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' | 18' |

אחרי end אחר (b)

ST has 700 ST has 1700 ST has

(c) He paragogic

In S T

Gn.4⁹ 7'N 19³⁴ 717 27¹¹ 717

Nu.31¹⁶ 717

Dt.3237 7'~

In MT

Gn. 6 σπ 7 14 οπ 24 65 13π 27 39 μπ 37 19 13π
42 35 απ 44 3 απ

Ex.1 9 | 7 19 13 ол 32 16 ол 36 4 ол

In.11 28 07

Nu.1 50 or 8 or 16 or 9 or 7 or 11 26 or 13 or 13 or 27 or 27 or 20 or 13 or 28 or 11 probability 29 or 12

Dt. 1 ³⁹ on 7 ¹⁷ 7' 11 ³⁰ on 12 ³⁰ 7' 14 ⁷ on 18 ²¹ 7' 32 ²⁰ on 38 ³⁰ 7'

(d) Two Forms of Suffix

- Gn. 18^{16} עמהם 21^{29} ברכה 27^{27} 22 ברכה 29^{9} עמהם 30^{38} אכ 37^{20} , 37^{20} , 37^{20} אכ 37^{20} , 37^{21} ניט 41^{21} כקן 41^{19} וי 37^{21} ניט קרב הן 41^{21} כקן 41^{19} וי 37^{21}
- Ex. 2^{8} 171957 2^{10} 1. A. ω h 4^{15} 1.9 18^{20} 0 AM 19^{22} 0 2 20^{8} 1707 3^{20} 25 20^{29} , 33 0 2 $(26)30^{4}$ 0 72 32^{10} 0 2 34^{13} $\begin{cases} 2^{10} \times 2^{10} \times 2^{10} \\ 2^{10} \times 2^{10} \end{cases}$ 36 $3^{10} \times 2^{10}$ 37 $3^{10} \times 2^{10}$ 38 $3^{10} \times 2^{10}$ 39 $3^{10} \times 2^{10}$ 30 $3^{10} \times 2^{10}$ 30 $3^{10} \times 2^{10}$ 31 $3^{10} \times 2^{10}$ 32 $3^{10} \times 2^{10}$ 31 $3^{10} \times 2^{10}$ 32 $3^{10} \times 2^{10}$ 3
- Lv. 1116 11:n3 1122 111:03 1129 11:n3 1527 012
- Nu. 412 בכיי 27 בליין 21 בה 21 בכיי 21 במו 22 צמם 21 בה 21
- פים 12 מזבחתיהם 12 ומזבתיהם 75 קדשהו 5 ונכנו 3 תחתיהם 12 מחתיהם 21 במיהם 21 צמהם 24 עמהם 29 שהו 22 שהו 21 ב

(e) Residuum - unclassified further

- Ex. 4^{14} 12232 12^{5} 1202071 4^{5} 23 16^{2} 131 16^{7} 132 21^{22} 773 1 28^{6} 7366 22^{25} 3666 26^{4} 7021162 29^{40} 7122 34^{6} 7122
- אבני ³⁰ 11 ושסעה 11 בקריבי 10 מבני 14 שסעה
- Nu. 3 ארם 3 בשבי 12 גו 16 וו 13 וו 14 8 בשב 25 בשבי 13 ארם 3 בשב 25 און 16 ארם 3 בשב 25 און 16 ארם 3 בשבי 14 און 16 ארם 14 ארם
- Dt. 4^{11} 223 9^{28} ' λ 32h 11^{17} 77h 12^{5} 0' ω 3 14^{5} 'h Λ 14^{28} 9^{h} 17^{15} ω Λ 0. ω 19^{15} h ω 1 24^{15} $1<math>\lambda$ 3h ω 2 '29 4 00' 1λ 3h ω 0 30 5 7 λ 2 ω 1 32 27 13 33 27 'h Λ 1 33 26 $1<math>\lambda$ 4221 33 29 7 λ 42

(ii) Abnormal Forms

a. Harly

- Gn. 12^{11} 12^{15} 12^{15} 12^{16}
- Nu. 11³² 1367 22²⁹ 433417

b. Late

$$G_{10}, g_{10}^{16}$$
 G_{10}, g_{10}^{16} G_{10}, g_{10}^{16}

Dt.21¹⁵(late form in M T) 781163 327 A81.

in S T or M T.

71ועה 28²⁵ מזועה

- Gn. 3^{16} (3^{17}) Nest 20 20^{12} (18^{13}) as 6^{18}
- e. Forms which do not belong to any of the above types. Hithpa'lel of $\sqrt{10\omega}$.

$$Gn.18^2$$
ין אסאוי 19^1 יו אסאוי 22^5 ויטאחוי 22^5 ויטאחוי 23^{12} יו אסאוי 24^{26} יו אסאוי 24^{26} יו אסאוי 24^{26} יו אסאוי 24^{26} יו אסאוי 24^{31} יו אסאוי 24^{31} יו אסאוי

- (2) Variants consisting of forms not found elsewhere in Samaritan Pentateuch.
 - (i) In M T of Hebrew Bible.

OUTside Pentateuch.

$$Gn.3^{16}$$
 $711771.$ $(Ru.4^{13} Ho.9^{11})$ $Ex.19^9$ $(Jb.15^{26} 1K.7^{26} 2Ch.4^5 Je.52^{21})$

(2) (contd.)

(iv) In no other known source.

Gn. 49¹⁰ IAAP'

ii. Syntax.

(i) Variants consisting of constructions found elsewhere

in Samaritan Pentateuch.

- (i) Normal constructions
 - a. Common for rare

Gn.
$$8^{22}$$
 oh 1' 38^{24} o : 1: 341^{32} or 33^{14} or 49^{20} problem Ex. 24^{10} rights

Lv. 19^{20} or or Nu. 10^{36} an is harmal of the state of the sta

- c. Constructions in identical or similar context.
 - (a) Differences of word order.

Ex.
$$23^{23}$$
 (23^{28}) — 25^{29} (37^{16}) — 34^{11} (33^{27}) — 14^{52} (14^{4}) — 14^{52} (14^{4}) —

Nu.1²⁰(1²) —
$$28^{22}(28^{15,30})$$
 — $28^{27}(28^{11})$ — $29^{22,28,31,34,38}$, (29^{16}) —

$$Dt.11^{1}(30^{16}) - 20^{17}(23^{23}) - 30^{16}(11^{1}) -$$

COT

Appendix A.

(b) Differences in verbal form

$$Gn.19^{20}(19^{17,17,22})$$
 אריז אונה $30^{39}(30^{38})$ האנה $46^{22}(46^{15})$ אריז אונה $46^{22}(46^{15})$ אריז אונה $37^{39}(25^{10})$ אריז אונספה $37^{39}(25^{10})$ אונספה $36^{30}(36^{4})$ אונספה $36^{30}(36^{4})$ אונספה $36^{30}(36^{4})$ אונספה $36^{30}(36^{4})$ אונספה $36^{30}(36^{4})$

(c) Absence of suffix in S T or M T

Ex.
$$23^{20}$$
 (23^{23}) '> n^{3n}

Lv. 13^{5} (30^{6} , 8 , 10 , 20) n^{21} (30^{6} etc.) n^{21} (30^{6}) n^{21} (30^{6})

(d) Residuum - unclassified further.

Gn.
$$8^{12}(8^{10})$$
 $3n \cdot 139^{15}(39^{18})$ · $n \cdot 17743^{18}(43^{12})$ 201771 Lv. $2^{7}(2^{5})$ 711717 Nu. $28^{3}(28^{6},10,15)$ A3y $32^{7}(32^{9})$ [IN · A Dt. $2^{25}(2^{24},31)$ $3n \cdot 31^{7}(31^{23})$ A · $2A$

- d. Variants which do not belong to any of the above types.
 - (a) Differences of word order

$$Gn.12^{16} - 19^{28} - 20^{12} - 26^{29} - 31^{16} - 31^{17} - 34^{12} - 35^{10} - 41^{56} - 50^{12} - 26^{8} - 29^{18} - 29^{21} - 30^{19} - 34^{4} - 35^{29} - 36^{1} - 29^{21} - 30^{19} - 34^{4} - 35^{29} - 36^{1} - 36^{1} - 36^{12}$$

(a) (contd)

$$10^{10} - 11^{16} - 11^{26} - 12^{8} - 13^{27} - 13^{30} - 14^{10} - 20^{13} - 20^{19} - 20^{22} - 21^{2} - 21^{21} - 13^{21} - 10^{10$$

$$Nu.8^{16} - 26^{10} - 26^{58} - 28^{24} - 32^{6} - 32^{25} - 32^{29} - 32^{31} - 32^{35} - 36^{11} -$$

$$19^{20} - 3^{19} - 5^{18a} - 12^{22} - 12^{28} - 14^{15} - 19^{20} - 20^3 - 30^9 - 34^{10} - 12^{28} - 14^{15} - 14^{15} - 12^{19} - 1$$

(b) Sequence of tenses

 31^{7} עשים 31^{42} עשים $31^{$

 17^{5} אקא 18^{26} וישפטן 19^{12} אאאר 22^{6} ב3שנו 22^{11} ב3שנו 26^{24} יוידו 36^{2} עשן 35^{35} ועשה 35^{10} יוירו 36^{2} אוריו 36^{2} וידו 40^{31} אוריו

וערץ 15²⁴ יערך 15³² יערך 15³² יערך 15³² יערך 16³² יערן 18⁵ יערן 18⁵ יערן 18⁵ יבפר

יטמא 119 והין 17³ יקריב 16⁵ הכיאי 1¹⁷ ושימו 19¹⁷ הכיאי 18¹⁷ הכיאי 19¹¹ הכיאי

 19^{12} אונ 20^{26} אונ 20^{26} אונ 21^{20} אונ 21^{26} אונ 21^{26} אונ 21^{26} אונ 21^{26} אונ 21^{26} אונ 21^{24} אונ

 28^{20} ויירשו 36^{10} ברכך 16^{10} ויירשו 30^{10} אמר 31^{23} ויירשו 33^{23} ויירשו 33^{23} ויירשו 33^{23} ויירשו 33^{23}

c. Infinitive absolute:

Qualifying finite verb.

Substitute for finite verb.

הצכו וחסרו 8 הצכוושבו 6 .8 Gn. 8

133 175t

הקריבו ליהוד

 10^{25} והפשט 20^{26} טשפהו 25^{17} וחפשט 14^{21} ווחפשט 14^{21}

d. Imperfect ending in 7-

In S T

ונזבחה²³ בלכה ²³ ואדכרה 6⁵ ואמרה פ²³ וארדה 11 פיינו בחה 126 ארדברה 126 ארדברה 126 איינו ברה 15 פיינו

Nu. 24 13 727 ~

 2^{6} ונעברה 2^{19} ונעברה 2^{19} ואדברה 2^{10} ונלכה 2^{11} ונסצה 2^{10} ונעברה 2^{10} ונעברה 2^{10} ונעברה 2^{10} ונעברה 2^{10} ונעברה 2^{10} ועברה 2^{10} ועברה 2^{10} וועברה 2^{10} ואת פשה 2^{10} ואת פשה 2^{10} ואת פשה 2^{10} ואת בעברה 2^{10} ואת בעברה 2^{10} ואת בעבר 2^{10} וארדה 2^{10} ואפס בר 2^{10} וארדה 2^{10} ואפס בר 2^{10}

In M T

d. (contd)

In MT

Gn. 12^3 77241 18^5 18^6

Nu.11 13 John 14 4 | Nu.11 14 20 17 20 19 20 19 Dt.32 20 7'AON 32 26 AND WHEN THE PROPERTY OF T

e. Impv. ending in 3 -

In ST

Gn.14²¹ 711 19⁹ 762 Nu.21¹⁶ 790~

In M T

Gn. 21^{23} y 2ω π 125^{33} y 2ω π 27^3 π 181 27^7 π π 29^{19} 2ω 37^{13} 7^3 43^8 π 3ω 47^{31} y 2ω π Ex. 3^{10} 7^3 Nu. 10^{29} 7^3 23^{27} 7^3 24^{14} 7^3 Dt. 26^{15} 9 9ω π

(f) Differences of conjugation.

Two actives

.(f) <u>Differences of conjugation</u> (contd)

Two actives.

 $6n.6^4$ ויו 8^{17} השחית 6^{17} השחית 6^{17} השחית 6^{17} ויו 8^{17} היונקה 15^3 בהשחית 19^{29} השחית 19^{13} וירש 15^3 הוציד 10^{18} בהשחית 10^{18} הוציד 10^{18} הוציד 10^{18} במשחית 10^{18} הוציד 10^{18} במשחית 10^{18} הוציד 10^{18} במשחית 10^{18} במשחית

 13^{22} העדה 18^4 ימוש 18^{22} ויוצאהו 18^{21} ימוש 18^{21} ימוש 18^{21} ויביאו 18^{22} אבעירו 18^{22} ימוש 18^{21} ויביאו 18^{22} ויביאו 18^{22} ויביאו 18^{22} ויביאו 18^{22}

שביתן 25⁴⁶ אשביתן 26²² אשביתן 26⁴³ האבדתי 26²² אובן און אר 26²² אובן ארבידן 25⁴⁶ אובידן 26⁴³ האבידן

Nu.14²⁴ π 10 114^{41b} π 138 π 1132 π 10 111 22²⁶ π 10 11 22³⁰ π 10 12 23¹⁵ π 10 23⁵² π 10 23⁵² π 10 23⁵² π 10 21 24 π 10 21 25 π 10 21 24 π 10 21 25 π 10 21 24 π 10 25 π 10 24 π 10 25 π 10 25 π 10 25 π 10 25 π 10 26 π 10 27 π 10 28 π 10 29 π 1

 9^{3} אבירו 2^{6} החזק 2^{6} החזק 2^{6} ואבדת 2^{6} אבירו 2^{7} החזק 2^{2} החזק 2^{2} אבירו 2^{8} הדבקה 2^{2} וואבדתם 2^{8} אבירן 2^{2} אבירן 2^{2} אבירן 2^{2} אבירן 2^{2} אוריש 2^{2} אחריש 2^{2} אבירן 2^{2} אוריש 2^{2} אוריש 2^{2} אוריש 2^{2} אוריש 2^{2}

Two passives.

Gn.2510 7127 4015 W2121

Ex.226 21211 3032 701.

Lv.1016 7170

Nu. 226 7N'1 246 0'101 247 NOIAAI

Active for passive

 $Gn.4^{26}$ $3n\pi$ $9^2 1.11 10^1 17.3111 17^17 7.314 33¹¹, <math>n=34^{10}$ 1111 11 41 45^{19} '118 46^{22} 773'

LYV

Appendix A.

Active for passive (contd)

שב 10⁸ בשום 12³⁹ והבאת 27⁷ גרשום 12³⁹ וישב

Lv. 4^{31} 7:0: 4^{35} 7:0: 6^{15} 7:07 6^{23} 7:2: 7^{18} 324 11^7 111 132 7:13 1356 1027 16^{10} 7:4: 19^{17} 324

 $Nu.3^{16}$ 1718 7^{84} 7^{86}

Dt. 7²³סייה 14¹⁹ ולכיר 17⁴ אאכלו 12⁸ מייס 28²⁰סייה 28²⁰ מייס 28²⁴ השמידן 28⁴⁵ השמידן 28²⁵ השמידן 28²⁶ השמידן 29²² אוריע 29²² השמידן 28³¹ השמידן 29²² אוריע 29²² אוריע 29²² השמידן 28³¹ השמידן 28³¹ השמידן 29³¹ המידן 29³¹ המי

 $6n.24^{33}$ $0011 36^{14}$ 0000144^9 00145^{24} 11000 11000 11000

Ex.5¹⁹ y73'

בע. 12² אור בע 15³ בו אוריע 20²⁰ ואאוי

(g) Agreement of subject and predicate

Gn4¹⁰ 798 10^{18} 7891 11^7 994 13^6 784 13^7 994 15^{16} 15^{16} 15^{17} 16 17^{11}

(g) (contd)

ויאמנו ³¹ איינכן ²⁹ וירבו ²⁰ ואיתה ¹⁶ ויהיו 1¹¹ ויהיו שאו 10¹⁷ נשאק 10¹³ ויגרשו 11¹¹ נכו 1³ וימצאו ²⁵ 11² יהיו 13¹² והין 13⁹ ויקרו 12²⁷ דברו 13¹⁶ וחמה 16²¹ היצכו 16⁴ ויצן 15²⁴ ויאמרו 16²⁵ ו 10עים 1630 1124.1 1632 IN3 172 AIR 173 17AVI 176 HSVI 1816 142 2118 1521 2125 MSII 2129 \$ por 2132 {370 v ונשאן 25²⁸ ויענו ³ 24² יענה 24² יהיו 25²⁸ אזכר 23¹³ 25²⁹ וספי 25³⁷ איל און 26³¹ והעלית 29¹⁵ אורך דשו 29⁴³ וורך דשו 30⁴ אין 30¹⁴ אורין 30¹⁹ און 30¹⁹ און 33¹⁰ שוה 3525 הביא 25ניבוא ועשה 3510 יעשה 35²⁵ ויקמו 10 יראץ 40³¹ ויראץ 36³¹ ויצשו 39³,7,8,9,22 ויראץ 40³¹ ויראץ 40³¹ ויראץ 1 י סמכו 1⁴³ יקריבו 7²⁵ יקריבו 1¹ן יקריבו 7⁸ יאכל 8¹⁴ יקריבו 1⁸ן יאכל אוקרצו 13⁵⁶ בי תפן 11³⁵ ויראו 9²⁴ וישם 20 ויסמך 14⁴ אפשה 14⁵,13,41,42,43,45,49,49 אפשה 19¹⁵ אפשה 19²⁴ אפשה 19²⁷ אפשה 19²⁸ אפאר 19²⁸ אפר 19²⁸ 2514 אסרו 25²⁸ אשהו 25^{31 אש בו 25³¹ אשהו 25³⁴ איקריבו 25²⁸ אשהו 25³¹ איקריבו} 2711 2177' Nu.343 1.71 510 7.7. 93 107. 96 1.7.1 1215 173212

 $Nu.3^{43}$ ויהיו 5^{10} יהיה 5^{10} ויהיו 5^{10} ויהיו 12^{15} ויהיו 12^{15} ויהיו 13^{24} ויירשו 14^{1} פראו 13^{24} ויירשו 14^{1} פראו 13^{24} ויירשו 14^{1} פראו 13^{24} ויירשו 13^{24} היה 13^{25} ויורשו 13^{25} אבדה 13^{25} ויורשו 13^{25} אבדה 13^{25}

(g) (contd)

Nu. 27^3 היה 27^4 | IJA 27^{21} { N_5 : 28^{17} | 32 | 28^{24} | 28^{17} | 32 | 31^{13} | 31^{29} | 32^{18} | 32^{18} | 31 | 31^{29} | 32^{18} | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 | 31 |

(h) Transitions in personal reference

Ex. 11^7 yth 12^{46} [N is in 22^{20} [Jin 22^{20} [S in 3 22^{22} [Jyh 23^9] s in 3 23^{22} [N in 23^{22}] 25^9 in 25^9

(i) Feminine termination.

In S T

 48^{22} אחא 49^{15} מובה 9^2 בסאאתו 9^2 פחה 9^2 ברעה 9^3 מובה 9^3 אחא 9^3 מובה 9^3 אחא 9^3 מובה 9^3 אחא 9^3 מובה 9^3

(i) (contd)

 $Ex.15^{11}$ אקא 30^{21} באפאין 28^{12} באפאין 28^{10} רבעית 30^{21} אפא 30^{21} אפא 30^{21} אפא אחאא

 $L_{V}.4^{21,24}$ היא 4^{32} כבשה 4^{32} היא 2^{30} היא 15^{23} היא 18^{23} היא 19^{28} היא 19^{25} היא 19^{25} היא 19^{25} היא 19^{25} היא 19^{25} היא 19^{25} היא 19^{25}

 $Nu.5^{15}$ Nia 7^8 Nia 7^8 Nia 13^{20} Nia 23^{10} Nia 31^{28} Nia 31^{37} Nia

Dt.15⁷ ΛΠΝ2 16⁵ ΛΠΝ2 17² ΛΠΝ2 23¹⁷ ΛΠΝ2
23² ΛΩΝ 27¹⁰ 1'ΛΩΠ 28⁷ ΛΠΝ 28²⁵ ΛΠΝ 31¹ 73 33¹²
1'Λ² Λ²

 $6n.26^{29}$ ארבע 9^5 יון 32^9 האחר 37^8 ויף 38^{25} סיא ארבע 46^{22} ארבע 46^{25} ארבע

Ex. 25^{12} Thus 25^{36} Thus 32^{25} Issue 36^{31} Thus 37^3 Thus 37^3

Nu.816 200 156 3132

של 30¹⁶ אין 30¹⁶ הזה 38⁶¹ מוב 32³². איש פצי 32³² אין אין 30¹⁶ הזה

(j) Plural (or dual) termination.

In ST

Gn.1²¹ $_{21}$, $_{37}$, $_{37}$, $_{37}$, $_{37}$, $_{37}$, $_{37}$, $_{37}$, $_{37}$, $_{37}$, $_{37}$, $_{37}$, $_{39}$, $_$

Here, as elsewhere, the S T readings are given.

(j) Plural (or dual) termination (contd)

In S T (contd) במכסות 12⁴ הכנים 8¹³ כדבריך 8 משלבין 12⁴ הכנים 12⁴ כדבריך 12⁴ ובבתיך יו יצין 149 ידיך 13¹⁶ ידיר 13⁹ בידיכם 112 באורואי 16⁴ וחיציו 15⁴ באורך 15⁴ וחיציו 2311 איך 3 2311 פרמיך 2412 בירמיך האבנים 2412 בירמיך ביים ביי 25^{31} אבנטים 26^{31} קרשים $26^{18},19,20$ קניה 25^{31} קניה 26^{31} דרכיך 32¹⁸ אבנים 32¹⁹ מידיו 32³⁰ אבנים 31¹⁸ הבשמים 35 בריחין 1 35 וצחטאאינו 94 צרינינו 94 בנקירו 35 בנקירו 35 בריחין 1 35 בריחין 1 35 בנקירו אינו 9 א קניה 1777 ירביה 37⁸ קצאיו 37⁸ חכמי 37¹⁷ אחשבוא 39³³ 110122 39⁴¹ 1'AFP מדי 63 קרבניך 213 מנחאיך 2¹³ הכהנים 7 קרבניכם 17 מדי פפין 17 פינו 16 אדיכם 33 מצביהן 25 מצביהן 16 באבנטים 18 באבנטים 18 פפין 13^{55} איניו 16^{21} איניו 17^4 איניו 16^{23} איניו 16^{23} וי 3 די 20²⁷ בעיניו ²⁰ באבנים 20²¹ מכסות בין באר זות בהיך באר אוא חויכם בהיך באר אוא האין בדיך באר אוא האין בהיך באר אוא האין Nu.22 הלוים 32 הבכורי 346 מבכורי 3⁴⁶ הלוים 4¹⁵ דגלוו במועדין ³9 במועדיו 9² פרחיה 8⁴ ירכיה 9³ נסכיו פדבריך 14 חדשיכם (10a בחוצדין 14 בחוצדין 15 בחוצדין 97 בחוצדין מקדשי (3)18⁹ מינה 15³¹ מעוגיו 15³¹ מעוגיו 15³¹ ונסכיו עצאיך 18²⁶ בצאיו 36 צאיך צאין 23⁶ בארוחתיכם 23¹⁵ בנחצ בכם 25¹⁷ ויסכין, 26^{5%} במועדין 28² הצוים 26^{5%} עצאין 28⁷,8 יאסריה ⁵05 נדריה 30⁵ דבריו 30³ ונסכיה 30⁵ מדריו ממש פחות 36 8 משפחות 36 8 משפחות 36¹ אות שות 36⁸ חטאתיכם 32²³ נדריה (j) (contd)

Dt. 2^7 7'1' 2^{33} 1'12 5^5 '227 6^8 7'7' 6^9 7'22 7^5 2'11' 9^{18} 1'11' 9^{18

In Mar

 $Ex.3^7$ 120 18^{20} 71.7 $22^{1.2}$ 15^{25} 15^{17} 17^{7}

 19^{16} 1000 100

Nu.11⁶ צין 24⁴ אין 20²⁴ אין 20²⁴ אין 20²⁴ אין 20²⁴ אין אין 20²⁴ אין 20²⁴ אין 20²⁴ אין ציון אינוו

2713 70× 27²³⁰ 17, 31² 70× 34⁴,5,9,12 11N811

 33^{11} אירן 33^{29} נידו 33^{29} ודו 32^{29} ממכרו 33^{29} אירך 33^{29} ידו 33^{29} ידו 34^{9} ידו 38^{29} ידו 33^{29}

S T readings

(k) Suffixes

Difference in person or number

Gn. 6^{20} anith $3 13^{10}$ 135 21^{29} [1723 30^{376} and $31^{2,5}$ anith 31^9 anith 31^{43} anith 32^1 and 32^1 and 32^1 and 32^{14} and 34^{10} and 41^{23} [13^2 and 44^{24} anith 44^{24} and 44^{30} and 44^{32} and 45^1 anith 47^{26} and 47^{26}

 $LV.1^{16}$ $|\Lambda 812$ 1^{16} $|\Lambda M$ 6^{8} $|\Lambda 10^{16}$ $|\Lambda 812$ 10^{1} $|\Lambda^{2}$ $|\Lambda^{2}$

 100^{122} ופקדין 2^{13} ופקדין 2^{13} ופקדין 2^{13} אבאם 2^{14} פקדין 2^{19} ופקדין 2^{19} עבדאם 16^{10} ופקדין 2^{10} ופקדין 2^{10} ופקדין 2^{10} אינו 16^{10} עליהן 16^{10} אינו 16^{10}

(k) Suffixes (contd)

 5^{10} בה 5^{20} בה $5^{$

Number of 2nd. suffix

Transitions in personal reference

 34^{25} סיים 34^{34} איניכס 4^{34} איניכס 11^{23} אוה אגבם 11^{23} איניכס 11^{23} איניכס 11^{23} איניכס 11^{23} איניכס 12^{9} סיים 12^{9} סיים 13^{1} איניכס 13^{1}

Absence of suffix

- $6n.2^{23}$ $6n.2^{23}$ $6n.2^{24}$ $6n.2^{23}$ $6n.2^{23}$ $6n.2^{23}$ $6n.2^{23}$ $6n.2^{23}$ $6n.2^{23}$ $6n.2^{23}$ $6n.2^{24}$ $6n.2^{24}$ 6n.2
- Ex. 1^{22} 10^{3} 10^{2} 10^{3} 10^{2}

Absence of suffix (contd)

Nu.3 16 13 23 11 23 11 23 11 25 15 10 20 27 13 27 11 27 22 27 15 27 22 27 17 27 17 27 12 27 17 27 17 27 12 27 17 27 17 27 17 27 17 28 17 28 18 28 18 28 18 28 18 28 18 28 18 28 18 28 18 28 18 29 28

With pronoun object expressed by AN with suffix

In S T

Ev. 105 and 1001

Dt.928 0AN N'273 118 02AN 715 A

In M T

Gn.187 (1/14) 3

Ex.4014 DAW2371

Lv. 1334 17701 2027 918274

Dt. 1812 00710 2726 011013 2814 7150

(1) Form with 7 locale

In S T

9 פת 12 פערה 13 פדמה 13 היאינה 13 היאינה 13 השמאלה 25 פערה 13 אפראה 13 אפראה 43^{15} ארצה 45^{25} ביאה 43^{18} אפראה 48^{7} ארצה 45^{25} ביאה 48^{7} מצריחה 50^{3}

(L) (contd)

$$20^{24}$$
 שמק 20^{29} שמק 20^{29} שמק 20^{20} שמק 20^{20} שמק 20^{20} שמקה 27^{20} צפונה 26^{20} אימנה 26^{20} אמידה 28^{29} אמידה 28^{29} אמידה 28^{20} אימנה 36^{25} אמידה 36^{25} אימנה 38^{20} אימנה 38^{25} אימנה 38^{25}

המזבחה 9¹³ המזבחה 6.vu

 $Nu.13^{21}$ המחה 14^{25} המדברה 34^3 המדברה 34^6 המחלה 34^6 המחלה 34^8 המחלה 34^9 פכונה 35^5 ימה 35^5 ימה 35^5 המוצה 35^5 המוצה 35^5 במרה 35^5 במרה 35^5 והירדנה 37 שמה 317 המוצה 317 במונה 317 פערה 35^5 במרה 317 שמה 317 המוצה 317 מארימה 318

In M T -X

Gn. 14^{10} Dw 15^5 Dirac 19^6 Nas 20^1 47% 23^{13} Dw 24^{16} 147 24^{45} 177 25^{10} Dw 28^2 Arz 28^{12} Nat 29^1 47% 29^3 Dw 32^4 47% 37^{24} 7127 38^{28} 147% 43^{30} Dw 46^1 782 46^{28} 47% 49^{31} 8^{20} 150 Dw 8^{16} Dirac 9^8 Dirac 9^{10} Dirac 10^{19} Dr 15^{27} Dirac 16^{33} Dw 21^{33} Dw 26^{33} Dw 29^{42} Dw 29^{43} Dw 30^{18} Dw 30^{36} Dw 33^9 37% 38^{30} Dw 30^{30} Dw

Lv. 58 /16147

Nu.17¹⁹ סש 33⁹ סיל 33⁵⁴ סש 34⁵ איז 34⁵ סיח

שם 16 השחים 30¹² ביחשה 31¹⁶ בשחים 31

S T readings.

(m) Prepositions

Difference

Gn. 7^1 ns 3u 10^{20} $07 \cdot 13 \cdot 19^5 \cdot 13u$ $19^5 \cdot 13u$ 19^5

Ex. 2^{10} Λ_2 3^{10} 7^{10} 3^{10}

(m) (contd)

 13 אליה 25 אליה 26 ציר 20 אליה 26 אליה 25 אליה 26 אליה 25 אליה 25

Absence

In ST

$$Gn.22^{21}$$
 $73y$ 22^{13} $73y$ 39^{2} 59 75 75 15^{18}

$$10.1^{44}$$
 אים 15^3 מזרת 11^{14} רית 15^{30} אשר 18^{17} אשר 18^{14}

Prepositional phrase

Ex. 37¹³ -- 75"

$$Ex.2^{14}$$
 600 31 55 040 12^{43} 774 341 14^{13} 7047 15^3 701 362 17^{10} 47 34^{17} 17^{10} 47 17^{10} 47 17^{10} 47 17^{10} 17 17^{10} 17 17^{10} 17 17^{10} 17 17^{10} 18 17^{10} 19 17^{10}

^{*} S T readings

(m) Prepositions

Absence (contd)

In M T (contd)

Prepositional phrase

יראה צו בנו או אמר צהם פפ יצרתי צו בנו עמו 19³⁰ ויאמר אציף 18³⁰ פח. 18³¹ אצה צק 48³¹ צקח עמו דאויאמר אציו דאהנצר אתנו 44³¹ אבה אכב²⁵ פס

Ex. 13^3 12 DANY: 29^{33} DZ N3D3 34^{32} 1.3N 1623 Nu. 24^{10} 73 'AND 24^{13} 1.3N 717: 31^{19} D3 131

(n) Definite Article

In S T

 $6n.1^{28}$ הייה 1^{30} הראש 1^{30} הראש 1^{30} החיה 1^{31} הפש 1^{31} ההיה 1^{31} החיה 1^{31} ההיא 1^{31} ההוא 1^{31} הרוא 1^{31} הרו

(n) (contd)

 $Ex.5^{18}$ המשך 10^{21} וה אטר 9^{33} הברך 9^{24} הלבנים 12^{46} הברך 14^{25} החוצה 14^{25} החוצה 15^{12} הערב 15^{12} הארף 15^{12} הערף 15^{12} הערב 18^{14} הערב 18^{14} הערב 15^{12} הערף 18^{14} הערב 19^{15} הער

בתוקדה 6 החלב 47 העצים 412 התובח 7 השל חים 6 . התוקדה 6 החלבת 7 התלא בה 27 התובח 9 התובח 11 הולך 11 הכבשים 11 הכלי 11 העתר 11 הנפש 11 הכלי 11 העתר 11 הנפש 11 הכלי 11 העתר 11 הנפש 11 העם 11 העם 11

(n) Definite Article (contd)

In M T 🛠

 $Gn.1^{29}$ fy 7^{15} 7w2 8^5 w7n 19^9 7nn 21^{28} 1Ns 22^8 7w 29^2 1Ns 41^{42} 271 42^{27} 7nn 45^6 247 50^{11} 47N

Ex. 5^7 $_{0.12}3$ 12^8 $_{13}$

בקר 62 מובר 1422 מים 65 בקר 1422 מובר 1515 מובר 1515 בקר 2327 מובר 1515 בפרים 2327 ואחד

Nu.9 13 eini 13^{22} piv 13^{28} piv 19^{13} ath 22^{11} s. ev 31^{28} ann 31^{28} ann 31^{36} [Ns 31^{38} right] 31^{47} fan 32^{33} fan 34^2 fan

Dt.14²² Nev. 21¹⁵ Anni 25¹¹ 711

(o) AN Accusative.

In S T

 $G_{11}, 2^9$ 3_{2} 3_{2} 2^{19} 3_{2} 4^{17} 0_{10} 8^9 17' 11^7 11^7 10^{10} 11^8 11^9 11^9 11^{22} 17' 15^{10} 12^{10} 12^{10} 13^{10

^{*} S T readings

(o) AM Accusative

In S T (contd)

	1	C) () (12 () (()	,						
Ex.1 ¹⁸	フユマオ	s_{10}	100	216	lux	421	3=	6 ¹ 7	6 H
			700						
			77'						
16 ²⁹	ሊወይሊ	194	אשר	23 ²⁸	2 35	•	25 ³¹	17 IJ	n a
	11361	28) AINW	28 ³⁹	7172	7	29 ²¹	{ 11.	72 72
29 ³⁵	ידם	30 ¹	۸ دا ۵۸ ^۱	3110	שני נ		32 ²	יא מי	i
	המחצות	39 ¹	7 '^"	39 ⁴⁰	0 { 7':	אדו המ	40 ²⁵	ALTI	7
Lv.14	17'		17' 4						
	764		7 4 2						
	11377		•						021N1 EE
Nu.13 ²⁵	הארץ	15 ⁴	קיבנו						28 ₀₀₁₂₀
			ハフピス	-		_	3,צם	2 24 ¹	.A 76N
	אנט'ו		, 2			-	_		s8 ASar
pt.2 ²⁵	שמעך	$m_{\rm S}$	זרוצו	1211	lhω				21 100
1231	35	1412	הנטר הפרס}? העזניה	1413	הדאה		-		הקאת[1]
1418	החסידה	1418	ידוביפת העט צף	π 14 ²³	Inw				5 12m
16 ⁶	ιρω		100		1716				lhw
2812	760								

In M T

Gn.17⁵ 7nw 21^{28} yzw 33^5 1'J'y 41^{51} Dw 44^{11} AAAAA 46^{30} 7'Jo

(o) AN Accusative (contd)

In M T (contd)

$$\mathbb{E}_{x,8}^{3}$$
 כפי 9^{29} כפי 9^{27} מרבת 9^{25} נאשריהם 34^{13} מרבת 36^{35} ואשריהם

(p) Residuum, unclassified further

(11) Abnormal Constructions

e. Constructions where no cogent reason can be adduced in favour of S T or M T

Absence of apocopation in Impf. or Impv. in S T

(ii)

e. (contd)

With Waw ("and")

$$Gn.12^7$$
 הריה (5 MSS) 17^1 הריה (4 MSS) 10^1 הריה (3 MSS) 19^{20} היה (18 MSS) 19^{20} היה (18 MSS) 24^{46} הריה 24^{51} הוראה (15 MSS) 26^{24} היראה (4 MSS) 31^{10} היראה (5 MSS) 31^{20} היראה (5 MSS) 31^{20} היראה (5 MSS)

Ex.63 האכה 915 הכאו

 \mathbb{N} יאע 2 ה 4 ב ואשאה 10 ויראה 20 ויראה 20 ויראה 23 ויקרא 23

 2^{10} ונפנה 3^{10} ונפנה 3^{10} ונפנה 3^{10} ונפנה 3^{10} ואפנה 3^{10} ואפנה 3^{10} ואעשה 3^{10} ואראה 3^{10} ווראה 3^{10}

- (2) Variants consisting of constructions not found elsewhere in Samaritan Pentateuch.
 - (1) In M T of Hebrew Bible

Outside Pentateuch

 \mathcal{N}_{A}^{T}

Gn.19³² (Jn.9¹⁰) ·> 3 49⁴ (verbal form, Jn.9⁴ Zp3⁴)

- 2. Differences of Grammar with divergence in sense
 - 1. Doctrinal
 - (1) Controverse regarding Gerizim

(1) Controversy regarding Gerizim (contd)

(2) Other Samritan beliefs and practices

(1) Unity of the deity

Gn. 20¹³ אארה 31⁵³ שפטי 35⁷ אלגו Ex, 22⁸ אירטיי

(ii) Avoidance of anthropomorphism

 $6n.16^{13}$ 707 16^{14} 707 24^{62} 707 25^{11} 707 $6x.34^{24}$ 707

(iii)Resurrection

Gn.3¹⁹ אפיך (iv) Legal enactment Lv.6¹³ Dt.16⁴ בין הערבים ii. Non-Doctrinal

(1) Variants where there appear to be cogent reasons in favour of S T

An asterisk denotes ST is supported by LXX.

M T readings are difficult in grammar and/or sense,
sometimes to the point of unintelligibility. They
are normally attributable to the typical causes of
scribal error

 $Nu.5^{27}$ 7^{28} 7^{28} 7^{28} 910^{16} 16^{1*} $16^{$

(2) Variants where there appear to be cogent reasons in favour of M T

(i) Euphemism in S T Lv.15^{18,24} 70'~

(ii) More common form in S T

Gn. 24²⁰ אוף איז 30³⁸ אוף בהשקות

(iii) Correction by S T

Ex. 36 7'A2N

(3) Variants where there appear to be no cogent reasons in favour of S T or M T

(1) Attributable to the typical causes of scribal error $Gn.2^{24}$ $Gn.2^{24}$ $Gn.2^{24}$ $Gn.2^{24}$ $Gn.2^{24}$ $Gn.2^{25}$ $Gn.2^{24}$ $Gn.2^{25}$ $Gn.2^{25$

Ex.187*(part) 1782.1

Lv.1920 Ani

(2) Not belonging to the above type

 $6n.2^7$ DTW 2^{25} DTW 3^8 DTW 4^{25} NTD 1^{9} 1^{10} 1^{10} 16^{4*} 1^{10} 16^{4*} 1^{10} 16^{10} 1

Ex. 237 p. 757

Nu.2127 'n= 2321 b'=~

Dt.437 * אריהם אחריהם 1114*,15* אחריהם אחריהם 179

111. Differences of Vocabulary

- 1. With agreement in sense
- i. Substitutions
- (1) Pure synonyms
- (1) Common words

b. Words suggesting misreading, miswriting, or mishearing in S T or M T

$$6n.27^{13}$$
 $9N/7N$ 33^{18} 013ω $/ 03\omega$ 47^6 $win / will$ 517 $Nn / 5$ 517 $Nn / 5$ 1036 $nn / 518$ 1036 $nn / 518$

```
c. Words in identical or similar context
Gn.78(619, 73, 79) 7271 727 / 10001 WW
Ex.5<sup>10</sup>(conventional formula) 1727'1 / 1704'1
   78(conv.form.) איר / וידגר (conv.form.) איר / וידגר / איר / וידגר
   18<sup>26</sup>(18<sup>22</sup>) איז אר / ויד אר (conv.form) איא אר / ויד אר (conv.form) אין ארן
   25<sup>20</sup>(36<sup>10</sup>) 700 30 700 / 1000 30 000
   263,5,6,17 (3610) ANN SNANN / AANN SN NON
   26 (later in verse) \Lambda 72 \pi 12 / 20 (conv.form.) 72 \pi 1 / 20
   51^{12}(conv.form.) 727'' / 708'' 36^{5}(conv.form.) 1727'' / 1708''
              TAN 3N TAN / 17AN 3N 6'N
Lv.5^2(5^1,3,4,5) '>/\gamma\warphi\sigma\bigg(later in verse) \warphi\sigma\bigg(\sigma\bigg)\right\righta\pi\sigma\bigg\\.
   יע בה / יעשה ( 179 178 הנגע /הנתק (1354) 1352 מגיע /תשיב (179 178) יע בה /יעשה ( 179 178)
Nu. 24<sup>10</sup> (22<sup>28,32,33</sup>) סיאחר / וידבר (conv.form.) איאחר / וידבר (conv.form.)
   276 (conv.form.) אמר / וידבר (conv.form.) ויאמר / וידבר (conv.form.)
   294(2821,29) און אירון אירון
   3518(3516,17) 201/ 10
Dt.5 18a (Ex. 20 17) 70 // ///// 16 18 (Ex. 36) 2/ / ////
   d. Words which do not belong to any of the above types
                                          2161 / 216 18
Gn. 1912 0.24307 /0.0147 2450
   <sub>24</sub>60
            צאחיו / 3'רצהו ב<sup>31</sup> שנאיו / איביו
   31<sup>33</sup> AINDWA / AAANA 36<sup>35</sup> 11707 / 170
   4222
           יאראי / דבראי 45<sup>25</sup>
                                          מצרים / ארץ מצרים
Ex. 1 פרצה א 2 מצר מפרים / פרצה
                                          הי3ך / הנצר
           17271 / NACY 10 1 NACY / 12211
    7<sup>86</sup>
          158 / NI CB B TENER / BDI
    1629
                                           אן / וישם / אירן / אירן / אירן
              אני / אנכי
                                 26<sup>35</sup>
    2226
                                   4022
    2715
          18cap / 18ena
```

d. (contd)

Variations in Divine Name

$$Gn.7^{1}$$
 אלהים איהוה 9 יהוה אלהים $^{14^{22}}$ אלהים $^{14^{22}}$ אלהים $^{14^{22}}$ אלהים $^{14^{22}}$ אלהים $^{14^{22}}$ אלהים 16 אלהים 16

Variations in Divine Name (contd)

$$Ex.3^4$$
 0.78 0.76

(ii). Uncommon Words.

b. Late

e. Words suggesting misreading, miswriting or mishearing in S T or M T.

d. Words in identical or similar context.

- (2) Synonyms giving greater precision or emphasis
 - (1) Common Words
 - b. Words suggesting misreading, miswriting or mishearing in S T or M T.

c. Words in identical or similar context.

$$Gn.39^8(39^6)$$
 $\pi n \times n$ $/ \pi n$
 $Ex.18^{12}(17^5)$ $'177^{1}$ $/ '32^{1}$
 $Nu.3^9(3^{12}, 8^{16})$ $7^{1} \wedge n$
 $35^{25}(35^{24})$ $\pi n = 1$

d. Words which do not belong to any of the above types.

- ii. Additions
- (1) Pure Synonyms
- (i) Common Words
 - a. Stylistic supplements in S T
 - (a) Explicit subjects or objects

$$36^{19}$$
 שוק אברהם 18^{19} וויברכהולייברך את אברהם 18^{19} וויברכהולייברך את אברהם 18^{19} וואסר אברהם 18^{19} אושע אברהם 18^{19} אושע אברהם 18^{19} אושע אלופיהם עשו 18^{19} אושע אלופיהם עשו 18^{19}

$$Ex.2^3$$
 INN 13 / '3 2^6 7779 NZ 1137 / '9 5^{13} 2

$$Lv.5^8$$
 $V.5^8$ $V.5$

(b)	<u>Words in appo</u>	sition		
	•			אדני / אדני אברהם
2427	אדני אברהם	אדני /	25 ⁵	132 705.3 / 13
26 ⁵	אברהם אביך	/ אברהם	284	אברהם / אברהם אביך
38 ¹³	1130 7113	/ ′3	39 ⁴	בצינין / בציני אדנין
473	אחי יוסף /	אחיו	487	רי / רחצ מחך
Ex. 75 S	צמי בני יטרו	/ / /:	.	
(c)	Additional Wa	w ("and	11	
Gn.1 ²⁶	ICTAIALI		₃ 16	341
6 ⁴	0.30171		6 ⁹	በ ለ
6^{15}	016611		619	ומכל הבשר
6 ²⁰	Jac S NOC	(MT. GB7 33	in)3	וגם
817	321		819	וכל החיה וכל הרמש
819	וכל הרמש		9 ⁵	ויאמיו
97	18761		910	ובבקמק
12 ¹⁹	7 n 3 t		141	ואכיוך
142	10145		174	ואני
1716	ו מצ כי		1719	187581
185	ואחר		2014	luz I
21 ²³	1000		24 ²¹	שוחריש (
24 ²⁵	וגם		24 ⁵⁵	ואחר
2733	0,21		30 ⁸	0.21
30 ³⁴	. 171		3113	ואשר וצגה
31 ³⁶	ומק		3137	וכי
32 ⁶	lnai		32 ²⁰	וגם וגם
33 ⁹	ากป		3427	121

(c) Additional Waw("and") (contd)						
Gn.34 ²⁸	ואת אאנם	35 ²⁴	ובני			
36 ¹⁰	7341	3611	1581			
36 ²⁰	1489	ჳ6 ^{2ჳ}	1561			
36 ²⁷	1485	36 ²⁸	ก3 <i>พ</i> เ			
37 ¹⁴	ז ראה .	^{37²⁷}	157621			
3910	יום ויום	4117	והנני			
4120	והבריאות	41 ²³	1 37.64			
4127	107'01	41 31	ואחרי			
41 ³⁴	ነ ሂው	(יצשה אד 65 (יצשה 41 41 אר)	7641			
426	והוא	4219	והביאו			
42 ³²	והאחר	4313	(216)			
4316	ואת בנימים	4416	ת והאצהים	っし		
4517	1431	46 ¹⁶	131601			
49 ³ 1	ושם קברו	(MT 756) 50 ²⁰	והאצהים	(אצהים דא)		
Ex.1 ²	ושמצון וצוי	13	131211 70001			
2 ²⁴	bus, vui	₃ 6	2021 1341			
3 ¹⁵	ואלהי יצחק	₃ 16	542,1			
45	708° 13811	5 ²²	18an 15			
6 ³	اله و العالم	614	1784			
6^{17}	'ग्रहा	7 ⁹	יחיו			
7 ¹⁹	ועצ יאריהם	8 ^{la}	ועצ היארים			
8 ⁹	ומן החצרות	8 ²⁰	$I \wedge \omega \cap \Lambda$			
8 ²⁵	ומעבדין	827	ן מ צבדין			

(c) (contd))
-------	--------	---

(c) (con	ia)			
Εx	. 9 ³	וב חמרים ובגמצים	9 ²⁵	ועד	
	929	NISIZII	1012	ואת כל פרי	
٠	111	ואחרי	11 ³ b	ا 20	
	115	ועד	1215	771	
	12 ²⁹	ועד	13 ⁴	INAQ	
	1514	וירגזו (אד	(د در تارا) 16 ⁵	יום ויום	
	172	IOIA ANI	1710	ואהרן	
	17 ¹⁶	ועך	1813	791	
	1814	791	18 ²¹	ושרי אאות	
	20 ^{17a}	IEW NUCL	2° 2110	ואם	
	2137	10'	224)در	
	22 ⁶	10'	227	NSD. NO DNI	
	22 ⁹	۱۵,	2212	036. 431	
	22 ¹⁴	N21	2216	מאו	
	2229	72ns31	22 ²⁹	מויבו	
	23 ¹¹	71131	23 ²³	וקקוי	
	23 ²⁶	૫ડેા	256	pal	
	25 ⁶	a 'no 21 no	257	 ଜଗଓ '12N1	
	25 ³⁹	14x c8	26 ³	4011	1
	278	121	27 ¹⁶	ן עמוד'הם	
	286	ا ۱۸ گور	30 ³²	ו קדש יהיה	
	31 ⁴	20131	32 ³⁴	וחוק	
			9-	•	

(c) (contd)

Ex.33 ²	והאמרי בונחן	343	1N871 021
34 ²⁰) 7132 331	35 ¹¹	ואת מהצו בביחין
3512	אא הכפרת	35 ¹⁶	ואת צרין
3517	ואת צמדין	35 ¹⁹	1722 AMI
35 ²⁴	וכצ מרים	35 ²⁵	14V EVISEV
35 ²⁹	301	رر 35 31	7112/21
35 ³⁵	1581864	36 ¹¹	121
38 ³	123 FAT ANI AN TATELA	39 ³⁶	32 241
3937	ואת ברתיק	39 ³⁹	ואת בדין
3940	ואת צמדיה	39 ⁴¹	ואת בגדי
Lv.18	ואג דראש	117	431
737		830	11722 391
830	ואת בגדין	11 ⁹	ואת זה
1112	351	1115	AMI
11119	והאנפה ואת הדגיפת	1120	301
139	7211	146	ואג הצפור
16 ⁴	ΛΙΛΙΝ	18 ¹⁴	341
1913	1.3× ×31	10	•···
1919	7761	19 ²⁶	1 EN AE NW1
19 ²⁷	וצא תקיפו	21 ¹⁴	•
22 ²¹	301		-2
23 ⁸	Q1'Z I	23 ⁴	กรีพเ
ر ۵	w t	2331	וכל מצמכה

(c) (contd)				_				
Lv.23 ³⁶	וביום		2)	301				
25 ²⁵	10'		2547	201131	25 ⁵	ספיח	INA	
27 ¹⁷	ad l		27 ³⁰	ו מפרי				
Nu. 27	ומטק		3 ²² 75	०००२ वा	ופקדי			
325	। वटकत्।		419	IINA				
4 ²³	771		4 ³⁶	7201				
510	ואיש אשך		716					
7 ²²	א איך שיוך	7 ²⁸	ו שציר	•	34			
7 ⁴⁰	ן ש עיך	746	ן שציר		7 ⁵²	ושעיר	ŧ	
758	ן שעיך	764	ן שציך		7 ⁷⁰	ושציר		
776	ושעיר	782	ושציך		7 ²⁵	ו ברבנו		
731	וקרבנו	737	וקרבנו		743	וקרבנו		
749	וקרבנן	7 ⁵⁵	וקרבנו		7 ⁶¹	וקרבנן		
7 ⁶⁷	וקרבנן	773	ו קרבנן		7 ⁷⁹	112771		
84	צד פרחיה	1	1111	5 121	ish a	ואו		
126	a 131 2	1	13 ²⁸	8	11378	l t		
14 ¹⁸	0.636 34	ŧ	19 ²	73.	y ~3	76671		
19 ¹²	7761	CMT	191 (יטהר	4	7 W T I	1		
19 ¹⁵	BINDI		211	8 <i>p</i>	ηγινω	1521		
225	הנה כסה	ř	221	1	צתק	1		
2212	5N ANC	!	23 2	5	ברך	וגם		
₂₄ 5	7.71264	1	241	4	78'	141		
26 ⁵	` \J2		26 ³	60	7 3	מ ו	(MT p31	, 3)
26 ³⁵	7341		26 ⁴	.2	73 ~	41		
26 ⁵¹	7261		265		<i>ለበ</i> ១ሬ	ומי	3 4 5	

(c) (contd)				
Nu.27 ²⁰	ا 3 مدا	28 ²¹	ן צשרון צשרון	
2827	3.41	28 ²⁸	1501	
28 ²⁹	ועשרון עשרון	28 ³⁰	γυν	
29 ³	1761	297	301	
2 9 ⁹	101	2910	וצשרון צשרון	
2911	71961	2914	'160	
2916	ነ ላርባሊፍ	29 ²⁴	DANIA	
29 ²⁵	3NA381	2931	TARIAL	
29 ³⁴	latung	29 ³⁵	וביום	
2937	DANIAL	30 ¹⁷	34 1,51	
31 ²	א 3° א ()	31 ²²	IMA	3 * 4 * 5 °
31 ³⁰	{ 30 h 3° 4°	31 ⁴³	V A5 10 1	
31 ⁵²)2 \omega l	₃₂ 5	Sni	
₃₂ 26	ונשינו ומקנינן	34 ⁸	7771	
Dt.1 ²⁸	ואינו	1 ²⁸	וצרים	
213	7/1	3 ⁸	ועד	
3 ¹⁴	ויאיר	437	10021	
5 ⁵	1 466,	58	301	
624 111	איתנו דמ) וצהחית	³¹⁾ 7 ⁶	721	
19h		97	341	
1010	NSI	1013	78631	
1017	והגבור	1020	1441	
111	וחקתין	11_{S}	المر ، دا	

(c) (contd) 144 Dt.1124 ואן הנהר 741 144 149 ITHR INA 1416 1518 14A 5C10 431 16⁵ N31 1619 ACIT 181 1711 ~31 178 1,2 1,51 1916 191 188 341 $50_{\rm J}$ ונם 20⁴ پنونو 31 והכנפני ...והחוי 2017 ואשר צא משכה 213 131 awa 2117 238 EN VVKE 281 λιων3ι ואת המהומה אמו 2837 30 p31 2851 67'A1 2910 2910 001431 771 29²³ 313 90171 מה חרי 3119 322 3111 20,01 32³ 32¹⁰ ויצרנהו ויצרנהן 1271 3211 (1.3712 371 3216 1271121 isnl 32²³ 3220 INPAL (MT 13) 3229 { 1321 | 13241 י וינק 3225

32⁴⁶

3319

33²³

33²⁸

 $N_1 e^{\lambda_1}$

gwl

11871

771

338 172171

32³⁸

33³

33¹²

33²²

33²⁴

11011

321

1.116

1771

0.0.5231

b. Words suggesting misreading, miswriting, or mishearing in S T or M T.

c. Words in identical or similar context

$$36^{21}$$
 (identical context) 700π 677π / 677π Lv.1 9 (1 13 , 1 17) 877π 1^9 (1 13 , 1 17) 877π 1^{17}

Nu.56 (Ex.16¹², Lv.4²)
$$10^{18}$$
 (8^{22,24}) 10^{18} (10^{14,22,25}) 10^{18} (10^{14,22,25}) 10^{18}

d. Words which do not belong to any of the above types.

$$Gn.3^{3}$$
 ה' / הזה אשר 620 ה' / הזה אשר 14^{10} 14^{10} א אשאי היא 16^{7} ועמרה 10^{7} עמרה 14^{10} א אשאי היא 16^{7} ועמרה 10^{7} עמרה 10^{7} וכ 10^{7} אשר 10^{7} ואחרי כן 10^{8} 10^{7} בתנת בני היא 10^{7} 10^{7} 10^{7} 10^{7} 10^{7}

d. (contd)

```
Ex. 316 SNOW 122 / " 520 130 AND / 'D
    627
         ין בני ישראל לפ מחצרים/ מארץ מצרים
    1028 ארץ מפרים 114 בי ד ל ארץ מפרים 114 מ' / ים
    ז' / דברו נא 12<sup>3</sup>
                             123 Smai 12 /1
    י ל בני ישראל 126
                             133 12 OANS! /4
    חמצרים / חארץ מצרים 133
                             1320 YOU DARZ /2
   1412 ~ 7 37 / 7
                             ה ל / הדרך אשר 1820 / ל
   אחד שנים 223
                            226 DIJU TAN / 6
   אחר שנים 825 / ש
                            244 04 07 / 6
   ב' / היריצות תהיינה 263 פ' / פרים בניבקר 245
   ו' / והקטרת אתם<sup>25</sup>פצ ר'/ רצה היא 123' / זכרון הנה
בי / מכל דבר לע.4<sup>28</sup> היא עליה / "חטא עליה בע.5<sup>24</sup> מ' / מטא עליה
Ex. 36 8 ושא ו
                         בעין / מעין מים <sup>36</sup> בע אנין /
Lv.176 בי ל יהוה אשר / "
                         Lv. 21 אשר יהיה / אי / אשר יהיה
Nu.23<sup>26</sup> כ' / כ' הדבר
                           2612 SINDER ADOUN /0100 A3
  26<sup>58</sup>
       צ' / בני צוי
                          2814 { TINIT } 3 / 3 / 3 25 3 NINT | 4
  3356
        צ' / צהם כן
Dt. 16 20 '> איה כ' / 'ב
                           1312 714 1501 / "
  1419 AND 120MA / 1
                           1612 DITER GAR / DITERZ
  276 's 33 IN /'N
                           1711 19 341 / 4
  צ' / צור או 1822
                           ה' / התורה אשר 248
  2411 12 / 2
```

(2) Synonyms giving greater precision or emphasis

(i) Common Words

a. Record of events implied but not explicitly recorded in M T

$$Gn.30^{37a} = 31^{11-13}$$
 $Ex.7^{18b} = 7^{16-18}$
 $7^{29b} = 7^{26-29}$
 6^{1b} in ST adda
 $7^{ba} = 7^{ba} = 7^{ba}$
 $7^{ab} = 7^{ba}$

c. Words in identical or similar context

Gn. $1^{14}(1^{15})$ | Entry | Start | Start

c. (contd)

```
ומבתך/וחבתך וחצבדיך ומצמך (8) א / חרשמי מצרים (22) באר/וחבתן
    8<sup>12</sup>(8<sup>1a</sup>,13) 7002 77. / 1
                                        12<sup>25</sup>(13<sup>5</sup>) הזאג בתרש הזה (13<sup>5</sup>)
    בפרעה ובכל חילו (14<sup>17</sup>) בפרעה ובכל
                                        19<sup>12</sup>(19<sup>23</sup>) את ההר (19<sup>23</sup>) או ההר
   א'ר משה מן ההר /ה
                                        25<sup>9</sup>(25<sup>40</sup>) 772 7AN /N
   27^{12}(27^{11}) \wedge \omega \pi = 7^{12}(27^{11}) \wedge \omega \pi = 7^{12}(27^{11}) \wedge \omega = 7^{12}(27^{11})
   27<sup>15</sup>(27<sup>11</sup>) Awns 7636 2°/6
                                      צ'/ צשרה אמה (27<sup>14</sup>) צשרה אמה (
   27<sup>16</sup>(27<sup>11</sup>) אשתי משבצות זהב ושתי (28<sup>23</sup>(28<sup>11</sup>) אי משבצות זהב ושתי (28<sup>23</sup>(28<sup>11</sup>)
   28<sup>20</sup>(28<sup>11</sup>) Azon 7561 /1
                                      2910(2942) תאם זוחי יהול /3
   39<sup>26</sup>(28<sup>34</sup>) און זהב (28<sup>34</sup>) '5
                                        (bis)
   40<sup>27</sup>(40<sup>23,25</sup> ) אים צפני יהוה
                                      16
  4029(406) 19306/6
בי / הצאן עצה (1<sup>3</sup>) בי און מוצד יקריבעו אל פאח אהל (1<sup>3</sup>) הצאן עצה און מוצד יקריב אאו
    באש/האש אשר על המזבח (18,12) 35 (18,12 אין מנחה היא
    414(3<sup>23</sup>) 27 / 29 / 2 417(46) 177 / 1
   פ' / פרכת הקדש (<sup>6</sup>)
                                      א /מזגח קטרת הסמים (<sup>25</sup>) א
   429 (424,33 72) AN IBAW. TWN DIPPZ / 5
   4<sup>30</sup>(4<sup>25</sup>) א מין מזבח העצה (4<sup>25</sup>) א מין מזבח העצה (6
   56(435, 510) TON ANDER 34/IANDERS 68(112) TON 127 17 / 7
610(611) 13 n3011 /b
   7^{3}(3^{3},94^{8})
                      הקרב/ הקרב ואת כל החלב אשר על הקרב
עציו את האפוד
```

c. (contd)

```
Lv.831 (Ex.2931) של הקום הקדש / ה
    113(Dt.146) אסע אשע אים (1110(119) אין וקשקשא במים ( 1110(119)
    1125 (157,8) בידון ורחף במים (1144) בידון אלהיכם (157,8) בידון ורחף במים (157,8) / יהוה אלהיכם
   14<sup>10</sup>(following phrase) אומם בני שנק / א
   17^4(17^{5,6}) הביאן צעשות אגן עצה או שצמים ציהוה צרצאנכם צריח (17^{5,6}) ויש חטהו בחוף ואצ אהצ מיעד צא הביאו עד בחוף ואצ אהצ מיעד צא הביאו עד שרף שרף טאים חצות (17^{5,6}) שיף טאים חצות (17^{5,6}) שיף טאא שרף טאא עדף טאא אום 17^{6}
    23<sup>18</sup>( סמים אמימם (earlier in verse) מי / שנים אמימם / 'ש
    24<sup>3</sup>(Ex.27<sup>21</sup>) אהרן ובניו (25<sup>35</sup>(25<sup>36</sup>) וחי אחיך (1
48(46,14) YE'ED COL' / Y
    274(277) a3ns ATAN / 'N
    27<sup>12</sup>(customary formula) > not nun / 'n
    אחד 3 חטאת (28<sup>15</sup>,30) אחד 3 חטאת (28<sup>15</sup>) אחד 3 חטאת (28<sup>15</sup>) אחד 3 חטאת (28<sup>15</sup>) אחד 3 חטאת (28<sup>15</sup>)
    28<sup>27</sup>(28<sup>31</sup>) פם (28<sup>31</sup>(28<sup>31</sup>) מימם יהיו צכם (28<sup>31</sup>) יהיו צכם (28<sup>31</sup>)
    29<sup>15</sup>(29<sup>16</sup>) בבשים ונסכיהם / '29
    2922,28,31,34,38(2916) 1017 / 1
Dt.25(29) ארצם ירשק / קוחו פעו (24) ב<sup>13</sup>(224) ארצם ירשק
   231 (234) יסיחן מצד חשבון האחרי / 'o
    3<sup>12</sup>(2<sup>36</sup>, 4<sup>46</sup>) אפט היום (2<sup>13</sup>) אי / על שפא / 'א
```

c. (contd)

d. Words which do not belong to any of the above types

d. (contd.)

```
Ex. 322 אמה שות למשון אשת הלאשו 46 ויוצאה המיקו / 1
  79 ADID IN AIN / 'D 820 THE TED / 'S
  הצם / הצם הזק גם א 3 ב 10<sup>24</sup> אהרן 3<sup>4</sup> אהרן 10<sup>24</sup>
  ב' / בכור אדם 1315 ושמרתם / ושמרתם ונשיתם 1217
  אין אחר שצם ישצם משדהן אין צמ ע' / נחו ויכס משדהן אין כתבואתה ואם כצ השדה 224 ע' / נחו ויכס אין יבעה ז' / זהב טהור 26<sup>37</sup> ל' עיני פקחים 23<sup>8</sup>
         ב' / בוים רבים 3424 ב' בכור אדם
  35<sup>22</sup> 3124 AY25 / 6
הל המזבחה צפני יהוה 20 ה' המשיח משר מצא את ידו ל 14.00
   מ' /מזובן טחא הוא כל ימי זב בשרו או התגים בשרו מזובו
Nu. 1423 , on 3 M3 Onan3/00243 224 ATT 3APT / SAPT
  בי / משה רית 28<sup>27</sup> י מבני יש האל ים
  בי / הנא יציא 306 השביעו / השביעי הזה 2912
שני.9<sup>28</sup> אמרו עם 10<sup>11</sup>
                                 הצם / הצם הזה
  ה' / הישר והטוב 1319 אחיך בן אביך או 137
  כ' / ככ 3 הגורה 24<sup>8</sup> אשה/אשה ובא אציה כ'
  האיש / האיש ההוא 22 צ' / חשך ענן 5 נ' / נחצתן ישראצ 329
 Included under this heading: additional 35 in ST
Gn.14 בל האנשים 30 ו'ר' / ואת כל דכשו 37 / ה
       אשר / וכל אשר 39<sup>23</sup> אשר אכן / רשאו
  4221
        ה' / כל הצרה
Ex.75 01750 30 / 'n 1418 01750 30 / 'n
        34 70 35 / 'T 3434 70N 35 / 'N
  3514 -35 / 7.35
                           ה' / כל המלאכה נצ<sub>00</sub>
```

 $L_{V}.4^{27}$ A(80 320 / A(800) 16^4 1742 32 / 2 20^8 'APO 32 / '0 2726 7122 32 / '2 $Nu.4^{32}$ '32 32 / '32 30^8 7'1 31^{12} ATY 32 / 'y $Dt.11^{32}$ O'02407 32 / '7 16^8 7243 32 / '6 27^{26} '127 32 / '7

iii. Omissions

(1) Pure Synonyms

(i) Common words

a. Stylistic supplements

(a) Explicit subjects or objects

(c)	Addi	tional Waw ("and")		
Gr	1.67	וצד צוף	713	וחם
	7 ²³	914 771	8 ²²	١٩١ ١٩١١
	95	ומיד	9 ¹⁸	οηt
	919	,3001	10 ²	17B1 3ZINI
	103	V5.21	106	७।७।
	107	10201	10 ²³	7121
	1032	1945	1127	7341
	1423	ואס אקח	1912	ובניך
	31 ³⁸	13.41	3410	וסחרוה
	34 ²¹	1201	34 ²⁹	ואת כל אשר בבית
	36 ²⁴	ו איק	36 ²⁶	ויאכן
	4135	ן שמרן	4318	vu 531
	4420	ן אחין	4510	וצמנך
	46 ⁹	וחצרן	46 ²⁴	751
	47 ²¹	ועד	50 ⁸	וימאו
	50 ¹⁵	2 w 11		
Ex.	3 ⁸	יאתהו	317	והחתי
	616	וקהת	618	וחברון
	8 ²⁰	3221	1017	והעאירו
	15,15	ועד בהמה	1221	וקחו
	1315	וצד בכוך	13 5	MADI
	50,10	712721	20178	ועבדן ואורו

-- + -- -

(c)	(contd))
-----	---------	---

() (= = = = =	~ /		
Ex.211	73N1	21 ⁵	anı
5511	DNI	24 ¹²	150315
24 ¹⁴	והנק	2 5 ³	9051
25 ⁴	NSONI	2531	וקנה
56 _J	12050	2616	ואמק
26 ²⁴	ויחדו	27 ¹¹	ועמדן
2714	ψηαι	284	ICALA
2820	0761	₃₀ 27	ואא השלחן
30 ²⁷	ואת השלחןואת המנורה	30 ²⁸	ואת מזבת ואת הכיך
31 ⁸	ואת השצחן	31 ⁸ {	ואת השצמן ואת המנורה ואת מזבח
31 ⁹	ואת הכיור	3110	1712 ANI
3111	ואת שמן	35 ⁵	9001
35 ⁶	14084	35 ¹⁴	אאר מצובע
35 ¹⁵	INA ATER	35 ²²	ΛΥΖΟΙ
36 ⁸	NSONI	36 ²¹	ואמק
36 ²⁹	וימדן	36 ³⁸	ואת צמודין
3717	1915	38 ²⁵	7661
₃₈ 28	74711	39 ²⁹	Nesk
3 9 ³³	ועמדין	39 ³⁸	14A 0150
Lv.39	JAN 6 085	6 ¹⁵	ן ה כהן
7 ¹⁹	וה בשר	1146	8-31
136	N31	15 ⁹	וכ3 הוגע
1633	וע ב הכים	2019	AITYI
^{51}s	11231	26 ⁴³	ו בי צן
			•

				Annual Contract of the Contrac	
(a)	(contd)	•			
Nı	a.3 ¹⁷	1911	3 ³⁸	ואהרן	
	4 ³⁴	١ 3 ١ ١	4 ³⁸	۸۱231	
	₄ 46	זצבית	16 ⁵	והק ריב	(יקריב זצ)
	1911	וטמק	23 ¹⁹	ודבר	
	23 ²⁰	וברך	(ST 772N) 2424	1391	
	26 ⁸	1721	26 ⁹	المر	
	26 ³⁴	ופקדיהם	36 (צפקד יהם דצ	1485	
	27 []]	וחגצה וחצבה	28 ¹⁶	<i>ነ ደ</i> ነተው	
	28 ²⁶	וגיום	34 ⁶	והיק	(יהיה בא
	3420	18 तथ द	₃₄ 22	וצמטה	
	3611	חגלה ו מלכק			
Dt.	. 17	22121	112	ashuni	
	115	ושרי חאשים	2 ³⁴	והנשים	
•	3 ¹⁶	ולד יבק	4 ⁴⁵	והחקים	
	5 ⁹	0.636 311	51.4	77271	
	517	NŠI	5 ¹⁸ d	א אחמך	131
	5 ²⁸	והחקים	67	727621.	- 710321
	611	וצעים	620	והחקים	•
	7 ¹¹	ואת ההקים	7 ¹³	ותירשך	
	7 ¹⁹	1571	87	ι Λπα λ	
	88	ו גפן ותאנה		וצקרב	
	113	الهم ممررا	1113	18 4271	
ì	1114	70711	1112		712321
	1212	ועגדיכם	1217	•	
	1218	77241	14 ⁵	1	
				•	

(c) (contd)					
Dt.14 ¹⁸	א דבן כיפת	1426	11,31		
152	וימא אאו	158	והצבט		
1611	ועבדך	1624	77241		
1810	ומנחש ומכשף	1811	3 NW 1		
1822	וצא יהיה	19 ¹¹	۱c'		
19 ²⁰	N31	21 ¹⁹	ואצ שער		
2122	ا د'	223	121		
24 ⁸	116431	26 ¹³	7231		
26 ¹⁹	3١ م	27 ¹²	90111	ויהודה	
27 ¹³	131211	28 ⁵⁶	77221		
28 ⁶⁵	ו כציון	32 ¹³	3041	(57	(ימכצהו
32 ¹⁵	10011	32 ²²	35 MAI		
33 ¹⁷	ודם רבבות	33 ²⁰	וטרף		

d. Words which do not belong to the above type.

Gn.17 [±]	Ωιγφλ	א' שנה /	27 ⁵⁵ NI	א' הוא / אפ
Ex.3 ¹⁸	11.34	/ 7AY1 'Y	924 0128	בכל ארץ מזרים / במ
116	במצרים	ארץ מצרים /	3>2	
Lv.435	הכהן	/ DAN 17	8 ²⁸ nn	נ' הוא / ני
1444	ΛΙΖΖ	/ ב אוח	20 ²⁶	כ' קדוש / כ
27 ²⁵	ペ フス	(ג' יהיה		
Nu.4 ¹⁹	414	/ איש איש	23 ¹³	אחר אשר/ אחר
26 ¹⁸	72	בני ג' /	28 ²¹ (17	ע ¹ אעשה / עש
Dt.147	הפרסה	הפרסק השסועה		·
32 ²⁸	DVIEA	עפוג המה/		

(2) Synonyms giving greater precision or emphasis

(1) Common words

c. Words omitted in identical or similar context.

$$Ex.21^{18}$$
 (21²⁰) נ' לפני יהוה לניתות (29²⁵(29¹⁸) באבן או באברף רנהו (21²⁰) כ' איש / כ3 (35²¹,23)

d. Words which do not belong to any of the above types.

d. (contd)

2. Differences of vocabulary with divergence in sense

1. Doctrinal

(1) Controversy regarding Gerizia

Ex.20^{17b} See above, p.109.

 Ut.5^{18} See above, p.109. 11^{30} שנם 30 אינה 4 בריזים 4 בריזים 4

(2) Other Samaritan beliefs and practices

(i) Majesty of God

$$Gn.48^{16}$$
 7^{3} 7^{8}

(iii) Day of Vengeance and Recompense

(1V) Legal enactments

(V) Veneration for religious leaders.

(Vi) Genealogies of patriarchs

*			
	Years.	•	•
Gn.5 ¹⁸	. 62)	
5 ¹⁹	785	\$	Jered
5 ²⁰ 5 ²⁵ 5 ²⁶	847	3	
5 ²⁵	67)	
> 5 ²⁶	653)	Methuselah
5 ²⁷	720)	
5 ²⁸	53)	
5 ³⁰	600)	Lamech
5 ³¹	653)	
11,12	135)	J.
1113	303)	Arpaksad

(V1) Genealogies of patriarchs (contd)

	Years.	
Gn.11 ¹⁴	130)	Shelah
1115	303)	01102011
1116	134)	Eber
1117	270)	11001
1118	130)	Peleg
11 ¹⁹	109)	" OT OF
1120	132)	* **
1121	107)	Reu
1122	130)	G O MIL OI
11 ²³	100)	Serug
1124	79)	Nahor
11 ²⁵	69)	21(0230)1
11.32	145	Terah

(Vii) Sectarian opposition

$$7^{60}$$
 10^{24}
 13^{9}
 $26^{38},39,41$
 34^{21}

ii. Non-doctrinal.

(1) Variants where there appear to be cogent reasons in favour of S T

Substitutions

M T readings are difficult in grammar and/or sense, sometimes to the point of unintelligibility. They are usually explicable as due to the typical causes of scribal error.

Additions

Gn.
$$4^8$$
 Words appear to be missing $8^{\frac{1}{4}}$ M T 1^8 Gn. 4^8 1^8

Additions (contd)

ומספר את רבע י / ומי ספר מרבעת ישראל 10+ 23 ומיספר

ויש מן / יאכצ יעקב וישבע ישמן / ארכצ יעקב וישבע ישמן / וישמן /

Omissions

הצוף קרח אלוף בעתם / אלוף בעתם אלוף בעתם פו פבב בצתי ביהוה בבדו / באלהים אחרים בצתי ביהוה בבדו / באלהים אחרים בצתי ביהוה

ויהיו בני בצע ארד ונצמן / צארד 2640+ ממע אמריאצ אשר / אשר 244+ משר א בני בריצה צחבר / צחבר ב645+

(2) <u>Variants where there appear to be agent reasons in favour of M.T.</u>

Substitutions

(i) Euphemistic words

שבל בה / ישכב 28³⁰ בחבשיו / בבשרו 30 בשרו

(ii) More common word

Gn.116 IJOT' / IOT' of Petermann, p. 21.

2116 (1605 / 1005) 496 70. / 70.

ע? ברכי / ברכי / ברכי / איז ברכי / ב

 $E_{x}.2^{23}$ $In_{x}.1 / In_{x}.1$ 2^{24} $O_{x}.1 / O_{x}.1$ 6^{5} $A_{x}.1 / A_{x}.1$

(ii) Contd.

Iv.17¹³(17³,8,10)

$$Nu.5^{26}$$
 0 $1 \pi r'0$ 0 $1 \pi r'0$ 0 $1 \pi r'0$ 11^{25} $3 \pi i$ $1 \pi r'0$ 11^{32} $1 \pi r'0$ 1

(iii) Words) in identical or similar context

Gn. 10¹⁹(15¹⁸ Dt. 11²⁴) 3173 Fig. 1215 Fig. 1215 Fig. 1215 Fig. 1215 Fig. 1216 Fig.

מבני / מבית

202(173,8,10) 120 / 120

(iii)(contd)

Additions

Word(s) in identical or similar context

Additions (contd.) $Ex.27^{19b}(25^4, 28^{45})$ $Ex.27^{19b}(25^4, 28^{45})$

(immediately after verse 19)

 $28^{30}(27^{19b})^{30}$ (27 19b) 30

 $Nu.17^4(17^2)$ $Nu.17^4(17^2)$ $Nu.17^4$ $Nu.17^4$ Nu

Longer Passages

Incorporations of passages from parallel narratives

Dt.5²⁸⁻³¹ Ex.2021b **1**6p-8 10¹⁰b acs-dos 12^{16b} 127-33 13^{33b} 14^{41a} 742 Wu.lolob 16b-8 12^{6b} ₁20b-25a 13^{33b} 127-33 14⁴la 742 1445 ST adds from Dt.144

Nu.20^{13b} =
$$3^{24-28}$$
 (omitting 3^{26a}) = 2^{2-6}
 21^{12a} = 2^{9}
 21^{13a} = 2^{17-19}
 21^{21a} = 2^{24-25}
 21^{22a} S T adds from Dt.2²⁷
 3^{27} 3^{2

Substitutions of passages from parallel narratives for a shorter original.

Ex.
$$18^{25}$$
 adds 70^{21} ST = Dt. 1^{29-18}
 20^{19a} yrwii ... 1^{20} ST = Dt. 5^{24-27}
Nu. 21^{22b} 7^{322} ... 8^{3} ST = Dt. 2^{28-29a}
Dt. 10^{6-7} ST = Nu. 33^{30-35}

Omissions

· A mibrogga

(3) Variants where there appear to be no cogent reasons in favour of S T or M T.

Substitutions

(i) Attributable to the typical causes of scribal error

$$Gn.14^{14+}$$
 $7711/9711 15^{10}$ $7102/7102 18^{15}$ $73/83$
 32^{10} 1019

$$11^{3}$$
 $| 10^{1} | 10^{1} | 10^{1} | 10^{1} | 10^{1} | 10^{1} | 10^{1} | 10^{1} | 10^{1} | 10^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} | 11^{1} |$

(ii) Occurring already in identical context, and equally suited to it.

(iii) Not belonging to either of the above types.

$$Gn.2^{2+}$$
 $Gn.2^{2+}$ $Gn.2$

Additions

(i) Attributable to the typical causes of scribal error.

$$G_{10.20}^{11}$$
 $G_{10.20}^{11}$ G_{1

(ii) Word(s) in identical or similar context.

(iii) Not belonging to either of the above types.

$$Gn.2^{24}$$
 $On.2^{24}$ $On.2$

ואביהוא וחביהוא אצנור ואיאמר 24 ואביהוא / ואביהוא אצעור ואיאמר 24 אביהוא אצעור ואיאמר 24 אביהוא אצעור ואיאמר

32 NW DANGT / 1

Mu.14¹²⁺ 7'24 A'Z AH 7AN / 7AN 26¹⁰ 9747 OAN / 'N
51¹⁵⁺ OA TOS / '7

Omissions

(i) Attributable to scribal error.

והראה ואת האיה והדיה / ואת הדאה ואת האיה באיה באים

(11) Occurring already in identical context or equally suited to it.

Gn. 3139+ יזיה / מידי לה

APPENDIX B.

Additional Paper on two Samaritan MSS.

1.

Facsimile of photograph of a portion of a Samaritan Pentateuch Scroll in Nablus.

The photograph, which was in the possession of the late Professor John Garstang, was of a MS affirmed by the Samaritans to be the oldest in their community and to have been written by Abisha the great grandson of Aaron thirteen years after the entry of the children of Israel into Canaan - the claim made for the Abisha Scroll. This photograph was made available to me for examination and I transcribed and collated with the von Gall text, when it was found to be clearly inferior as shown below. It appears that the Samaritans had followed their practice of exhibiting another MS and not the Abisha Scroll In the Pedersen Festshcrift Kahle refers to a specimen page of a Samaritan Scroll, containing the text of Lv. 15 and 16, and sent by Garstang to Sir Frederick Kenyon. and by him to A. E. Cowley 2. This may be the same photograph as I have examined.

General description of the MS.

At certain places the MS has been destroyed and there are lacunae in the text. At others the letters can be discerned only with extreme difficulty. On occasion, and especially at the end of a line, the letters become almost microscopically small in order to admit of the completion of a word within the line. The first letter of each line is usually separated by a small space from the remaining letters of its word and stands immediately under the first letter of the preceding line. Often this holds good also for the last letter of each line, or the last two letters.

Interpunction.

where the writing is not too cramped the words are sometimes separated by a dot. This is generally the case when letters are being written somewhat larger and especially when spaced out to fill up a line e.g. the final line of Chapter 15. The sign : is used to indicate a pause at the end of or within a verse. The sign -: or -.'. indicates a longer pause e.g. at the end of a section. Embellishments such as --.'. --- appear in the blank line that separates the D'SP when such coincide with one of the D'W79 e.g. at the end of Chapter 15.

Collation of MS with von Gall text.

Complete collation of which the following is a summary.

Divergences from von Gall text 7.

These seven variants admit of a twofold sub-division.

1. Variants that agree with Samaritan readings listed in the von Gall critical apparatus but not followed by him.

<u>MS</u> •		von Gall, with MT
Lv.15 ¹⁹	7211	725
1610	יצמיך	יצמד

- 2. Variants that stand alone, without support from either the von Gall text or his critical apparatus and without support of MT or LXX.
 - וא. 15²⁵ Omits אהינה before ביאי ודאה, leaving a clear gap in the text.
 - Reads 20% for 200%

 Either word might be correct. 200% appears the more natural word in line with 200% but for that reason is not necessarily correct. 20% occurs in both later in the verse in relation to '30%. The general context is indecisive as verse 20 distinguishes "lying" and "sitting" but verse 23 has the phrase "and if upon the bed or anything whereon she sitteth".
 - Appears to read yan for yain but considerable blurring in the MS.
 - 0mits guttural in final word and reads the meaningless אר for אונס אר
 - Additional DAN after $7\omega y$ a word that could easily be added.

In the light of these foregoing readings it cannot be/

be maintained that the text of the MS, in the passage available for examination, is superior to that of von Gall. On the contrary it is evidently inferior.

Footnotes, Appendix B. 1.

- 1. Thomson, The Samaritans, p. 403
- 2. Kahle, Pedersen Festschrift, p. 189.

.

APPENDIX B.

2.

The Manuscript of the Samaritan Pentateuch.

R. 16.41

Trinity College Library, Cambridge.

In the Cairo Geniza, p.49, Kahle lists the vowel signs commonly used in Samaritan MSS and in Note 6 to this page he mentions that only a few Samaritan MSS of the Pentateuch are provided with this kind of punctuation. He instances, among others, a Samaritan Pentateuch in Trinity College Library, bought 1917, dated A.D.1332, although he does not give further details regarding it. This MS is not included by von Gall in his Introduction as having been used in his critical edition of the Samaritan Pentateuch so that to my knowledge there is no published account of its nature or value. I decided to make a full examination of it and visited Cambridge on two occasions when I had opportunity of studying the MS in detail, as a result of which I now provide the following description.

General Appearance of the MS.

The MS is written on paper and is the work of three hands which will be designated R^1 , R^2 , and R^3 . Where the oldest hand is missing there are supplements by one or other of/

of the two later hands. The second hand has also been repaired by the third and final hand. The oldest part of the MS is composed of cartabombycina, a paper which the Arabs received from China at the beginning of the 8th century and used well on to the 15th. The paper of the second part of the MS, while more recent, is also Eastern and without any indications such as watermark. The paper of the third and latest part is quite fresh and modern.

The MS has half-leather binding, with a flap ornamented in typical, oriental fashion. The sides are of paper over cardboard. In all probability the binding is 18th century, as it appears to have been done by the second hand - according to the MS of date 1754 A.D. See below, p.

Size of the folio is approx. $10\frac{1}{2}$ " x $8\frac{1}{2}$ ", quarto.

Fly-leaf has a long Arabic inscription completely filling a and overflowing to b.

The MS is written right across the page and there is no Aramaic or Arabic rendering. The oldest part, hereafter designated R¹, has been supplemented and repaired by two later hands, R² and R³: these three parts will now be considered from the following standpoints:-

- 1. Distribution of the text among the 3 copyists
- 2. Date and general characteristics
- 3. Vocalisation.

1. Distribution of the text among the 3 copyists.

As in all Samaritan codices folio la is blank. The writing of Genesis 1. begins at the top of b.

Rebrew letters enclosed in brackets are supplied by a later hand. Lacunae are denoted by square brackets.

G	O	n	Θ	s	1	S	

R^2	10 ff	11 - 1115
\mathbb{R}^{1}	7 "	ארבע ארבע - 19⁵ (פר - ארב ע ארבע ארבע
\mathbb{R}^3	15 "	195 73:37 - 3116 13:5231
R^{1}	5 "	31 ¹⁶ - 14 - 35 ⁴ 173×
R^2	9 "	354 במצרים 42 ¹ באירום מוא הנואים
$_{ m R}^{f 1}$	13 "	421 100(11) - 50 ²³ 212
$_{ m R}^{f 2}$	1 f	במצרים 50 ²⁶ - גם 50 ²⁶
Exodus		•
\mathbf{R}^{1}	62 ff	

Exodus		
$\mathbf{R}^{\mathbf{I}}$	62	ff

<u>Leviticus</u>	
7	
R*	

Numbers
M

R^{1}	42 ff	11 - 24 ²¹ 7001
R^2	1. f	24 ²¹ N'N - 26 ⁴ 700
R^{1}	17 ff	264 121 - 368 SATE. 1°
R^2	1 f	36 ⁸ כיחו [٠] ריחו [٠]

neuteronom	Σ.					
$_{ m R}$ 1	21	ff	1.1	- 12³¹ 7	173~	
\mathbb{R}^2	1	f	12	'> /° -	13 ¹⁷	ה ציך
R ¹	8	ff	13 ¹⁷	30 NNI -	2110	(1') 2 w
\mathbf{R}^{1}	1	f		N1671 -		
\mathbf{R}^{1}	9	ff		'(⊃) 1°		
$\mathbb{R}^{\mathbf{Z}}$	4	ff	2 8 ⁶⁴	۶۶ -	31 ¹⁹	(7,3)
\mathbf{R}^{1}	3	ff	31 ¹⁹	3×761 -	33 ⁹	(ENCN'N')
R ³	1	f	33 ⁹	1 1 NA	347	
R^2	1	f	34 ⁸	12211 -	3412	(3 47 61)

- 2. Date and general characteristics.
 - R¹. According to the Cryptogram, which begins at Exodus 15²³ and ends at 18²⁵ (see below, pp.275-6) the writer was Sa'dallah ben Yitran of the Bene 'iqara, date 733 A.H. (1332 A.D.)

R¹ contains 26 lines to the page and about 26 letters to the line. It is written in a clear, legible hand and the letters have been executed with a certain care. There are fairly frequent mendings and repairs by the two later hands R2 and R3, although in the great majority of cases these strips are comparatively small and contain only a few letters. On the whole this part of the MS is in an astonishingly good state of preservation. The interpunction is clear and simple, with a dot separating each word, and paragraph If the end of a paragraph coincides ending -: with the end of one of the ary ? . it is followed by a blank line, or lines, and sometimes this space receives typical embellishments such as (after Genesis 2^7).



R¹ makes the lines of writing fit exactly over each other. If the scribe, writing a word near the end of the line, thought that he might have insufficient room for the inclusion of the following word, he adopted/

.

adopted quite an ingenious device. He completed his word, except for the final letters which he placed in a group close up to the margin, with the result that there was a space between them and the previous letters of the word. This space varies greatly; it may be half an inch in length, or an inch, or still more. But naturally it sometimes happens that there is no need to employ such a space viz., when the number of letters just fills the line. The whole point is that the two final letters of each line are placed directly under each other and in this way the lines are all of equal length. This method of writing is followed by many Samaritan MSS. Unlike the Jews, the Samaritans did not employ 'literae dilatabiles' to fill up the line to the margin.

There are also instances in R¹ of identical letters and words written underneath one another, perhaps for a dozen lines or more. For example, in Leviticus 18⁴ (and ff) and extending for 19 consecutive lines the letters 3 are written exactly underneath one another. Again, in Leviticus 26¹² if the words

Alabara are written underneath one another for 8 consecutive lines on a of the folio and for 19 lines

on b.

Again, there are certain passages written in hemistich fashion. The lines are split in the middle for the space of rather more than a letter, and the result is a blank column running down the centre of the text. Very frequently, also, the last two letters at the end of the half-lines are written together in the same way as the last two letters at the end of the complete lines. This division of the text into two appears to be an external one and there is little regard for the sense of the passage: least, the blank central column does not necessarily coincide with interpunction marks, for frequently it cuts right through a phrase or even a word. Yet the Samaritans must have had some understanding regarding what portions of the text were to be written in this distinctive manner as many of their MSS agree in this usage .

The passages so written in R1 are:

- 1. Exodus 15¹⁻²¹
- 2. Numbers 21¹⁷⁻²⁰
- 3. " 237-10
- 4. " 23¹⁸⁻²⁴
- 5. " 24³⁻¹⁰
- 6. " 24¹⁵⁻²¹
- 7. Deuteronomy 3130 3243
 These/

These passages are all of a rhythmic or poetic nature and perhaps this manner of writing was an indication of poetic form, just as in a somewhat analogous fashion each line in an English poem usually begins with a capital letter. To the objection that other poetic passages of the Pentateuch are not written in this way, the reply might be made that perhaps the poetic nature of these passages was not clearly discerned by the Samaritans - at least to begin with.

On a of the second R^1 folio of the MSA (Gen. 12^{7-9}) there is a short section, seven lines in all, that shares to some degree in this hemistich writing. The lines are divided into two with the customary space running down the centre of the text. However, the inner half-line ends in one letter only and not The endings of the outer half-lines are all lost and there is a vertical strip by a later hand. R. which to some extent observes two final letters at the end of the line. The outer half-lines all commence with | and, as might be expected, these are placed directly underneath one another and indeed separated from the following letters by a small space. Two other passages may be mentioned: in Exodus the Ten Commandments, including the Samaritan 'Tenth Commandment 1/

Commandment' are numbered by letters in the margin² and also in Leviticus the πsp , $9^{22}-10^2$, giving the Aaronic Blessing, is darkened probably by the kisses of devout readers.

R1 uses the device of the cryptogram, or to give it the Samaritan name. taskil: the line is divided in the middle, so that an interval is left, usually rather more than the breadth of a letter. letters of a word may be on one side of this space. and the remainder on the other. This interval is sedulously observed from line to line so that the result is a vertical column cutting right down the centre of the text and sometimes running through quite a number of pages. Every now and then a letter of the text is placed inside this vertical The letters in this central column can be read, not only horizontally in the actual lines in which they occur, but also vertically, and when so read give information regarding such things as the writer, owner and date of the MS. Frequently each letter in a cryptogram has a small dot placed above it, and quite regularly the final letter of each word a small line placed underneath it. Four cryptograms have been observed in R1. The largest occurs. Exodus 1523-1825, and states that Sa'dallah ben Yitran of the Bene 'igara wrote the

MS for Abraham ben Ab Romemuta of the Bene Mitaba in Egypt in 733 A.H. (1332 A.D.) It adds that it was the first (Torab) that he had written and concludes by praising God. A short cryptogram occurs halfway through the Pentateuch, Leviticus 7¹²⁻¹⁶, [7107 '57 Traces of others occur, Leviticus 27¹⁸⁻²¹] AN 1...AIADD [74 and Deuteronomy 32⁵¹⁻⁵²] 7³ h ...

At the end of Exodus, after the summation of the there comes a colophon stating that the MS belonged to Ah Geluga ben Ab Romemata ben Abrham of the Bene Mitaba in 77 A.H. (1376 A.D.) and with the blessing "let him teach thereby sons and sons' sons".

At the end of Exodus R¹ gives the summation of the D'sp as 200. A later hand, R², supplies the ending of the other books (see above, p.268), giving the summation for Genesis, Leviticus and Numbers as 250, 134 and 220 respectively. For the ending of Deuteronomy see R² immediately following.

At the end of Deuteronomy, unlike the other books, there is no summation of the p'sp. On the lower half of a of the last folio there are a few lines of Arabic writing (see below, p.278) and on b of this folio is a curious, rhyming poem, a kind of catena of scriptural phrases, ending with conventional reference to Joseph, Moses/

 \mathbb{R}^2

Moses and Aaron, and giving the name of the writer, Shalama ben Jacob ben Ab Sakwa ben Abraham of the Bene Danaf from Damascus and the date 1168 A.H. (1754 A.D.)

R² contains 35-42 lines to the page and about 26 letters to the line. Although the letters are quite legible, they are generally not executed with such care as in R1. An incomplete word at the end of a line is sometimes completed by one or two letters in the margin. There are 15 instances of one letter in the margin and 3 of two in Genesis 11-1115: in Genesis 354-421 there are 10 instances of one letter and 4 of two and an instance of a whole word 3~ \nabla /). In this way the letters are not crowded and yet the word is completed before passing to the next line. Accordingly, R2 does not adopt the device of R1 to make the lines fit exactly over one another. Yet, apart from the above mentioned marginal letters, there is a general attempt to make the lines approximately the same length. Sometimes the last word of the line has the interval between each letter lengthened out so that the word may complete the line. At the end of some lines the final letter or, more rarely, the last two letters may be written close up to the margin and with/

with a short space separating from the previous letters, in a manner somewhat analogous to that followed regularly by R¹. But no definite rule is followed by R² and very frequently the line is a solid block of closely written letters without any of the devices just mentioned.

R² has no cryptograms. There are no instances of identical letters or words written below each other. line upon line. Finally, there is no attempt to preserve the distinctive form of a passage written in hemistich fashion. For example, Numbers 24¹⁵⁻²¹ (R¹) is written in this way. Immediately following, the MS has been repaired with an R² folio which is written in the ordinary fashion. Now the poem does not conclude till 24²⁵ and presumably in the original the hemistich style of writing would have extended that length - at least it does do in other Samaritan MSS.

八人

 \mathbf{R}^{3}

At the end of Deuteronomy there are some lines of Arabic saying (see above, p.276) that this Torah was finished by Jacob ben Halil Al-Mufarraq in Bait Furiv in 1331

A.H. (1913 A.D.) On the flyleaf there is a long Arabic inscription, completely filling a and overflowing to b (see above, p.269). The "miserable, poor one", Sa'd, son of the late Sallam, son of Ismail, son of Salih, recounts in considerable detail three dreams regarding Paradise/

Paradise and praising Moses, the "apostle of truth". The date at the end is 1294 A.H. (1877 A.D.) R3 contains 36-38 lines to the page and about 29 letters to the line. The letters tend towards a flowing, cursive style and appear to have been more hastily written than in R or R2. A pencil square is drawn on the page and the writing is contained within this. The line is generally full of letters as in R2. Occasionally a letter will fall outside if the line is very full or sometimes the letters will be crowded to try and get them into the square. In the latter case the letters often slant upwards to the left: an instance of this occurs in Genesis 203. R³ has no cryptogram. It furnishes no examples of identical letters or words written underneath one another. Neither do any passages appear to be written in hemistich form. There may be a certain amount of doubt regarding an R³ strip that mends Numbers 24 15-21, where R is written in this way: some attempt appears to have been made to preserve the R1 form and possibly it broke down on account of the fact that 39 letters had to be written by R3 in a comparatively restricted In other places, e.g. the extensively repaired Deuteronomy 3130-3243, R3 makes no effort to preserve the hemistich form.

Finally,/

Finally, some general conclusions are now suggested with regard to the writing and general characteristics of the MS.

- 1. The earliest copyist, R^1 , has proceeded with the greatest care, R^2 , while less carefully executed, is superior to R^3 .
- 2. R¹ has obviously had a certain understanding regarding a distinctive way, such as hemistich form, in which certain passages of the Pentateuch were to be written. Many of the Samaritan MSS agree in writing certain passages in this way.
- 3. R¹ has a long tasqil peculiar to this MS but also shows traces of tasqils common to many Samaritan MSS.
- 4. Such distinctive modes of writing are presumably very old. A tradition like this takes time to develop especially among a people so conservative as the Samaritans.

3. Vocalisation.

R¹, R² and R³ all employ supralinear vowel signs. The vocalisation is not complete. According to Kahle, the Samaritan vocalisation is similar to the Palestinian³, which was more primitive and imperfect than the Tiberian⁴. The first stage in vocalising a text would be to indicate pronunciation by occasional vowel signs and only a later stage would the text be fully vocalised as in the Tiberian⁵.

This system of occasional indication of the pronunciation is clearly revealed in the three copyists of In relation to other Samaritan Pentateuch MSS. few of which have vocalisation at all, R1 is quite richly supplied with supralinear signs, and both R² and R³ also indicate pronunciation in this way, though more sparingly. In my opinion the evidence now to be furnished is valuable in that the supralinear system of the MS is different from that noted by Kahle as the normal Samaritan system (Geniza, p.49). Some of the signs are different, and they are greater In the statement of the vocalisation signs, in number. that is now to be given. Petermann's transliteration is added to the Hebrew though it is to be remembered that the Samaritan pronunciation in his day may have differed from that prevailing at the time when the MS was vocalised. I have also considered the views regarding Samaritan vocalisation held by him, 6 . Z.Ben Hayyam Von Gall and others.

Vocalisation signs of frequent occurrence.

$$\frac{\sqrt{}}{2} = 0.4$$

$$\frac{\sqrt{}}{2}$$

Used with ambiguous forms, and will have a specific significance in each case

Vocalisation signs of less frequent occurrence. Restricted to initial gutturals.

These two signs were noted as far back as Cowley, who was of the opinion that they implied something as to the quality of the guttural, perhaps that it was to be pronounced instead of being as usual silent.

The nature of the sound remained obscure. It occurred to the present writer that conceivably was a small Arabic ain written above the guttural. Both Kahle and Roberts refer to abbreviated Hebrew letters written above the consonants to

which they applied. 10 If so, above a guttural, or gutturals, would indicate that the gutturals in the text carrying this sign were to be read as . Another problem was the relationship between and , and after some search in the Pentateuch the following passages were found quite close together:

Clearly the signs could be used indiscriminately in certain cases. With the known preference of the Samaritans for the vowel "a" especially with gutturals, the presumption might be that 'was followed by "a", thus 'a. It was possible that "a" might be indicated by a small Arabic alif above a consonant, and thus 'a might be represented by a or combining them, 's. This conjecture has been strengthened by Ben Hayyim's view that both 'and indicated the same sound: when the initial guttural only (i.e. without the following vowel) was represented, 'e was used. However, the vowel of initial 'was invariably "a" and when this vowel, already sounded, was actually written, is or 'would result.

References have been made elsewhere in the Thesis to

the fact that the Samaritan pronunciation of Hebrew was different from that of the MT. This different tradition of pronunciation is still further evidenced by the foregoing examples of these supralinear vocalisation signs.

Footnotes, Appendix B. 2.

- 1. Cf. the list given by v. Gall, Introduction p.LXVI.
- 2. As also in 3 MSS cited by v. Gall, Introduction, p.LXVI.
- 3. Geniza, p. 49.
- 4. B.J.Roberts, The Old Testament Text and Versions, p. 58.
- 5. E. Wurthwein, The Text of the Old Testament, p. 17.
- 6. Petermann, Formenlehre, pp. 4-5.
- 7. Von Gall, Introduction, p. LXVI.
- 8. Z. Ben-Hayyim, "The Samaritan vowel-system and its graphic representation" in <u>Archiv Orientalni</u>, XXII, Prague, 1954, pp. 515-30.
- 9. A.E.Cowley, Samaritan Liturgy, Oxford, 1909, p. xxvii.
- 10. Geniza, p. 48. Roberts, OT Text and Versions, p. 54.

.

ABBREVIATIONS.

Texts.

DSS Dead Sea Scrolls Texts in Discoveries in the

Judaean Desert: I, II, and III.

MT Masoretic Text.

ST Samaritan Text.

LXX Septuagint

Periodicals.

AJSL American Journal of Semitic Languages and

Literature. Chicago.

BZAW Beiheft zur Zeitschrift für die alttestamentliche

Wissenschaft. Giessen.

HUCA Hebrew Union College Annual. Cincinnati.

JBL Journal of Biblical Literature. Boston, Mass.

JJS Journal of Jewish Studies. London.

JTS Journal of Theological Studies. London.

TSK Theologische Studien und Kritiken. Hamburg.

VT Vetus Testamentum. Leiden.

VTS Vetus Testamentum, Supplements to (Vols.I-IV). Leiden

ZAW Zeitschrift für die Alttestamentliche Wissenschaft,

Giessen.

ZDMG Zeitschrift der Deutshen Morgenländischen

Gesellschaft. Leipzig.

General Abbreviations and Signs.

Gn Genesis

Ex Exodus

Lv Leviticus

Nu Numbers

Dt Deuteronomy.

General Abbreviations and Signs (cont.)

BDB Brown, Driver and Briggs: Hebrew and English

Lexicon of the Old Testament. Oxford, 1906.

GK Gesenius Hebrew Grammar edited and enlarged

by E.Kautzsch. 2nd English ed. by A.E.Cowley.

Oxford, 1910.

ICC International Critical Commentary.

Pf. Perfect

Impf. Imperfect

Imv. Imperative.

Inf. estr. Infinitive construct

Inf. abs Infinitive absolute

Pt. Participle

Equivalent

Parallel

V Root, stem

' Sign of abbreviation (in Hebrew words)

First, Second, Third ... occurrences of same letter or word in the verse.

......

London, 1958.

BIBLIOGRAPHY.

Very early books, though consulted, have been omitted from this list.

Books.

- Burrows, M. The Dead Sea Scrolls. London, 1956.

 More Light on the Dead Sea Scrolls.
- Castro, F.P. <u>Sefer Abisa</u>. Edicion del fragmento antigua del rollo sagredo del Pentateuco Samaritano de Nablus. Madrid. 1959.
- Cowley, A.E. Samaritan Liturgy. Oxford, 1909.
- Cross, F.M. Jr. and
- Freedman, D.N. Early Hebrew Orthography. New Haven, 1952.
 - The Ancient Library of Qumran. London, 1958.
- Diening, F. <u>Das Hebräische bei den Samaritanern</u> (Bonner Orientalisfische Studien, Heft 24).

 Stuttgart, 1938.
- von Gall, A. <u>Der Hebräische Pentateuch der Samaritaner</u>. Giessen. 1918.
- Gaster, M. A Samaritan Scroll of the Law of the X1th Century. London, 1901.
 - The Samaritans (Schweich Lectures, 1923).
 London, 1925.
 - Samaritan Oral Law and Ancient Traditions.
 London, 1932.
- Geiger, A. <u>Nachgelassene Schriften</u> 5 vols. Berlin, 1875-8.
- Gerleman, G. Synoptic Studies in the Old Testament.

 Lund, 1948.
- Gesenius, W. <u>De Pentateuchi Samaritani Origine Indole et Auctoritate Commentatio Philologica-Critica</u>.

 Halle, 1815.

- Heller, Ch. The Samaritan Pentateuch. An Adaptation of the Masoretic Text. Berlin, 1923.
- Kahle, P. The Cairo Geniza (Schweich Lectures, 1941).
 London, 1947.
- Kennicott, B. Vetus Testamentum Hebraicum cum variis lectionibus. 2 vols. Oxford,1776-80.
- Macdonald, J. Memar Marqah. BZAW, Heft 84. 2 vols. Berlin. 1963.

The Theology of the Samaritans. London, 1964.

Meyer, E. (and Luther, B.)

<u>Die Israeliten und ihre Nachbarstämme.</u> Halle. 1906.

Milik, J.T. Ten years of Discovery in the Judaean Wilderness and Their Importance. London, 1959.

Montgomery, J.A.

The Samaritans; the Earliest Jewish Sect.

Their History, Tehology and Literature.

(Bohle Lectures, 1906) Philadelphia, 1907

- Noth, M. <u>Die Welt des Alten Testaments</u>. Berlin, 1953.
- Nutt, J. W. Fragments of a Samaritan Targum. London, 1874.
- Petermann, H. <u>Versuch einer Hebräischen Formenlehre nach</u> der Aussprache der Heutigen Samaritaner. Leipzig, 1868.
- Roberts, B.J. The Old Testament Text and Versions.

 Cardiff, 1951.
- Robertson, E. Catalogue of the Samaritan Manuscripts in the John Rylands Library at Manchester.

 Manchester, 1938
- Skinner, J. The Divine Names in Genesis. London, 1914.
- Stevenson, W.B. Grammar of Palestinian Jewish Aramaic.
 Oxford, 1924.
- Thomson, J.E.H. The Samaritans. Their Testimony to the Religion of Israel. Edinburgh, 1919.
- Torrey, C.C. <u>Esra Studies</u>. Chicago, 1910.

Wernberg-Møller.

 \mathbf{p}_{\bullet}

The Manual of Discipline (Studies on the Texts of the Desert of Judah). Leiden, 1957.

Articles or Monographs.

in Periodicals or Festschrifts.

Bowman, J. "The Leeds Samaritan Decalogue Inscription".

Proceedings of the Leeds Philosophical Society,
VI, viii, 1951, pp. 567-75.

"The Exegesis of the Pentateuch among the Samaritans and Rabbis. Mudtestamentishe Studien viii, 1950, pp.220-62.

"The Samaritans and the Book of Deuteronomy".
Transactions of Glasgow University Oriental
Society, xvii, 1958-59, pp. 9-18.

"The Importance of Samaritan Researches".

Annual of Leeds University Oriental Society.

I. 1959, pp. 43-54.

Ben-Hayyim, Z. "The Samaritan vowel system and its graphic Representation". Archiv Orientalni, XXII, 1954, pp. 515-30.

Castro, F.P. "El sefer Abisa," Sefarad, xiii,1953, pp.119-29.

Dugmore, C.W. "Two Samaritan MSS in the Library of Queens' College, Cambridge". JTS xxxvi,1935, pp.131-47.

Geiger, A. "Zur Theologie u.Schrifterklärung der Samaritaner. ZDMG, xii. 1858, pp.132-42.

"Neuere Mittheilungen uber die Samaritaner". ZDMG, xvi, 1862, pp. 714-28.

"Die Gesetzlichen Differenzen Zwischen Samaritanern u. Juden."

ZDMG, xx, 1866, pp.527-73.

Kahle, P. "Untersuchungen zur Geschicte des Pentateuch-Textes". TSK, 1xxxviii, 1915, pp.399-439.

- Kahle, P. "The Abisha Scroll of the Samaritans".

 (contd.) Studia Orientalia Ioanni Pedersen dicata.

 Hannia, 1953, pp.188-192.
- Hempel, J. "Innermasoretische Bestatigungen des Samaritanus". ZAN, xi, N.F. 1934, pp.254-74.
- Roberts, B.J. "The Divergences in the pre-Tiberian Masoretic Text". JJS, i.g. 1949, pp. 147-55.
- Robertson, E. "The Ancient Scroll of the Samaritans".

 <u>Gaster Centenary Publication</u>, London, 1958.

 pp. 1-7.

Review of F.P.Castro's "Sever Abisa", Madrid, VT, xii, 1962. pp. 228-35. 1959.

- Rowley, H.H. "Sanballet and the Samaritan Temple".

 Bulletin of the John Rylands Library, Manchester.

 Vol. 38. No. 1, September, 1955.
- Slotki, I.W. "Long and Shorter Versions of Ancient Hebrew Poems." AJSL, 50, 1933-34, pp.15-31.
- Skehan, P.W. "Exodus in the Samaritan Recension from Qumran."

 JBL, lxxiv, 1955, pp. 182-87.

"The Qumran Manuscripts and Textual Criticism". VTS. iv. 1957. pp. 148-60.

- Sperber, A. "Hebrew based upon Greek and Latin Transliterations". <u>HUCA</u>, xii-xiii,1937-38, pp.151-274.
 - "Hebrew based upon Biblical Passages in Parallel Transmission". HUCA, xiv, 1939, pp.161-249.
- Talmon, S. "Synonymous Readings in the Old Testament".

 Scripta Hierosolymitana, viii, 1961, pp. 335-83.
